

Teresa Maria Armada Brás

Master in Environmental Engineering - Environmental Systems and Management Profile

The Effects of Extreme Weather Disasters in the European Food Availability

Thesis submitted to NOVA School of Science and Technology, NOVA University of Lisbon for the degree of Doctor of Philosophy (PhD) in Climate Change and Sustainable Development Policies

Supervisor: Doctor Maria Júlia Fonseca de Seixas, Associate Professor with Aggregation, NOVA School of Science and Technology, NOVA University of Lisbon

Co-supervisor: Doctor Jonas Jägermeyr, Science Collaborator, NASA Goddard Institute for Space Studies

Co-supervisor: Doctor Nuno Carvalhais, Leader of the Model-Data Integration Group, Max Planck Institute for Biogeochemistry

Júri:

- President: Doctor António da Nóbrega de Sousa da Câmara, Professor, NOVA School of Science and Technology, NOVA University of Lisbon
- Examiner(s): Doctor Ana Iglesias, Professor, Escuela Técnica Superior de Ingeniería Agronomía, Alimentaria y de Biosistemas, Universidad Politécnica de Madrid

Doctor João Carlos Andrade dos Santos, Associate Professor with Aggregation, Escola de Ciências e Tecnologia, Universidade de Trásos-Montes e Alto Douro

Vogal(s): Doctor José Manuel Osório de Barros de Lima e Santos, Professor, School of Agriculture, University of Lisbon

> Doctor António da Nóbrega de Sousa da Câmara, Professor, NOVA School of Science and Technology, NOVA University of Lisbon

> Doctor Maria Júlia Fonseca de Seixas, Associate Professor with Aggregation, NOVA School of Science and Technology, NOVA University of Lisbon



The Effects of Extreme Weather Disasters in the European Food Availability

Copyright © Teresa Maria Armada Brás Faculdade de Ciências e Tecnologia da Universidade Nova de Lisboa

A Faculdade de Ciências e Tecnologia e a Universidade Nova de Lisboa têm o direito, perpétuo e sem limites geográficos, de arquivar e publicar esta dissertação através de exemplares impressos reproduzidos em papel ou de forma digital, ou por qualquer outro meio conhecido ou que venha a ser inventado, e de a divulgar através de repositórios científicos e de admitir a sua cópia e distribuição com objetivos educacionais ou de investigação, não comerciais, desde que seja dado crédito ao autor e editor.

This work has been made possible by the financial support of Fundação para a Ciência e Tecnologia with the grant PD/BD/114570/2016

Acknowledgements

Firstly, I would like to express my gratitude and respect to Prof. Júlia Seixas, my supervisor, for her guidance during my research, for encouraging me to explore a growing research field at the Energy and Climate research group, and for her continuous and strong support that allowed me to carry out my PhD.

I am very grateful to Nuno Carvalhais and Jonas Jägermeyr, my co-supervisors, for their guidance, dedication, and fruitful discussions during my PhD. Thanks to Nuno, for encouraging me to learn a new programming language (an essential element to accomplish my research), for asking me many questions while stimulating my critical thinking, and for his positive attitude and enthusiasm that brought me confidence. Thanks to Jonas for his excellent dedication, for the sharing of knowledge, for raising the level of difficulty while motivating me to enhance the quality of my research.

I also want to thank to Prof. José Lima Santos for his valuable insights which allowed me to widen my research from different perspectives.

Thanks to Nuno Pacheco and Prof. Teresa Calvão for their availability to provide me useful clarifications on ArcGIS.

My warmest thanks to my former and present colleagues at the Centre for Environmental and Sustainability Research (CENSE), for their friendship, support and for the good lunch breaks. Big thanks to Patrícia Fortes, Sofia Simões, João Pedro Gouveia, Luís Fazendeiro, Pedro Palma, Katherine Mahoney, Filipa Amorim, Joana Monjardino, Hugo Tente, Sofia Teixeira, Paulo Pereira, Luísa Mendes, Babar Mujtaba, and Evaldo Costa for the daily companionship. Thanks also to Patrícia and Sofia for the opportunity to take part of the CLIM2POWER ERA-NET research project, which allowed me to continue with my PhD research while developing complementary data processing skills. My most sincere appreciation to Flávia Collaço, Luís Dias, and Juliana Barbosa, my comrades and good friends, for sharing the happiness of my achievements, for the fruitful discussions, and for helping me finding ways of overcoming difficulties along this journey. Thanks also to my colleagues and professors of the sixth edition of the Doctoral Program on Climate Change and Sustainable Development Policies, for the enthusiastic atmosphere that contributed to an exciting start of the PhD, and in particular to Filipa Ferreira for her friendship and for always motivating me to overcome the challenges that I came across during this journey.

This thesis have been made possible with the support of Fundação para a Ciência e Tecnologia, the Faculty of Sciences and Technology of NOVA University of Lisbon, and CENSE, which helped me with research grants and funding to take part in conferences and workshops and materials. I also thank to

the secretariat of the Department of Environmental Sciences and Engineering (DCEA), Filomena Gomes, Sandra Ferreira, and Sandra Alberto, for making bureaucracies easier.

Finally, I would like to thank my family and close friends for their encouragement and support during this journey. A special thanks to Marta Ruxa, Ana Rodrigo, and Francesca Poggi for helping me in disconnecting from the PhD, relaxing and having good times, especially through outdoor activities.

To my parents, César and Margarida, for the example of determination and strong will, love, affection, advices, unconditional support and encouragement. To my sisters Ana, Joana, Helena, and Inês for our strong sisterhood, for listening me, for their advices, unconditional support and for pushing me to do better and never give up. Thanks to my nephew Francisco for always make me smile. Finally, to my husband Bruno Baptista for always be by my side, for listening and believing in me and for being a source of strength and love during this journey.

Abstract

Over the last 20 years, it was reported an increase in worldwide extreme weather disasters (EWD), such as droughts (+29%), floods (+134%), and extreme temperature (+232%) like heatwaves. While the mortality rate of these events decreased, EWD are associated with a significant increase in economic damage and in the number of people affected (> 3 billion). The EWD can significantly impact agriculture by exacerbating fluctuations in crop yields and, consequently, in food availability and food prices. Thus, by means of the interconnections of the world food system, EWD have potential to threaten local to global food security. The challenges for agriculture are not only linked to changes in the long-term average climate, but particularly to EWD, which are usually more impactful and generally more uncertain. However, in the occasion of EWD occurrence, national and international disaster loss databases typically report populations affected and damage to human infrastructure, but rarely report damage or losses in the agriculture sector. As a result, agricultural impacts associated with these events are not well quantified across larger spatial scales. In particular, it remains insufficiently understood what are the trends in crop losses, and what are the implications that EWD may represent to food trade.

The European Union (EU) was chosen as a case study, as it is one of the largest global exporter and importer of agri-food products, with its food system deeply linked with other regions. Here, disaster records were used as a metric for extreme weather event impact analysis. Records of droughts, heatwaves, floods, and cold waves (EM-DAT) were combined with observational agricultural data (FAOSTAT) to evaluate disasters crop responses in Europe and in its Non-EU food suppliers. A superposed epoch analysis (SEA) – a time series statistical method used in data analysis – was used to estimate the impact of EWD on the average production, yield, and harvested area of selected crops. The larger implications of disaster impacts in Non-EU food suppliers to the EU food imports, were explored based on the import share per supplier (EUROSTAT). At the EU level, in addition to the SEA to estimate crop impacts, the trend of production anomalies was evaluated over time, per disaster type and per bioclimatic region. The research carried out allows to assess the effects of EWD in the EU food availability, while expanding the analysis to different crops and geographical regions. In particular, the EU crop production resulting from the occurrence of such events.

Despite a diversified external market, the EWD impacts on crops grown in Non-EU suppliers represent a substantial and negative exposure to EU food imports. Production losses of soybeans, tropical fruits, and cocoa associated to droughts and heatwaves but also floods, lead to an overall decline, up to 16%, in the EU import-weighted share of each crop. At the EU level, the severity of aggregated heatwave and drought impacts on crop production roughly tripled over the last 50 years. In particular, in every new year with a drought, the EU cereal production losses increase by 3%. The frequency of droughts, heatwaves,

floods, and cold waves significantly increased over time. Major losses are found for cereals, but also vegetables and oil crops in the Eastern countries, while smaller losses are estimated in Southern but also Central European countries. Even though using a weather disaster record for crop impact analyses has limitations, it offers a unique and standardized metric indicating that, at the EU level, climate change is already driving increasing crop losses in observational records. Understanding the effects of EWD on crop responses in the past and present climate contributes to the discussion of strategies and priorities in view of improving food systems resilience, including on the potential role of trade policies to support adaptation actions.

Keywords

Extreme Weather Disasters; Historical Impacts; Agricultural Crops; European Union; Non-EU Food Suppliers; EU Import Share-Weighted Impacts; European Bioclimatic Zones; Food Security; Food Availability; Climate Change.

Resumo

Nos últimos 20 anos foi registado, a nível global, um aumento dos desastres climáticos extremos (EWD), tais como secas (+29%), cheias (+134%) e temperaturas extremas (+232%) como ondas de calor. Embora a taxa de mortalidade desses eventos tenha diminuído, os EWD estão associados a um aumento significativo nos danos económicos e no número de pessoas afetadas (> 3 biliões). Os EWD podem causar impactos significativos na agricultura, exacerbando as flutuações na produtividade das colheitas e, consequentemente, na disponibilidade alimentar e nos preços dos alimentos. Por conseguinte, através das interconexões do sistema alimentar global, os EWD podem potencialmente ameaçar a segurança alimentar global e local. Os desafios para a agricultura não estão apenas ligados a mudanças médias no clima no longo prazo, mas particularmente, à ocorrência de EWD, que geralmente são mais impactantes e mais incertos. No entanto, aquando a ocorrência desses eventos, as bases de dados nacionais e internacionais que registam as perdas associadas aos EWD, geralmente reportam as populações afetadas, assim como danos à infraestrutura humana, mas raramente reportam danos ou perdas no setor agrícola. Como resultado, os impactos agrícolas associados a esses eventos não estão bem quantificados em grandes escalas espaciais. Em particular, também não é claro quais são as tendências nas perdas das colheitas, e quais são as implicações que os EWD podem representar para o mercado de alimentos.

A União Europeia (UE) foi escolhida como caso de estudo, por ser um dos maiores exportadores e importadores mundiais de produtos agroalimentares, estando o seu sistema alimentar profundamente ligado a outras regiões. Neste trabalho, os registos de desastres climáticos foram usados como uma métrica para análise do impacto de eventos climáticos extremos. Os registos de secas, ondas de calor, cheias, e ondas de frio (EM-DAT) foram combinados com dados agrícolas observados (FAOSTAT) para avaliar a resposta das culturas agrícolas aos EWD na UE e nos países exportadores de alimentos. A análise de época superposta (SEA) – um método estatístico de análise de séries temporais – foi usada para estimar o impacto dos EWD na produção, produtividade, e área de colheita de um conjunto de culturas. Avaliou-se o impacto dos EWD nos países produtores de alimentos que exportam para a UE, bem como as implicações associadas às dependências comerciais da região (com base nas estatísticas de balanços comerciais da EUROSTAT). Ao nível da UE, para além da implementação da SEA para estimar o impacto dos EWD nas culturas produzidas na região, avaliou-se também a tendência das anomalias da produção agrícola ao longo do tempo, por tipo de EWD e por região bioclimática. Este trabalho de investigação permite avaliar os efeitos dos EWD na disponibilidade alimentar da UE, sendo a análise expandida a diferentes culturas e regiões bioclimáticas. Em particular, foi avaliada a exposição da dependência das importações de alimentos da UE aos EWD, e também o grau de perda na produção agrícola da UE resultante da ocorrência de tais eventos.

Apesar da UE ter um mercado externo diversificado, os impactos dos EWD nas culturas produzidas nos países exportadores não Europeus, representam uma exposição substancial e negativa à importação Europeia de alimentos. As perdas de produção de soja, frutos tropicais e cacau, associadas à ocorrência de secas e ondas de calor, mas também de cheias, podem potencialmente reduzir, até 16%, a importação alimentar da Europa. Ao nível da produção na UE, as perdas agrícolas associadas às ondas de calor e secas praticamente triplicaram nos últimos 50 anos. Em particular, em cada novo ano com uma seca, as perdas na produção de cereais na UE têm vindo a aumentar 3%. A frequência de secas, ondas de calor, cheias, e ondas de frio aumentou significativamente ao longo do tempo. As maiores perdas são estimadas para os cereais, mas também vegetais e oleaginosas nos países do Leste Europeu, enquanto perdas menores são estimadas no Mediterrâneo, mas também nos países da Europa Central. A utilização de registos da ocorrência de EWD na análise do seu impacto na produtividade agrícola, apesar de ter limitações, consiste numa métrica única e padronizada que indica que, ao nível da UE, as alterações climáticas já estão a causar perdas crescentes na agricultura. Compreender os efeitos que os EWD tiveram e têm na agricultura, contribui para a discussão de estratégias e prioridades com vista a melhorar a resiliência dos sistemas alimentares, incluindo o potencial papel das políticas comerciais para apoiar ações de adaptação.

Palavras-chave

Desastres climáticos extremos; Impactos históricos; Culturas Agrícolas; União Europeia; Fornecedores não Europeus de alimentos; Impactos ponderados por percentagem de importação; Zonas Bioclimáticas Europeias; Segurança Alimentar; Disponibilidade de Alimentos; Alterações Climáticas.

Table of Contents

1	Intro	Introduction			
	1.1	Relevance of the study	1		
	1.2	Research background	2		
	1.2.1	Brief history of food	2		
	1.2.2	The effects of a changing climate in the food system	6		
	1.2.3	The European Union in the food system	8		
	1.3	Research questions	12		
	1.4	Research design and structure of the dissertation	12		
	1.5	References	13		
2	Revie	w on the climate change impacts on food availability and access	21		
	2.1	Introduction	21		
	2.2	Methods	23		
	2.3	Reviewed climate change impacts on food security dimensions	23		
	2.3.1	Impacts on food availability	25		
	2.3.2	Impacts on food access	29		
	2.4	Contribution for food governance and policy coherence	33		
	2.5	Conclusion			
	2.6	References	39		
3	Expo	ure of the EU food imports to extreme weather disasters in exporting countries	43		
	3.1	Introduction	44		
	3.2	Methods	46		
	3.2.1	Mapping external dependency and sufficiency of EU agri-food supply	46		
	3.2.2	Impact of EWD on crop production in exporting countries relevant for the EU	47		
	3.2.3 Exposure of EU agri-food imports		51		
	3.3	Results	52		
	3.3.1 the E	Assessing the impact of EWD in crop production in the exporting countries relevant J 52	for		
	3.3.2	Assessing the exposure of EU agri-food imports	53		
	3.4	Discussion	53		
	3.5	Conclusion	60		
	3.6	References	60		
4	Drou	ht and heatwave crop losses tripled over the last five decades in Europe	63		

	4.1	Introduction	64
	4.2	Methods	65
	4.3	Results	68
	4.3.1	EU crop response to extreme weather disasters	68
	4.3.2	2 Crop impact and frequency of extreme weather disasters over time	70
	4.3.3	Severity of extreme weather disasters across different climate regions	71
	4.4	Discussion	72
	4.5	Conclusion	78
	4.6	References	79
5	Gene	eral discussion and conclusions	
	5.1	Key contributions	
	5.2	Contributions at the policy level	91
	5.3	Final remarks and future research	92
	5.4	Outputs	94
	5.5	References	95
6	Арре	endix	
	6.1	Appendix A	99
	6.2	Appendix B	109
	6.3	Appendix C	121

List of Figures

Figure 1.1	Contributions to global production from major producing regions in 2000
Figure 1.2	Weather-related food price increases
Figure 1.3	Top four EU countries with the highest crop category production share
Figure 2.1	Methodology for a systemic literature review
Figure 2.2	Observed climate change impacts on food availability
Figure 2.3	Projections on worldwide crop yield change due to climate impacts
Figure 2.4	Future climate change impacts on food availability
Figure 2.5	Future climate change impacts on food access
Figure 3.1	Geographic distribution of EU food import dependency
Figure 3.2	EU self-sufficiencies per crop category and import shares per external supplier
Figure 3.3	Composite figure of EWD influences on observed crop production, yield and harvested area. 59
Figure 3.4	Impacts of EWD on external EU suppliers and on EU imports per crop category
Figure 4.1	Impacts of Extreme Weather Disasters on European crops
Figure 4.2	Drought and heatwave crop responses in the first and second half of the observation record. 71
Figure 4.3	Cereal production anomalies during years of reported EWDs75
Figure 4.4	Number of annually reported EWDs in Europe76
Figure 5.1	Potential exposure of EU food imports to EWD in non-EU suppliers
Figure 5.2	Drought and heatwaves impacts in crop production in the EU and in its bioclimatic regions88
Figure 5.3	Projected climate change impacts on crop production by 2050 under RCP8.5

List of Tables

Table 1.1 Food security threats driven by unsustainable human practices	5
Table 1.2 Research design, including the general goal of this thesis, the research questions that are	
addressed in each chapter, and the main methodological approaches	12
Table 2.1 Strategies selected from the literature that strength food availability and access.	37
Table 3.1 List of goals and respective data sources used in this study	45
Table 4.1 Composite drought and heatwave impacts by crops and climate region	77
Table 5.1 Revisiting the research design including the main findings to the research questions	90

Acronyms

- AQUASTAT FAO's Global Information System on Water and Agriculture
- CAP Common Agricultural Policy
- CER Cereals
- Cfb Temperate Oceanic Climates
- Csa Hot-Summer Mediterranean Climates
- Dfb Warm-Summer Humid Continental Climates
- Dfc Subarctic Climates
- EM-DAT The International Disasters Database
- EU European Union
- EUROSTAT European Statistic Database
- EWD Extreme Weather Disasters
- FAO Food and Agriculture Organization
- FAOSTAT Food and Agriculture Organization Database
- IPCC Intergovernmental Panel on Climate Change
- ISIMIP Inter-Sectoral Impact Model Intercomparison Project
- KG Koeppen-Geiger Classification
- Non-CER Non-Cereals
- RCP Representative Concentration Pathways
- SEA Superposed Epoch Analysis

"I encourage all of us, whatever our beliefs, to question the basic narratives of our world, to connect past developments with present concerns, and not to be afraid of controversial issues."

> Yuval Noah Harari (2015) Sapiens: A Brief History of Humankind

1 Introduction

1.1 Relevance of the study

As for many civilizations that have come and gone, also today, food is the weak link in our modern society (Yearley, 2013). World population is growing and, demanding more food, while grain stocks have been falling (Benton, 2019; Torero, 2016). From water scarcity to an increasing reliance on fossil fuels, and ending on climate change, a set of unsustainable human practices are threatening the production of food (Lang and Ingram, 2013; Santos, 2010). Food has been produced under the maximization of commerce based on the production of fewer crops supplying excessive calories and standardised diets (Benton, 2019; Lang and Ingram, 2013). As a result, more than 60% of the calories in the human diet is highly dependent on just four grains – wheat, maize, rice, and soybeans – which are grown in only a handful of countries. Such high level of concentration implies that the world's capacity in coping with geographical risk is limited (Global Food Security Programme UK, 2015a; Tai *et al.*, 2014).

Weather-related shocks are particularly damaging to crops and to food production systems, as they can significantly influence the year-to-year variability in crop yields at various spatial scales, and trigger price spikes (Jägermeyr and Frieler, 2018; Lesk *et al.*, 2016; Puma *et al.*, 2015; Ray *et al.*, 2015; Rosenzweig *et al.*, 2014b; Vogel *et al.*, 2019). By means of the interconnections of the world food system, extreme weather events have thus potential to destabilize food systems and threaten local to global food security (Lesk *et al.*, 2016; Nelson *et al.*, 2014; Rosenzweig *et al.*, 2014b), affecting producers and consumers (Haile *et al.*, 2017; Rosenzweig *et al.*, 2001). The severity of an extreme weather event and the vulnerability and exposure of the human and natural systems to it will determine whether it results in a disaster (IPCC, 2012). National and international disaster loss databases typically report populations affected and damage to human infrastructure, but rarely report damage or losses in the agriculture sector (FAO, 2015). As a result, there are major data gaps of the extent that extreme weather disasters (EWD) impact the agricultural sector.

The European Union (EU) is one of the world cereals breadbaskets (Berkhout *et al.*, 2018; Ciscar *et al.*, 2018). The EU is also highly dependent on crops that do not naturally grow in the region, and come from countries considered to be highly vulnerable to a changing climate (EU, 2018; Hanks and Craeynest, 2014; IPCC, 2014). Consequently, the effects from EWD on trade dependencies, and in the EU agricultural productivity are growing concerns (EEA, 2019). As an example, the European heatwave and drought, in summer 2018, led to widespread cereal production losses (8%) and triggered a sharp increase in commodity prices (<48%), in many countries across the continent (DG AGR, 2018; EC, 2018). Future projections suggest an increase in summer dryness in most parts of Europe, with longer and more intense heatwaves and droughts (IPCC, 2019). In view of potential future aggravations in global EWD frequency and intensity due to climate change (IPCC, 2019, 2012), there is still little quantitative evidence of how historical impacts

of such events affected the production of different crops and in different European regions. It remains also unclear how the agricultural impacts of EWD in Non-European countries, affect the EU food availability. This information is crucial to define appropriate risk reduction policies and investments in agriculture (FAO, 2015), and also to understand if and how trade policies can support climate adaptation strategies (EEA, 2019).

1.2 Research background

1.2.1 Brief history of food

At the end of Pleistocene or early Holocene, about 10 000 years ago, the Agriculture Revolution changed the way humans lived and paved the path for significant interventions in the Earth systems (Duarte Santos, 2012; Harari, 2015; Simmons, 2010). The transition from nomadic hunter-gatherer lifestyles to farming was a long evolutive process where humans learned how to domesticate plants and animals, and to develop techniques for drying, smoking, and storing food (Duarte Santos, 2012; Harari, 2015). The earliest known developments of agriculture began in the Near East and slowly, sprang up in different parts of the world although in an independent way (Harari, 2015), as by the end of the last glacial period (i.e. between 110 000 to 10 000 years B.P.), the sea level rise did not allow for any physical connection between Eurasia with the Americas, Australasia or with the West Pacific islands (a few of which populated for at least 30 000 years) (Christian, 2018). About 18 000 years ago, although with erratic episodes, temperatures and precipitation started rising and, slowly, climate become humid and warmer, and also more stable, thus setting the scene for a viable agriculture (Duarte Santos, 2012). With the gradual movement to permanent villages and with the increase in food supply, the population began to grow from 5-8 million nomadic foragers before the transition to agriculture, to 200 million humans 2000 years ago (Harari, 2015). The challenge to sustain an increasing population was settled, and there was no turning back.

During the Industrial Revolution, which lasted from roughly the mid-1800s through World War I, a major human intervention in the nitrogen cycle was made with the production of artificial fertiliser. Peru and Chile, for example, exported to North America and Europe, hundreds of millions of tonnes of sodium nitrate and nitrogen-rich guano, intensely increasing agricultural productivity (Mellilo, 2012). Other crucial advances included the use of agricultural machines powered by fossil fuels, improved crop rotation systems, selective breeding, or the production of chemical pesticides. Such innovations contributed to amplify food production and underpinned a rapid growth of population – to 900 million by 1800 – and major increases in life expectancy (Christian, 2018; Jägermeyr, 2020).

After World War II, when many developing nations struggled to feed their people, disease-resistant and high-yield crops – in particular cereal grains – were introduced by using genetic modification through the work of Norman Borlaug (Christian, 2018; Mellilo, 2012). Cereal yields tripled as a result of the generalized adoption of new varieties of high-productivity crops combined with a threefold expansion in irrigated areas and the widespread use of fertilisers (Santos, 2010). The increasingly large-scale, intensive and productive

agriculture, and also the possibility to import food from an expanding global and liberalised trade marked, have promoted the surplus of food at prices that were, on average, cheaper decade by decade (Benton, 2019; Santos, 2010). As a result, the Green Revolution favoured global crop production and underpinned a demographic explosion – from 2 600 million in 1950 to 7 795 million people in 2020 (Christian, 2018; The World Bank, 2019; UNFPA, 2020).

The food trade has then being developed under the maximization of commerce based on the production of fewer crops supplying excessive calories and standardised diets (Benton, 2019). Consequently, more than 60% of the calories in the human diet is highly dependent on just four grains – wheat, maize, rice, and soybean (Yearley, 2013) – which are grown in a reduced number of countries (Fig. 1.1) (Global Food Security Programme UK, 2015b; Tai *et al.*, 2014). This high level of concentration implies that the world's capacity in coping with geographical risk is limited, since any shock to production in those countries will have an effect on global prices and price volatility (Puma *et al.*, 2015; Torero, 2016).



Figure 1.1 Contributions to global production from major producing regions in 2000. Regions or countries: US – United States of America; Europe – all countries in Europe, not including Russia; China – mainland China only; S Asia – all member countries of the South Asian Association for Regional Cooperation; SE Asia – all member countries of the Association of Southeast Asian Nations; S America – all countries in South America (excluding Central American countries) (Tai *et al.*, 2014).

The economic growth, particularly the rising incomes in China, India, and more recently in sub-Saharan Africa, lead to substantial changes in consumption patterns which are more varied but also richer in animal-protein-based (i.e. also called western diets), thus driving crop utilisation for livestock feed (Lang and Ingram, 2013; Santos, 2010; Timmer, 2005; Torero, 2016; Yearley, 2013). As a result, worldwide cereal demand has been growing at 2-3% per year, while cereal reserves have been declining – from 700 million tonnes in 2000 to less than 400 million tonnes in 2007 (meaning about 64 days of carryover stocks in 2007) (Lang and Ingram, 2013; Santos, 2010; Timmer, 2005; Torero, 2016; Yearley, 2013). When cereal stocks are low relative to use, markets are less able to cope with supply and demand shocks. Thus, supply shortfalls or a rise in demand will lead to larger price increases (FAO, 2009).

The food problem starts being particularly noticeable after a series of small weather-related production shocks in 2007–2008 that coupled with historically low stock levels, a financial crises and also due to the strong link with the energy markets (FAO, 2009), led to dramatic and sustained increases in the price of cereals, and food in general (Global Food Security Programme UK, 2015a; Santos, 2010). As a result of

these series of events, some countries imposed export barriers to ensure their own food supply, leading to a doubling of world grain prices (FAO *et al.*, 2011). A similar grain price spike in 2010-2011 occurred when intensive heatwaves hit Eastern Europe, Russia, and the United States. These price spikes (Fig. 1.2) created a number of significant impacts around the World, particularly in the countries hit with the weather shocks but also in import food-dependent nations through the interconnections of the world food trade (Global Food Security Programme UK, 2015a; Yearley, 2013). Particularly in countries with fragile governance, food price spikes spawned numerous food protests and riots, such as in Thailand, Egypt, Haiti, Mexico, and also in Middle East and North Africa partially sparkling the Arab Spring and triggering national and international refugee movements and social fragmentation (Global Food Security Programme UK, 2015a; Puma *et al.*, 2018; Yearley, 2013).

Food supplies and grain stocks are tightening, and food security – the state of having, at any time, a reliable access to a sufficient quantity of affordable, and nutritious food (Committee on World Food Security, 2015) – is undermined by a combination of threats driven by dramatically unsustainable human practices (Duarte Santos, 2012; Lang and Ingram, 2013; Santos, 2010; Yearley, 2013). As for many civilizations that have come and gone, also today, food is the weak link. For the Sumerians (4100-1750 B.C.), food shortages resulted from salinisation as a consequence of over-irrigation (Yearley, 2013), while for the Mayans (from 2000 B.C.), deforestation (to give place to agricultural crops) lead to soil erosion and to the intensification of droughts (Cook *et al.*, 2012). Now, even though we know what have failed with ancient civilizations, our growing population – who is demanding more food and higher animal-protein-based diets –, is facing water scarcity (driven by over-irrigation and depletion of aquifers), a lessening response in crop productivity to the use of fertilisers, an unacceptable level of food waste, a loss of biodiversity and agricultural land (driven by urbanization and desertification), and a high reliance on fossil fuels (Table 1.1). On the top of these threats, climate is changing (FAO, 2019a, 2011a; Lang and Ingram, 2013; Santos, 2010; Yearley, 2013).



Figure 1.2 Weather-related food price increases.

Historical FAO food and cereal price indices (as percentage of 2002–2004 averages), with vertical lines indicating events when a top five producer of a crop had yields 25% below trend line which is indicative of a seasonal climate extreme. At the same time, food prices are increasingly associated with the price of crude oil (blue line), thus making difficult the attribution of price changes (Porter *et al.*, 2014).

By 2050, with projections pointing to a population of 9.7 billion (UN, 2013) that will demand 60-70% more food (Alexandratos and Bruinsma, 2012; The Government Office for Science, 2011), 55% more water, and with over 40% of the global population living in river basins experiencing severe water stress (HLPE, 2015), feeding a growing population within the limits of sustainability, and under the threat of climate change, becomes a bigger challenge (Jägermeyr *et al.*, 2017). Notwithstanding, the problem is already at the table. There are 821 million people undernourished, mainly as a result of vulnerable and low-yield agricultural systems (FAO *et al.*, 2018; Jägermeyr, 2020), often in countries which are stage of political instability, armed conflicts and, overall, with a poor governance. Controversially, 2 billion are overweighted (World Health Organization, 2017) and 672 million obese (FAO *et al.*, 2018), as food-insecure families tend to choose less expensive foods that are often high in caloric density and low in diversity, micronutrients and fibre (FAO *et al.*, 2018).

Recognizing that the food problem is a complex and major global issue and is being undermined by a combination of social, economic, and environmental threats, the effects of a changing climate – per se an unique threat because of the millennial time scale of anthropogenic carbon within surface carbon reservoirs – on food availability are the focus of this study.

Threat	How the threat is undermining Food Security?
Water scarcity	Water scarcity, exacerbated by a changing climate, is driven by the needs of an increasing population, by over-irrigation and depletion of aquifers. Worldwide, 16% of the cultivated land produces 44% of crop production but consumes 70% of global freshwater withdrawals (HLPE, 2015). Nearly 20% of the water used for irrigation is estimated to be provided by non-renewable groundwater (Wada <i>et al.</i> , 2012). Overall, the efficiencies of irrigation systems are very low (<30%), particularly in south Asia and sub-Saharan Africa (Jägermeyr <i>et al.</i> , 2015). Such inefficiency along with the increasing consumption of water, is leading to a rapid depletion of aquifers in key grain producers – China, India, and the United States (Dalin <i>et al.</i> , 2017; Holden <i>et al.</i> , 2018). However, without irrigation global cereal production – providing > 60% of the food energy intake (FAO, 1997)) – would decrease by 20% (Siebert and Döll, 2010).
Biodiversity loss	Biodiversity underpins a wide range of ecosystems services, such as keeping soils fertile, pollinating crops, cleaning water, and fighting pests and diseases (FAO, 2019b). The loss of biodiversity is decreasing the resilience of ecosystems, including agricultural systems. The fact that more than 6000 crop species have been cultivated for food, but just nine account for 66% of total crop production, means that agriculture is highly based on monocultures, which are less resilient to temporal fluctuations in climate, and to shifts in pest occurrence and diseases, thus potentially undermining food production (FAO, 2019b; Lin, 2011).
Soil loss	Desertification – intensified by climate change, and driven by unsustainable farming practices and deforestation – as well as urbanization – resulting in the impermeabilization of soil for the construction of cities and its infrastructures – are leading to the loss of fertile soil at rates that are orders of magnitude greater than mechanisms that replenish soil (Amundson <i>et al.</i> , 2015; Santos, 2010).
Low fertiliser responses Low fortiliser responses Low for the genetic improvements that we have induced on cereals up until now. The technological that was at the basis of the Green Revolution – and was responsible for a triplication cereal production since 1950 – is highly focus on the grain, regardless on the plant root thus constraining its potential of maximum efficiency (Santos, 2010).	
Fossil fuel reliance	The food sector accounts for about 30% of the world's total energy consumption (FAO, 2011b), which is used to produce fertilisers, herbicides, pesticides, diesel for machinery, electricity for irrigation, heating, drying, processing, storage and packaging (Vermeulen <i>et al.</i> , 2012)). The sector represents 19-29% of global emissions (Vermeulen <i>et al.</i> , 2012), and thus is a considerable contributer to climate

Table 1.1 Food security	threats driven b	y unsustainable l	human practices
-------------------------	------------------	-------------------	-----------------

Threat	hreat How the threat is undermining Food Security?			
	change (IPCC, 2019). From those emissions, agricultural contributes with 80-86% including indirect emissions associated with land-cover change, livestock, rice fields and synthetic fertilisers (FAO,			
	2016a; Vermeulen et al., 2012). The food crises and the energy crises are thus very much interlinked.			
	Moreover, the rise in oil prices have led to investments to new sources of energy – such as biofules			
	- which, through soil occupation - directly competes with food production, thus leading to higher			
	prices of food (Santos, 2010).			
Food waste	Roughly one-third of food produced for human consumption is lost or wasted globally, which amounts to about 1.3 billion tonnes per year (FAO, 2011a). In the developing world, food waste is driven by poor production, harvesting and storage practices and by food losses due to pests and diseases. In developed countries, consumers behaviour is pointed as the major cause of food waste, such as the rejection of imperfect products or the poor management of their food inventory (FAO, 2011a).			
	The economic grow and higher incomes, particularly in emergent societies, coupled with			
Changing	urbanization and the increasing influence of the retailing sector, is pushing up the consumption of			
diets	varied, high-quality but also meat-based diets, which require more energy, water and land resources			
	(Lang and Ingram, 2013; von Braun, 2007).			
	A warming climate is directly affecting the amount of food through its direct impacts on crop yields,			
	and also through impacts on water availability and quality, pests and diseases, pollination services,			
Climate	and through higher OO_2 concentrations in the atmosphere – which is today higher than at any point is time sizes the next 200 000 years D D , thus examine the farther struct grapheness offset			
change	In time since the past 800 000 years B.P., thus amplifying the Earth's natural greenhouse effect			
-	(NOAA, 2020) – anecting biomass and nutritional quality of crops (IPCC, 2019). Such adverse enects			
	in frequency and intensity with climate change (IPCC, 2019)			

1.2.2 The effects of a changing climate in the food system

Climate stability is under threat, largely because of a growing reliance on fossil fuels and (but not only) industrialised and unsustainable forms of agriculture (Rockström *et al.*, 2009). Since the Industrial Revolution, changes in the long-term mean of temperature and precipitation, as well as changes in weather variability, have been observed.

Most of agriculture – and thus the production of food – remains highly dependent on climate even despite major technological improvements, because solar radiation, temperature, and precipitation are the main drivers of crop growth (Rosenzweig *et al.*, 2001). Thus, the spatial patterns of crops yield across the Planet are governed by the current spatial distribution of climate drivers. In the same way, the relative productivity of the seasons is determined by the weather variability. Climate change affects climate variables by changes in their means but also in their variability, which is just as important as changes in the average (Global Food Security Programme UK, 2015a).

Under high temperatures the plant curls its leaves in order to reduce its exposure to the sun, which reduces the photosynthesis and thus crop yield. High temperatures and reduced soil moisture by means of a decline in precipitation, can lead the stomata to close (to diminish the evapotranspiration), which reduces the CO₂ intake and thus crop photosynthesis and yield (Yearley, 2013). Long-term average climate can, however, bring some localized benefits for agriculture, such as the increased precipitation, the length of growing seasons or the influence of higher levels of CO₂ on yields (Christidis *et al.*, 2015; Deryng *et al.*, 2014; Elliott *et al.*, 2015; Hov *et al.*, 2013; Iglesias and Garrote, 2015; Mueller *et al.*, 2015; Rosenzweig *et al.*, 2014a)

Evidence shows that average changes on long-term temperatures are shifting production seasons, and that crop responses are highly variable, and not only negative, from place to place and for different food items. Future impacts are expected to be consistent with the trajectory of past impacts, with the majority of locations experiencing crop losses, while some locations may benefit particularly under crop adaptation (IPCC, 2019; Porter *et al.*, 2014). A changing climate raises new factors and possibilities potentially exposing societies to risk, such as potential changes in the current spatial production patterns – in particular in the world breadbaskets (see Chapter 2).

In addition to those uncertainties, climate change - through an increase weather variability - may lead to crop yield fluctuations, which can have major impacts on the livelihoods of subsistence farmers and may trigger significant global price fluctuations (Frieler et al., 2017; Piesse and Thirtle, 2009; Porter et al., 2014; Puma et al., 2015). Low yield variability shall lead to stable food supply (and thus to stable farmer incomes), preventing price spikes. On the contrary, high yield variability is particularly damaging to crops and to food production systems, as it can trigger price spikes, thus potentially destabilizing food systems and threaten local to global food security (Lesk et al., 2016; Nelson et al., 2014; Puma et al., 2015; Ray et al., 2015; Rosenzweig et al., 2014b). Globally, nearly one third of observed yield variability (i.e. 22 million tonnes for maize, 9 million tonnes for wheat, 3 million tonnes for rice and 2 million tonnes for soybean) can be explained by the additional climate variability that derives from climate change (which in some world regions can be more than 60%) (Ray et al., 2015). Other study shows that growing season climate factors - including mean climate as well as climate extremes – explain 20% to 49% of the variance of yield anomalies (depending on the crop); 18% to 43% of the explained variance is attributed only to climate extremes (Vogel et al., 2019). By means of trade and sufficient grain storage, under normal climate conditions or small weather year-toyear fluctuations, the global food system can compensate for local crop losses. However, under extreme weather conditions - as in 2007-2008 or 2010 - and in the absence of grain reserves coupled with an intensive, little diversified in crops, and thus less climate-resilient agriculture, the global food system is extremely under threat. The severity of an extreme weather event and the vulnerability and exposure of the human and natural systems to it will determine whether it results in a disaster (IPCC, 2012).

While long-term average climate may benefit few locations, at least up to a certain level of temperature and CO₂ increase, the impacts from extreme weather disasters (EWD) – like droughts, heatwaves, and floods – are invariably negative (Global Food Security Programme UK, 2015a; Hov *et al.*, 2013; IPCC, 2019, 2012; Lesk *et al.*, 2016; Vogel *et al.*, 2019). Particularly, the most extreme events imply robust disaster risk reduction and management strategies in the structure of the food system, in addition to the long-term average climate adaptation. For example, climate change may result, on average, in an area getting wetter, but if the variance on precipitation is also increasing, it is also possible for both floods and droughts to become more common. Another example is an increase in temperature variance without a change in the mean, which may imply an increase in the frequency of both hot and cold extremes, as well as in the intensity of the extremes (Global Food Security Programme UK, 2015a; IPCC, 2013). The rarest conditions are the

most uncertain and difficult to study, but because they are also typically the most impactful, their study is most important (Global Food Security Programme UK, 2015a).

The challenges for agriculture are therefore not only linked to changes in the long-term average climate, but particularly to changing weather extremes. Climate models provide a good understanding of how climate may change in the future, thus by means of crop-based models or statistical models, one can better quantify its impacts on crop yields (Moore and Lobell, 2015; Ray et al., 2019). However, our understanding of the way extreme events change is much less certain, as well as any inference on their impact (Global Food Security Programme UK, 2015a; IPCC, 2013; Min et al., 2011; Müller et al., 2017; Roberts et al., 2017). Empirical research has investigated the impacts of extreme weather events on crops for individual countries, regions, at the farm level (Lüttger and Feike, 2018; Powell and Reinhard, 2016; Troy et al., 2015), or across world regions by using crop data at sub-national spatial resolution (Vogel et al., 2019), all combined with climate data by means of extreme weather indicators, such as absolute, threshold or duration indices. Such empirical approaches may, however, underestimate the crop effects from EWD because similar extreme weather events may have divergent effects depending on the vulnerability of the exposed system (Lesk et al., 2016). That recent study (i.e. Lesk et al., 2016), by means of a statistical approach, estimated the influence of EWD in cereal production on aggregated world regions. The EWD impacts on other important crops and regions, as well as, the associated implications to import dependences have, however, not been explored.

National and international disaster loss databases typically report populations affected and damage to human infrastructure, but rarely report damage or losses in the agriculture sector (FAO, 2015). As a result, there are major data gaps of the extent to which EWD impact crop yields. In particular, there is still little quantitative evidence of increasing trends in crop losses associated to disasters, and also on its implications through trade (Lesk *et al.*, 2016; Puma *et al.*, 2015). The main focus of this thesis is on the impact of EWD on agriculture. A better understanding of the crop responses to EWD is of most importance to defining appropriate risk reduction policies and investments for agriculture (FAO, 2015). Such knowledge is also crucial to understand if and how trade policies can support climate change adaptation strategies and actions (EEA, 2019). More emphasis must be given to the study of the impact of EWD in agriculture, especially because there is an agreement that some of these events are becoming more likely as a result of climate trends (IPCC, 2019; Porter *et al.*, 2014).

1.2.3 The European Union in the food system

1.2.3.1 The international dimension

The European Union with 28 Member States (EU) is one of the leading global players in food and agriculture, and its food system is deeply linked with other regions (Berkhout *et al.*, 2018; Ciscar *et al.*, 2018). The EU produces about 13% of world's cereal production -6% of the world's maize and 18% of the world's wheat -

representing 24% of global cereal exports (Ciscar *et al.*, 2018; FAO, 2019c; Knox *et al.*, 2016). The EU represents nearly 50% of the world's wine (Wine Institute, 2017), and 70% of the world olive oil exports (International Olive Council, 2018). The region is also the world's leading producer of sugar beet, contributing 50% of the global sugar production (Ciscar *et al.*, 2018; Knox *et al.*, 2016).

The EU is a large market with over 500 million consumers and is deeply integrated into global markets through the World Trade Organization. The international dimension of the agricultural sector is part of the *Common Agricultural Policy* (CAP), which is set out in the *Treaty on the Functioning of the European Union* (that specifically targets the stabilizing of imports and exports as a means to address market volatility and deliver on the objectives of the CAP) (European Commission, 2019). The majority of the EU-trade is internal, with nearly 73% of the EU food exports and imports being directly traded within EU countries (Berkhout *et al.*, 2018). The EU exports account for 50% of agricultural food and feed products (i.e. commodities, other primary and processed agricultural products), 33% of food preparations and beverages, and 11% of non-edible agricultural products. The agricultural food and feed products account for 80% of total EU imports, followed by food preparations and beverages (9%), and non-edible products (10%) (EU, 2018).

On the extra-EU trade, nearly 40% of exports – mostly beverages and food preparations – are to the United States of America, China, Switzerland, Russia, and Japan. The top EU-extra suppliers are Brazil, the United States, Argentina, Ukraine, and China, and in a lower extend but increasing, Indonesia and India. Third countries account for nearly 30% of total EU extra imports (Berkhout *et al.*, 2018), supplying the EU with products that are not grown in the EU itself due to its natural conditions (e.g. tropical fruit, coffee and fresh or dried fruits), products that are mostly used for animal feed (e.g. oilcakes and soybeans), and also used as ingredient in further processing (e.g. palm oil) (EU, 2018).

1.2.3.2 The agricultural sector

Agriculture and food-related industries and services provide over 44 million jobs in the EU, and 22 million people are directly employed in the sector itself (EEA, 2019). The agricultural sector contributes, on average, with 2.5% of the GDP (FAO, 2016b), whereas income from agriculture varies across European regions, and is generally high in relative economic terms in Portugal, Spain, Greece, France, Bulgaria and Romania (European Commission, 2009). The sector is, on average, the second major consumer of freshwater resources (27%) after industry (42%) and followed by municipal (25%), thus contrasting with estimations at the global level where agriculture irrigation accounts for 70% of total water withdrawal (FAO, 2016b).

Nearly 39% (or 173 million hectares) of the total EU land area is used for agricultural production (EUROSTAT, 2019). Denmark and Hungry have the highest rates (>40%) of cultivated area among the total area of the country and The Netherlands, Greece and Italy have the highest shares (>40%) of irrigation among their cultivated areas (FAO, 2016b). Cropping area is mostly occupied with cereals (65%) largely grown in the North and Central Europe, vineyards and olive trees (9%) both cash crops and mostly grown

in the Mediterranean region. The cropping area also grows oil crops (13%), vegetables (4% each), roots and tubers, sugar and orchards (2% each) (FAO, 2019c). The highest producers of wheat and barley are France (24%), Germany (17%), and the United Kingdom (10%); of maize is France (23%), Romania (15%), Italy (12%), and Hungry (11%); and of sugar beet is France (30%), Germany (23%), and Poland (10%). The main EU producers of olives are Spain (52%), Italy (23%), and Greece (20%), while for grapes are Italy (31%), France and Spain (each with 24% of production) (Fig. 1.3).

In the EU, cereals are used for animal feed (66%), human consumption (33%), and biofuels (1%). Oilseeds (mostly rapeseeds, sunflower and soybeans) are used for food, feed, fuel, and industrial purposes, but also for vegetables oils and meal (which are an important component of animal feed). Less than 10% of the soybean consumed in the EU (i.e. soybean, soybean meal and/or soy oil) is grown in the region, being the remaining imported from Brazil, Argentina, the United States, Paraguay, and Canada (Berkhout *et al.*, 2018). Nearly two thirds of the rice consumed in the EU is grown in the region, being the remaining supplied by India, Thailand, Pakistan, Egypt (among other countries) (European Commission, 2020; EUROSTAT, 2016; FAO, 2017).



Figure 1.3 Top four EU countries with the highest crop category production share.

Countries acronymus according to ISO3 codes. Crop data is acquired from FAO (2017) and is averaged between 2008 and 2017. Temperate cereals include wheat and barley; Tropical cereals include millet and sorghum, Temperate roots include sugar beet.

1.2.3.3 The impacts of a changing climate in Europe

As in the majority of the world regions, the EU food system has been adversely affected by a changing climate. For example, from 1974 to 2008, it is estimated that, on average, a combination of changes in temperature and precipitation patterns negatively affected yields of maize (up to -25%) and wheat (up to -15%), although yield gains are estimated for rapeseed (3%) (Ray *et al.*, 2019). Higher yield losses have been estimated for the Mediterranean countries, and gains in few northern European countries (Moore and Lobell, 2015). Recent studies confirm that observed changes in climate have already affected crop suitability in Europe, raising concerns about changes mostly for the cultivation of typical local crops, such as olives and grapevines in the Mediterranean area. On the other hand, longer growing seasons – favouring crop yields – are recorded particularly in Northern and Eastern Europe, as a consequence of increased temperatures (EEA, 2019).

As discussed in section 1.2.2, while negative but also positive changes are projected with a changing climate, the effects of EWD are invariably negative. Most recently, the 2018 heatwave and drought lead to overall cereal production 8% lower than the previous five-year average (DG AGR, 2018), which lead to crop production losses, fodder shortages for livestock, and triggered sharp commodity price increases (up to 48%) (DG AGR, 2018; EC, 2018; EM-DAT, 2018; Hanel *et al.*, 2018). In the EU, droughts and heatwaves are projected to become more frequent and intense (IPCC, 2019), and crop yields are therefore expected to increasingly vary from year to year. This may increase the sector's vulnerability to further climate impacts, particularly without adaptation (IPCC, 2014).

On the other hand, especially for crops that are not naturally grown in the EU, there is a high dependence of food imports from the developing world, particularly from regions considered highly vulnerable to climate change (EU, 2018; Hanks and Craeynest, 2014; IPCC, 2014). With the expectation of EWD becoming more frequent and intense with a changing climate (IPCC, 2019), a cascade of climate impacts outside Europe may affect the price, quantity, and quality of products, and consequently trade patterns, which in turn may affect the food sector's income in Europe (EEA, 2019).

The *EU Strategy on Adaptation to Climate Change* and the CAP, in particular the new proposal 2021-2027, represent two main policy groups in the agriculture sector that encourage the implementation of climate adaptation measures in Member States (EEA, 2019; European Commission, 2017). In order to specifically define and implement a set of adaptation measures there is, however, the need to better understand the consequences of EWD in the EU agriculture, and in the trade of agricultural commodities. Such information is also of relevance to understand if and how trade policies should support climate adaptation strategies and actions (EEA, 2019). In particular, how historical impacts of EWD affected the production of different crops and in different European regions remains insufficiently understood. How EWD affecting crop yields in Non-European countries will have effects in the EU remains unclear. This thesis contributes to close these research gaps.

1.3 Research questions

The general goal of this research is to contribute to a better understanding of the effects of EWD on food production and on its implications through trade dependences, while taking advantage of observational records. To this end, the European Union (EU) was selected as a case study due to its profound connections within the global food market. It is recognized that the EU has been, and is foreseen to be, negatively affected by average climate change. It remains, however, insufficiently understood how the historical impacts from EWD affected the EU in terms of its own food production and also through its import dependences. This work provides contributions to answer to the following research questions (RQ):

RQ#1: What is the exposure of the EU food imports to extreme weather disasters?

RQ#2: What are the impacts of extreme weather disasters in the EU food production?

RQ#3: What are the trends on EU crop losses during extreme weather disasters years?

1.4 Research design and structure of the dissertation

In order to answer to the research questions identified in Section 1.3, two original research studies were developed, in addition to a literature review. The research questions addressed in each chapter, and the main methodologies used, are displayed in Table 1.2.

Table 1.2 Research design, including the general goal of this thesis, the research questions that are addressed in			
each chapter, and the main methodological approaches			

To contribute to a better understanding of the effects of extreme weather disasters on food production

goal	al and on its implications through trade dependences, while taking advantage of observatio		
Chapter	Title	Research questions	Methods
Chapter #2	Review on the climate change impacts on food availability and access	What are the current and future hotspots of climate change impacts on food production? (context question)	Systematic literature review of scientific literature
Chapter #3	Exposure of the EU food imports to extreme weather disasters in exporting countries	RQ#1: What is the exposure of the EU food imports to extreme weather disasters?	Time series statistical analysis based on a compositing approach (i.e. superposed epoch analysis)
Chapter #4	Drought and heatwave crop losses tripled over the last five decades in Europe	RQ#2: What are the impacts of extreme weather disasters in the EU food production? RQ#3: What are the trends on EU crop losses during extreme weather disasters years?	Time series statistical analysis based on compositing approach and on normalised anomalies

General

Chapter 1 introduces the dissertation, by presenting the relevance of the study and the research background while explaining the scope and the research questions directing the work.

Chapter 2 consists in a systematic analysis of peer-reviewed literature about climate change impacts on food availability and access. This research study is performed for a deeper understanding of the average climate change impacts on food supply. It identifies the hotspots of the food system exposure to current and future long-term average climate, while considering the effect of adaptation. It contributes for the selection of a case study to answer the identified research questions – the European Union, an exposed region considering the climate change impacts on the food system. This review study is under review in a peer review journal (October 2020).

Chapter 3 highlights the Extreme Weather Disasters (EWD) impacts on specific crops in export-oriented countries by using a compositing approach. It presents the larger implications of such impacts through trade dependencies, based on the import share per external supplier country. The focus is on the EU agri-food sector, for which its external dependency is mapped, and its potential exposure to EWD is assessed. This work contributes to answering RQ#1 and is published in *Food Security* (2019).

Chapter 4 provides important, new information on how historical EWD affected crop production in Europe, being the analysis stratified for different crops, time periods and bioclimatic zones. Averaged event impacts are quantified by using a compositing approach based on observational crop and EWD data. The severity of the events over time is evaluated by assessing the normalised crop production anomalies. While the answer to RQ#2 quantifies the impact (e.g. the degree of change in crop production) due to EWD, the answer to RQ#3 evaluates the trend of such impacts over time. This work is under review in *Environmental Research Letters* (October 2020).

Chapter 5 contains a general discussion that summarizes the key contributions of this thesis along with the answers to each research question. This chapter also discusses contributions at the policy level – on the food supply and consumption sides – with potential to overcome the exposure to EWD. This chapter also includes final remarks and future research, and the outputs that resulted from this work.

1.5 References

- 1. Alexandratos, N., Bruinsma, J., 2012. World agriculture towards 2030/2050: the 2012 revision. Rome, Italy.
- Amundson, R., Berhe, A.A., Hopmans, J.W., Olson, C., Sztein, A.E., Sparks, D.L., 2015. Soil and human security in the 21st century. Science 348, 1261071. https://doi.org/10.1126/science.1261071
- 3. Benton, T.G., 2019. Using scenario analyses to address the future of food. EFSA Journal 17, e170703. https://doi.org/10.2903/j.efsa.2019.e170703

- Berkhout, P., Achterbosch, T., Berkum, S. van, Dagevos, H., Dengerink, J., Duijn, A.P. van, Terluin, I., 2018. Global implications of the European Food System: A food systems approach. Wageningen Economic Research, Wageningen. https://doi.org/10.18174/448884
- 5. Christian, D., 2018. Origin Story A Big History of Everything, First edit. ed. Editorial Presença, Lisbon, Portugal.
- 6. Christidis, N., Jones, G.S., Stott, P.A., 2015. Dramatically increasing chance of extremely hot summers since the 2003 European heatwave. Nature Climate Change 5, 46–50. https://doi.org/10.1038/nclimate2468
- Ciscar, J.C., Ibarreta, D., Soria, A., Dosio, A., A.Toreti, Ceglar, A., Fumagalli, D., Dentener, F., Lecerf, R., Zucchini, A., Panarello, L., Niemeyer, S., Pérez-Domínguez, I., Fellmann, T., Kitous, A., J.Després, Christodoulou, A., Demirel, H., Alfieri, L., Dottori, F., Vousdoukas, M.I., Mentaschi, L., Voukouvalas, E., C.Cammalleri, Barbosa, P., Micale, F., Vogt, J.V., Barredo, J.I., Caudullo, G., Mauri, A., Rigo, D. de, Libertà, G., Durrant, T.H., Vivancos, T.A., San-Miguel-Ayanz, J., Gosling, S.N., Zaherpour, J., Roo, A. De, Bisselink, B., J.Bernhard, L., B., Rozsai, M., Szewczyk, W., Mongelli, I., Feyen, L., 2018. Climate impacts in Europe: Final report of the JRC PESETA III project, EUR 29427 EN. Publications Office of the European Union, Luxembourg. https://doi.org/10.2760/93257
- 8. Committee on World Food Security, 2015. Framework for action for food security and nutrition in protracted crises.
- Cook, B.I., Anchukaitis, K.J., Kaplan, J.O., Puma, M.J., Kelley, M., Gueyffier, D., 2012. Pre-Columbian deforestation as an amplifier of drought in Mesoamerica. Geophysical Research Letters 39. https://doi.org/10.1029/2012GL052565
- 10. Dalin, C., Wada, Y., Kastner, T., Puma, M.J., 2017. Groundwater depletion embedded in international food trade. Nature 543, 700.
- Deryng, D., Conway, D., Ramankutty, N., Price, J., Warren, R., 2014. Global crop yield response to extreme heat stress under multiple climate change futures. Environmental Research Letters 9, 034011. https://doi.org/10.1088/1748-9326/9/3/034011
- 12. DG AGR, 2018. Short-term outlook for EU agricultural markets in 2018 and 2019 (No. 22). Brussels.
- Duarte Santos, F., 2012. Alterações Globais: os desafios e os riscos presentes e futuros, First Edit. ed. Fundação Francisco Manuel dos Santos, Lisbon, Portugal.
- EC, 2018. COMMODITY PRICE DASHBOARD [WWW Document]. August 2018 edition. URL https://geotee.gr/lnkFiles/20181009134118_4.pdf (accessed 4.29.20).
- 15. EEA, 2019. Climate change adaptation in the agriculture sector in Europe. Luxembourg. https://doi.org/10.2800/537176
- Elliott, J., Müller, C., Deryng, D., Chryssanthacopoulos, J., Boote, K.J., Büchner, M., Foster, I., Glotter, M., Heinke, J., Iizumi, T., Izaurralde, R.C., Mueller, N.D., Ray, D.P., Rosenzweig, C., Ruane, A.C., Sheffield, J., 2015. The Global Gridded Crop Model Intercomparison: data and modeling protocols for Phase 1 (v1.0). Geoscientific model development 8, 261–277. https://doi.org/10.5194/gmd-8-261-2015
- 17. EM-DAT, 2018. The Emergency Events Database Universit'e catholique de Louvain (UCL) -CRED, D. Guha-Sapir. Brussels, Belgium [WWW Document]. URL https://www.emdat.be/index.php (accessed 1.5.19).
- EU, 2018. Monitoring Agri-trade Policy. MAP 2018–1. Agri-food trade in 2017: another record year for EU agrifood trade.
- European Commission, 2020. Cereals, oilseeds, protein crops and rice [WWW Document]. URL https://ec.europa.eu/info/food-farming-fisheries/plants-and-plant-products/plant-products/cereals_en (accessed 11.11.20).
- 20. European Commission, 2019. Agriculture and rural developmen Trade and international policy analysis [WWW Document]. URL https://ec.europa.eu/agriculture/trade-analysis_en (accessed 5.24.19).
- 21. European Commission, 2017. The European Union Explained: Agriculture A partnership between Europen and farmers. https://doi.org/10.2775/45239
- European Commission, 2009. REGIONS 2020 -THE CLIMATE CHANGE CHALLENGE FOR EUROPEAN REGIONS. Brussels.
- 23. EUROSTAT, 2019. Agri-environmental indicator irrigation [WWW Document]. eurostat Statistics Explained. URL https://ec.europa.eu/eurostat/statistics-explained/index.php/Agri-environmental_indicator_-_irrigation
- 24. EUROSTAT, 2016. EU trade since 1988 by HS2-HS4.
- 25. FAO, 2019a. The State of the World's Biodiversity for Food and Agriculture. FAO Commission on Genetic Resources for Food and Agriculture Assessments, Rome, Italy.
- FAO, 2019b. The State of the World's Biodiversity for Food and Agriculture, J. Bélanger & D. Pilling (eds.). Rome, Italy. https://doi.org/CC BY-NC-SA 3.0 IGO
- 27. FAO, 2019c. FAOSTAT [WWW Document]. URL http://www.fao.org/faostat/en/ (accessed 7.30.19).
- FAO, 2017. FAOstat, Food and Agricultural Organization [WWW Document]. URL http://www.fao.org/faostat/en/ (accessed 1.5.19).
- FAO, 2016a. Greenhouse gas emissions from agriculture, forestry and other land use. FAO'S work on climate change GHG emissions. Rome, Italy. https://doi.org/I6340En/1/10.16
- FAO, 2016b. AQUASTAT Main Database. Food and Agriculture Organization of the United Nations (FAO) [WWW Document].
- 31. FAO, 2015. The Impact of Natural Hazards and Disasters on Agriculture and Food and Nutrition Security A Call for Action to Build Resilient Livelihoods. Rome, Italy.
- 32. FAO, 2011a. Global food losses and food waste.
- 33. FAO, 2011b. ENERGY-SMART FOOD FOR PEOPLE AND CLIMATE Issue Paper. Rome, Italy.
- 34. FAO, 2009. High food prices and the food crisis experiences and lessons learned. Rome, Italy.
- 35. FAO, 1997. The State of the World's Plant Genetic Resources for Food and Agriculture.
- FAO, IFAD, IMF, OECD, UNCTAD, WFP, Bank, T.W., the WTO, I. and the U.H., 2011. Price Volatility in Food and Agricultural Markets: Policy Responses.
- FAO, IFAD, UNICEF, WFP, WHO, 2018. The State of Food Security and Nutrition in the World 2018. Building climate resilience for food security and nutrition. Licence: CC BY-NC-SA 3.0 IGO. Rome, Italy.
- Frieler, K., Schauberger, B., Arneth, A., Balkovič, J., Chryssanthacopoulos, J., Deryng, D., Elliott, J., Folberth, C., Khabarov, N., Müller, C., Olin, S., Pugh, T.A.M., Schaphoff, S., Schewe, J., Schmid, E., Warszawski, L., Levermann, A., 2017. Understanding the weather signal in national crop-yield variability. Earth's Future 5, 605– 616. https://doi.org/10.1002/2016EF000525
- Global Food Security Programme UK, 2015a. Extreme weather and resilience of the global food system (2015). Final Project Report from the UK–US Taskforce on Extreme Weather and Global Food System Resilience. UK.
- 40. Global Food Security Programme UK, 2015b. Climate and Global Crop Production Shocks (2015). Final Project Report from the UK–US Taskforce on Extreme Weather and Global Food System Resilience. UK.
- Haile, M.G., Wossen, T., Tesfaye, K., von Braun, J., 2017. Impact of Climate Change, Weather Extremes, and Price Risk on Global Food Supply. Economics of Disasters and Climate Change 1, 55–75. https://doi.org/10.1007/s41885-017-0005-2

- Hanel, M., Rakovec, O., Markonis, Y., Máca, P., Samaniego, L., Kyselý, J., Kumar, R., 2018. Revisiting the recent European droughts from a long-term perspective. Scientific Reports 8, 9499. https://doi.org/10.1038/s41598-018-27464-4
- 43. Hanks, K., Craeynest, L., 2014. The EU'S 2030 Energy and Climate Change Package Fit for a food and energysecure world? Oxfam International.
- 44. Harari, Y.N., 2015. Sapiens: A Brief History of Humankind. Vintage, London, IK.
- 45. HLPE, 2015. Water for food security and nutrition High Level Panel of Experts, Comiittee on World Food Security. Rome, Italy.
- 46. Holden, N.M., White, E.P., Lange, M.C., Oldfield, T.L., 2018. Review of the sustainability of food systems and transition using the Internet of Food. npj Science of Food 2, 18. https://doi.org/10.1038/s41538-018-0027-3
- Hov, Ø., Cubasch, U., Fischer, E., Höppe, P., Iversen, T., Gunnar Kvamstø, N., Kundzewicz W., Z., Rezacova, D., Rios, D., Duarte Santos, F., Schädler, B., Veisz, O.B., Zer, U., 2013. Extreme Weather Events in Europe: preparing for climate change adaptation. Oslo.
- Iglesias, A., Garrote, L., 2015. Adaptation strategies for agricultural water management under climate change in Europe. Agricultural Water ManagementManagement 155, 113–124. https://doi.org/10.1016/j.agwat.2015.03.014
- International Olive Council, 2018. World Olive Oil Figures [WWW Document]. URL https://www.internationaloliveoil.org/wp-content/uploads/2020/04/HO-W901-29-11-2019-P.pdf (accessed 4.24.18).
- IPCC, 2019. Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems [P.R. Shukla, J. Skea, E. Calvo Buendia, V. Masson-Delmot.
- IPCC, 2014. Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part B: Regional Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Barros, V.R., C.B. Field, D.J. Dokken, M.D. Mastrandre. Cambridge, United Kingdom and New York, NY, USA.
- IPCC, 2013. Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Stocker, T.F., D. Qin, G.-K. Plattner, M. Tignor, S.K. Allen, J. Boschung, A. Nauels, Y. Xia,. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA,.
- IPCC, 2012. Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation. A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change [Field, C.B., V. Barros, T.F. Stocker, D. Qin, D.J. Dokken, K.L. Ebi, M.D. Cambridge, UK, and New York.
- 54. Jägermeyr, J., 2020. Agriculture's Historic Twin-Challenge Toward Sustainable Water Use and Food Supply for All . Frontiers in Sustainable Food Systems .
- Jägermeyr, J., Frieler, K., 2018. Spatial variations in cultivars pivotal to understand global fluctuations in maize and wheat yields. Science Advances 1–11. https://doi.org/10.1126/sciadv.aat4517
- Jägermeyr, J., Gerten, D., Heinke, J., Schaphoff, S., Kummu, M., Lucht, W., 2015. Water savings potentials of irrigation systems: global simulation of processes and linkages. Hydrology and Earth System Science 19, 3073– 3091. https://doi.org/10.5194/hess-19-3073-2015
- Jägermeyr, J., Pastor, A., Biemans, H., Gerten, D., 2017. Reconciling irrigated food production with environmental flows for Sustainable Development Goals implementation. Nature communications 8:15900. https://doi.org/10.1038/ncomms15900
- 58. Knox, J., Daccache, A., Hess, T., Haro, D., 2016. Meta-analysis of climate impacts and uncertainty on crop yields in Europe. Environmental Research Letters 11, 113004. https://doi.org/10.1088/1748-9326/11/11/113004

- 59. Lang, T., Ingram, J., 2013. Food security twists and turns, in: O'Riordan, T., Lenton, T. (Eds.), Addressing Tipping Points for a Precarious Future. Oxford University Press, Oxford.
- Lesk, C., Rowhani, P., Ramankutty, N., 2016. Influence of extreme weather disasters on global crop production. Nature 529, 84–87. https://doi.org/10.1038/nature16467
- Lin, B.B., 2011. Resilience in Agriculture through Crop Diversification: Adaptive Management for Environmental Change. BioScience 61, 183–193. https://doi.org/10.1525/bio.2011.61.3.4
- Lüttger, A.B., Feike, T., 2018. Development of heat and drought related extreme weather events and their effect on winter wheat yields in Germany. Theoretical and Applied Climatology 132, 15–29. https://doi.org/10.1007/s00704-017-2076-y
- 63. Mellilo, E.D., 2012. The First Green Revolution: Debt Peonage and the Making of the Nitrogen Fertilizer Trade, 1840–1930. The American Historical Review 117, 1028–1060. https://doi.org/https://doi.org/10.1093/ahr/117.4.1028
- 64. Min, S., Zhang, X., Zwiers, F.W., Hegerl, G.C., 2011. Human contribution to more-intense precipitation extremes. Nature 5–9. https://doi.org/10.1038/nature09763
- Moore, F.C., Lobell, D.B., 2015. The fingerprint of climate trends on European crop yields. PNAS 112, 2670– 2675. https://doi.org/10.1073/pnas.1409606112
- Mueller, B., Hauser, M., Iles, C., Rimi, R.H., Zwiers, F.W., Wan, H., 2015. Lengthening of the growing season in wheat and maize producing regions. Weather and Climate Extremes 9, 47–56. https://doi.org/https://doi.org/10.1016/j.wace.2015.04.001
- Müller, C., Elliott, Joshua Chryssanthacopoulos, J., Arneth, A., Balkovic5, J., Ciais, P. *et al*, 2017. Global gridded crop model evaluation: benchmarking, skills, deficiencies and implications. Geoscientific model development 10, 1403–1422. https://doi.org/10.5194/gmd-10-1403-2017
- Nelson, G.C., Valin, H., Sands, R.D., Havlík, P., Ahammad, H., Deryng, D., Elliott, J., 2014. Climate change effects on agriculture : Economic responses to biophysical shocks. PNAS 111. https://doi.org/10.1073/pnas.1222465110
- NOAA, C. go., 2020. Climate Change: Atmospheric Carbon Dioxide [WWW Document]. URL https://www.climate.gov/news-features/understanding-climate/climate-change-atmospheric-carbon-dioxide (accessed 11.4.20).
- Piesse, J., Thirtle, C., 2009. Three bubbles and a panic: An explanatory review of recent food commodity price events. Food Policy 34, 119–129. https://doi.org/10.1016/j.foodpol.2009.01.001
- 71. Porter, J.R., Xie, L., Challinor, A.J., Cochrane, K., Howden, S.M., Iqbal, M.M., Lobell, D.B., Travasso, M.I., 2014. Food security and food production systems". In: Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Cha. United Kingdom; New York, NY, USA.
- Powell, J.P., Reinhard, S., 2016. Measuring the effects of extreme weather events on yields. Weather and Climate Extremes 12, 69–79. https://doi.org/http://dx.doi.org/10.1016/j.wace.2016.02.003
- 73. Puma, M.J., Bose, S., Chon, S.Y., Cook, B.I., 2015. Assessing the evolving fragility of the global food system. Environmental Research Letters 10. https://doi.org/10.1088/1748-9326/10/2/024007
- Puma, M.J., Chon, S.Y., Kakinuma, K., Kummu, M., Muttarak, R., Seager, R., Wada, Y., 2018. A developing food crisis and potential refugee movements. Nature Sustainability 1, 380–382. https://doi.org/10.1038/s41893-018-0123-z
- 75. Ray, D.K., Gerber, J.S., MacDonald, G.K., West, P.C., 2015. Climate variation explains a third of global crop yield variability. Nature Communications 6, 5989. https://doi.org/10.1038/ncomms6989

- 76. Ray, D.K., West, P.C., Clark, M., Gerber, J.S., Prishchepov, A. V, Chatterjee, S., 2019. Climate change has likely already affected global food production. PLOS ONE 14, e0217148.
- Roberts, M.J., Braun, N.O., Sinclair, T.R., Lobell, D.B., Schlenker, W., 2017. Comparing and combining processbased crop models and statistical models with some implications for climate change. Environmental Research Letters 12, 95010. https://doi.org/10.1088/1748-9326/aa7f33
- Rockström, J., Steffen, W., Noone, K., Persson, Å., Chapin, F.S., Lambin, E.F., Lenton, T.M., Scheffer, M., Folke, C., Schellnhuber, H.J., Nykvist, B., de Wit, C.A., Hughes, T., van der Leeuw, S., Rodhe, H., Sörlin, S., Snyder, P.K., Costanza, R., Svedin, U., Falkenmark, M., Karlberg, L., Corell, R.W., Fabry, V.J., Hansen, J., Walker, B., Liverman, D., Richardson, K., Crutzen, P., Foley, J.A., 2009. A safe operating space for humanity. Nature 461, 472–475. https://doi.org/10.1038/461472a
- Rosenzweig, C., Elliott, J., Deryng, D., Ruane, A.C., Müller, C., Arneth, A., 2014a. Assessing agricultural risks of climate change in the 21st century in a global gridded crop model intercomparison. PNAS 111, 4–9. https://doi.org/10.1073/pnas.1222463110
- Rosenzweig, C., Iglesias, A., Yang, X.B., Epstein, P.R., Chivian, E., 2001. Climate change and extreme weather events - Implications for food production, plant diseases, and pests. GLOBAL CHANGE & HUMAN HEALTH, NASA Publications 2, 90–104. https://doi.org/10.1023/A:1015086831467
- Rosenzweig, C., Jones, J.W., Hatfield, J.L., Ruane, A., Thornburn, K.J., Antle, J.M., Nelson, G.C., Porter, C., Janssen, S., Basso, B., Ewert, F., Wallach, D., Baigorria, G., Winter, J.M., 2014b. The Agricultural Model Intercomparison and Improvement Project (AgMIP): Protocols and pilot studies 35. https://doi.org/10.1016/j.agrformet.2012.09.011
- 82. Santos, J.L., 2010. The Environmental Crisis and the Future of Agriculture, in: Environment at the Crossroads Aiming for a Sustainable Future. Carcanet Press Ltd., pp. 61–79.
- Siebert, S., Döll, P., 2010. Quantifying blue and green virtual water contents in global crop production as well as potential production losses without irrigation. Journal of Hydrology 384, 198–217. https://doi.org/https://doi.org/10.1016/j.jhydrol.2009.07.031
- 84. Simmons, A.H., 2010. The neolithic revolution in the Near East : transforming the human landscape / Alan H. Simmons ; with a foreword by Ofer Bar-Yosef. The University of Arizona Press.
- 85. Tai, A.P.K., Martin, M.V., Heald, C.L., 2014. Threat to future global food security from climate change and ozone air pollution. Nature Climate Change 4, 817–821. https://doi.org/10.1038/nclimate2317
- 86. The Government Office for Science, 2011. The Future of Food and Farming: Challenges and choices for global sustainability Final Project Report. London,UK.
- 87. The World Bank, 2019. DataBank [WWW Document]. URL https://databank.worldbank.org/data/home.aspx (accessed 5.21.19).
- 88. Timmer, C.P., 2005. Food Security and Economic Growth: an Asian perspective. Asian-Pacific Economic Literature 19, 1–17. https://doi.org/10.1111/j.1467-8411.2005.00155.x
- Torero, M., 2016. Alternative Mechanisms to Reduce Food Price Volatility and Price Spikes: Policy Responses at the Global Level BT - Food Price Volatility and Its Implications for Food Security and Policy, in: Kalkuhl, M., von Braun, J., Torero, M. (Eds.), . Springer International Publishing, Cham, pp. 115–138. https://doi.org/10.1007/978-3-319-28201-5_6
- 90. Troy, T.J., Kipgen, C., Pal, I., 2015. The impact of climate extremes and irrigation on US crop yields. Environmental Research Letters 10. https://doi.org/10.1088/1748-9326/10/5/054013
- 91. UN, 2013. World Population Prospects The 2012 Revision. New York.
- 92. UNFPA, 2020. World Population Dashboard [WWW Document]. United Nations Population Fund. URL https://www.unfpa.org/data/world-population-dashboard (accessed 12.14.20).

- 93. Vermeulen, S.J., Campbell, B.M., Ingram, J.S.I., 2012. Climate Change and Food Systems. Annual Review of Environment and Resources 37, 195–222. https://doi.org/10.1146/annurev-environ-020411-130608
- Vogel, E., Donat, M.G., Alexander, L. V, Meinshausen, M., Ray, D.K., Karoly, D., Meinshausen, N., Frieler, K., 2019. The effects of climate extremes on global agricultural yields. Environmental Research Letters 14, 054010. https://doi.org/10.1088/1748-9326/ab154b
- von Braun, J., 2007. The World Food Situation: New Driving Forces and Required Actions. IFPRI's Biannual Overview of the World Food Situation presented to the CGIAR Annual General Meeting. International Food Policy Research Institute.
- 96. Wada, Y., van Beek, L.P.H., Bierkens, M.F.P., 2012. Nonsustainable groundwater sustaining irrigation: A global assessment. Water Resources Research 48. https://doi.org/10.1029/2011WR010562
- 97. Wine Institute, 2017. Wine Institute World statistics. World Wine Consumption by Country [WWW Document]. URL https://www.wineinstitute.org/resources/statistics (accessed 4.24.18).
- 98. World Health Organization, 2017. Obesity and Overweight [WWW Document]. URL https://www.who.int/newsroom/fact-sheets/detail/obesity-and-overweight (accessed 4.8.19).
- 99. Yearley, S., 2013. Full Planet, Empty Plates: The New Geopolitics of Food Scarcity, Lester R Brown, W W. Food Security 5, 285–286. https://doi.org/10.1007/s12571-013-0246-z

2 Review on the climate change impacts on food availability and access

This study is under review in the journal Global and Planetary Change:

Brás, T.A., Seixas, J. Review on the climate change impacts on food availability and access.

Author contributions:

T.B. and J.S. designed the research. T.B. compiled and analysed the data. The paper was written together with contributions from both authors.

Abstract

Climate change is affecting food security through increasing temperatures, changing precipitation patterns and higher frequency of extreme events. Here, we perform a systematic literature review on the observed and future impacts of the long-term climate trends on crop responses and on prices, considering papers published between 2013 and 2019. We also discuss the usefulness of the reviewed material for the food-climate governance schemes.

Our review shows that impacts are highly variable, and not only negative, from place to place and for different food items. However, in a medium-term future, under the higher emission scenarios and even considering crop adaptation in few countries, global breadbaskets – USA, China, Europe, Southern Asia, and Southern America – may see a decline on crop production, from -5% to -55% (depending on crop and region), with harmful consequences on the worldwide food supply and on its prices. Other countries or regions may see positive changes, especially if crop adaptation is implemented. The acknowledgement of regional climate impacts variability on production and consumption spatial patterns, along with a cross-regional analysis on water scarcity, highlights the need for a climate-food-governance towards food security for all countries.

Keywords

Climate Change; Food Security; Availability; Access; Observed and Future Impacts; World Regions

2.1 Introduction

Within the past decades, food security has remained a major global issue, especially in less developed countries (FAO, 2007; Barret *et al.*, 2010). The increased attention on the subject was particularly noticeable after the 2007–2008 and 2010 world food price crises (FAO, 2008). In 2017, 821 million people in the world

were estimated to be undernourished, a figure that is increasing in almost all regions of Africa and in South America (FAO *et al.*, 2018). An integrated analysis on food dimensions, through a set of indicators, allows to highlight the causes and consequences of food insecurity (FAO, 2013) in a country or region and may foster the definition of strategies, policies and the design of governance schemes to guaranty food security at the long-term (FAO, 2013; FAO *et al.*, 2014). Food *availability* considers the supply of food (i.e. production and imports including food aid), with adequate nutritional levels and according to cultural standards (FAO *et al.*, 2014; Fukuda-Parr and Orr, 2013), and food *accessibility* refers to the physical and economic access for people to acquire the food they need. The access dimension comprises indicators of physical access and infrastructure, as railway and road density, and economic access represented by the domestic food price index, and family income, among others. The stability of food security accounts for the risks to availability, and access from shocks, such as natural disasters, food price volatility, fluctuations in domestic food supply and political instability (FAO *et al.*, 2014; Stringer, 2016).

Food security is being undermined by a combination of threats which could be grouped (according to Lang and Ingram (2013)) in economic and territorial forces, such as inappropriate price signals, fossil fuel reliance, urbanization, globalization and armed conflicts; social forces such as population growth and demand for food, the nutrition transition and diet-based ill-health patterns, the culture of choice, and the high levels of food waste. In addition to those, climate change is also a major threat to food security and intensifies other environmental forces such as water scarcity, soil and biodiversity losses (Lang and Ingram, 2013). Climate change may disturb the stability of the food system by affecting any of the food security dimensions (Krishnamurthy *et al.*, 2015). Long-term changes in the patterns of temperature and precipitation will shift production seasons, increase the supply variability and the risks in agriculture livestock, forestry, and fisheries.

Identifying the food commodities and regions that have been and will be most affected by climate trends will contribute to build resilient food systems and support the definition of adaptation measures. These will feed food policies and governance schemes that can anticipate and better manage different types of shocks affecting the food system. This study presents a systematic and integrated literature review on the climate change impacts (observed and future projections) on food security availability and access in world regions, by analysing papers published between 2013 and 2019. We perform a critical analysis on the usefulness of the reviewed material for a food-climate governance scheme aiming to tackle and manage food security for all countries, under climate change impacts. The paper is organised around five sections: section 1 introduces the scope and goal of the paper, section 2 presents the methodology used, and section 3 systematizes the reviewed climate change impacts on food security dimensions for different world regions. Section 4 discusses the usefulness of considering climate change impacts for food governance and policy coherence, and section 5 concludes.

2.2 Methods

We perform a systematic literature review following Pickering and Byrne (2014) methodology for data search, by considering 42 studies, published between January 2013 and May 2019, with observed and projected climate change impacts on selected food items. Only original scientific papers written in English are considered for the literature review although other literature (e.g. synthesis, reports and working papers) is used to support the analysis.

The process for data collection, characterization and analysis is described in Figure 2.1. Data on climate change impacts on food availability and access is grouped according to (a) producer region: Africa (which expands to Northern Africa and Sub-Saharan Africa), Europe (i.e. European Union, but also Western and Southern Europe, and Eastern and Northern Europe), Asia (i.e. Western and Southern and Southeast Asia, and Central and Eastern Asia), America (i.e. Northern and Central America, and Caribbean and South America), and global scale; and (b) food item: "Cereals" (which expand to wheat, maize, rice, barley, sorghum) that are key to human and livestock feeding and together represent nearly 60% of the worldwide energy supply (Mouillé *et al.*, 2008)), and "Other crops" (which expand to oil crops (i.e. soybean, rapeseed, oil palm, groundnuts), roots and tuber (i.e. cassava, yam, potato), sugar crops (i.e. sugar beet and sugarcane), and coffee).

For an easier analysis of future projections, three temporal scales are considered: near-time future (NF) from 2020s-2030s (5 papers), medium-time future (MF) from 2040s-2060s (22 papers) and long-time future (LF) from 2070s-2100 (11 papers). Detailed results are presented for the MF, on section 3, since most projections among reviewed papers refers to this period. All the results identified on the reviewed papers are listed in Tables A1-A3. The usefulness of the reviewed material for food-climate governance schemes is discussed on section 4 by considering past and future climate change impacts on each food security dimension.

2.3 Reviewed climate change impacts on food security dimensions

Relevant reviews were published between 2013 and 2019 outlining major impacts of climate change on relevant food items (Sanchez *et al.*, 2014; Tripathi *et al.*, 2016), mostly focused on observed (Zinyengere *et al.*, 2013) and projected impacts and uncertainties (Rötter *et al.*, 2014). Recent developments on the use of multi-model ensembles in climate change impacts on crop diseases were reviewed (Newbery *et al.*, 2016), as well as, on the effect of adaptation measures in yields by using ensemble and climate modelling and observed data (Challinor *et al.*, 2014, 2013). Zhao *et al* (2017) investigates the impacts of temperature on yields of grain crops by compiling published results from different analytical methods (i.e. global grid-based and local point-based models, statistical regressions, and field-warming experiments). All these published reviews focus in a specific dimension of food security, while here we perform an integrated overview by

reviewing the available scientific material regarding past and future climate change impacts on food availability and access. While the recent Intergovernmental Panel on Climate Change report (2019) reviews relevant literature on past and future climate change impacts on food security dimensions, we enrich the assessment by also including results on the projected impacts with and without crop adaptation measures, across spatial scales, and in particular for the world breadbaskets. In addition, this manuscript discusses the usefulness of considering climate change impacts for food governance and policy coherence.



Figure 2.1 Methodology for a systemic literature review.

Data collection through systematic methodology (following Pickering and Byrne (2014)) and data characterization and analysis according to the number of studies (n) on observed (blue) and future (grey) climate change impacts on food availability and access. These impacts are analysed for the indicators (ind.) of each food security dimension, such as yield and production (for availability), price and prevalence of undernourishment (for access). The published literature considered in this review is indicated in Tables A1-A3.

The next sections present the observed and future climate change impacts, organised per food security dimension, namely on availability and access. As explained in section 2, results on the future projected impacts are presented only for the medium-term future (MF) from 2040s-2060s, but the results for all three periods are indicated in Appendix A.

2.3.1 Impacts on food availability

2.3.1.1 Observed impacts

Observed climate change impacts on food availability are based in 14 papers, with considerable results for worldwide regions taken from Ray *et al* (2019). Nine of the papers are relative to Asia (mostly China), five to America, three to Europe and one to Africa (Fig. 2.1, Table A1).

Global climate change impacts are estimated to have affected yields (or production) mostly for wheat, maize, rice, barley, sorghum, soybean, oil palm, rapeseed, cassava and sugarcane, with losses ranging from - 24.5% (i.e. maize in Eastern and Northern Europe) to -0.1% (i.e. rice in Northern and Central America), and gains ranging from +0.2% (i.e. soybean in Central Eastern Asia) to 24.9% (i.e. rapeseed in Sub-Saharan Africa) (Fig. 2.2, Table A1). Most of the historical crop losses are estimated in European and Sub-Saharan African countries, gains are estimated in Latin American countries, and mixed impacts in Asia and Northern and Central America (Ray *et al.*, 2019). Worldwide, for each Celsius degree in global mean temperature, yields are estimated to decrease, on average, by 7.4% for maize, 6.0% for wheat, 3.2% for rice, and 3.1% for soybean (Asseng *et al.*, 2015; Zhao *et al.*, 2017).

Asia (in particular, China) is the most studied region and where climate change impact on crop yield is varied. Overall, even though in Central and Eastern Asia (CEA) and in the Western and Southern and Southeast Asia (WSSA) there are, respectively, yield gains up to 5.9% and 1.9% (both for rapeseed), most of climate change impacts are associated to crop losses: in CEA up to -12% for maize and -4.8% for rice, and in WSSA up to -15.9% for oil palm (Ray *et al.*, 2019; Wang *et al.*, 2016; Wei *et al.*, 2014).

In China, mean climate changes overall benefited yields by 2% across ten crops - barley, cassava, maize, rice, oil palm, rapeseed, sorghum, sugarcane, soybean and wheat – although there are exceptions in a few provinces with rice and wheat yield declines (Ray *et al.*, 2019). Such variance in results has also been reported in other studies: Chen, Zhou and Zhou (2014) estimated a decrease in wheat production attributed to an average temperature increase and to a decrease on diurnal temperature range (at a national scale). However, increasing temperature trends leading to wheat yield increases is reported by Wang *et al* (2014), who considered aggregated provincial data of China, highlighting the implications of the inconstant elasticities for crop yield with variations of climate variables. A decline in rice yield has been attributed to a warming trend, which has been pointed to offset or even reverse the positive effect of CO₂ enrichment in rice yields (Wang *et al.*, 2016), and to a shortened growth duration (0.15 - 0.27 day/y) (Wang *et al.*, 2014).

On the other hand, a positive effect on rice yield in China due to an increase of the minimum daily temperature was observed by Zhou *et al.* (2013). In that study, the contribution of climate change explained nearly 7% of rice yields increase, while non-climatic factors were major drivers. In addition, the implementation of adaptation measures to warming have greatly facilitated rice production (e.g. breeding new varieties and adapting cold-resistant cultivars) (Zhou *et al.*, 2013). It was also reported that precipitation change has not significantly been attributed to variations in rice yield (Wei *et al.* 2014, Chen *et al.* 2014). Also in China, a declining in maize yield has been attributed to a warming trend, although a decrease in diurnal temperature range has been associated to an increase in maize production at a national scale (Bu *et al.*, 2015; Chen *et al.*, 2014; Wei *et al.*, 2014; Zhao *et al.*, 2015). Regarding other Asian countries, Ray *et al* (2019) estimated production and yield losses in rice across India, Vietnam and Philippines, and losses in wheat production in India and Turkey.

In Europe, most of climate change impacts are associated to crop losses. In the Eastern and Northern Europe (ENE) yield losses are widespread for maize (-24.5%), sorghum (-9.5%), barley (-9.1%), wheat (-2.1%), and in a less extend for rice (-0.4%). However, averaged gains are estimated for rapeseed (3.1%) and sugarcane (2.7%) (Ray *et al.*, 2019). More specifically, Moore and Lobell (2015) attribute to the long-term temperature and precipitation trend losses on barley in the United Kingdom (UK) (-2%) and in Ireland (-1%) but also gains for wheat (4 and 9%, respectively), and for sugar beet yields in Denmark and UK (3 and 4%, respectively).

In Western and Southern Europe (WSE) averaged yields decreased between -3 to -6% for rice and maize, -8 to -15% for barley, rapeseed and wheat, and -18 to -21% for sorghum and soybean, whereas yield increased by 2.7% for sugarcane (Moore and Lobell, 2015; Ray *et al.*, 2019). Moore and Lobell (2015) estimate yield losses in the Mediterranean countries for wheat from -15 in Italy to -2% in Portugal, barley from -8 in Greece to -4.5% in Portugal, maize from -8 in Italy to -1% in France but with gains in Greece (+9%). Losses in sugar beet yields are also estimated from -12.5 in Italy to -10% in Greece, while gains of 1.5% are estimated in France. That study also refers that climate trends explain nearly 10% of the stagnation in European wheat and barley yields, being the changes in agriculture subsidies and environmental policies as likely explanations for a declining in yield growth.

A decrease in yields of major crops – wheat, barley, maize, and rapeseed – is estimated for parts of the steppe region in Russia and in the grain belt of Western Siberia. In Ukraine yield of barley, maize and sorghum have been negatively affected (Ray *et al.*, 2019).

In Northern Africa (NAf), climate change benefited cassava (18%) and sorghum yields (17.9%), in addition to wheat (12%) and soybean (10.9%). But in a less extend, gains of a heat- and drought-tolerant sorghum (0.7%) and cassava (1.7%) are estimated in Sub-Saharan Africa (SSA), and in a higher extend of rapeseed

(24.9%). On the other hand, in both regions, losses are estimated for maize (-4.3 in Naf, -5.8% SSA), rice (-1.3%, -3.1), barley (-6.8, -0.6%) and sugarcane (-5.1, -3.9%) (Ray *et al.*, 2019).

In Northern and Central America (NCA), climate change impacts on crop yields are mixed: gains are estimated mostly for sorghum (4.3%), soybean (3.3%), sugarcane (1.7%), and slightly for maize (0.5%), while losses are observed mainly for oil palm (-7.2%), cassava (-2.9%), and barley (-2.5%) (Ray *et al.*, 2019). In the USA, barley, rice and wheat yields declined whereas maize, sorghum, soybean and sugarcane yields increased (Ray *et al.*, 2019). In Mexico, a gain in wheat yields by 19.6% is associated to a CO₂ increase. As in the NCA, also in the Caribbean and South America (CSA), yield gains are estimated for maize, soybean, and sugarcane (all around 3%), while losses of rice (-0.7) and wheat (-1.6%), and mostly of oil palm (-7.2%) are observed. In Central America and Colombia, Avelino *et al.* (2015) reported the impact of temperature range decrease on coffee production due to coffee rust epidemics (caused by a fungus), with impacts on local profitability, which constrained food access, since coffee is often the only source of income to buy food and supplies for grain cultivation.

2.3.1.2 Projected impacts

Projections on food availability (Fig. 2.3, Table A2) analysed by 30 papers, highlight that 80% of the number of assessments of worldwide crop yield change is negative in 2050s (in comparison to a baseline ranging from 1961 to 2010, according to the study considered), while only 20% is positive. A few of these positive projections include adaptation measures, the effect of CO₂ fertilisation or the impacts of ozone (O₃) pollution.

Projections for a medium-term future (MF) show that impacts on yield will be mostly negative across world regions – especially for cereals, soybean, and roots and tubers (i.e. cassava and yam in Naf) (Fig. 2.4). There are, however, projected yield gains even though most of them are associated to the adaptation measures or to the effect of higher concentrations of CO_2 and O_3 .

The effect of higher CO₂ concentration remains one of the largest uncertainties of the climate change impacts on agriculture. In theory, and especially for C3 crops (such as wheat, rice, soybeans, and trees), higher CO₂ concentration in the atmosphere has the potential to increase photosynthesis and water productivity of plants, thus reducing crop water requirements. However, the effect of CO₂ can be offset by higher temperatures and altered precipitation patterns and varies according to the crop type (i.e. C3 or C4 (such as maize and sugarcane)) (Fader *et al.*, 2015). Even though there are uncertainties about how climate and O₃ pollution interact to affect agriculture, Tai, Martin and Heald, (2014) found that O₃ trends can exacerbate but also offset significantly climate impacts, depending on the scenario, thus suggesting the importance of air quality management in agricultural planning.

In a few SSA countries, maize yields are projected to decline, on average by -6% (RCP8.5) but to increase up to 24% (RCP8.5) if a drought tolerant variety is considered. Without adaptation, yield changes could be

even more negative in Burkina Faso (-8%, RCP8.5) (Waongo *et al.*, 2015), Nigeria (up to -30%, A1B) even by considering the effect of CO₂ fertilisation (Mereu *et al.*, 2015), or in Gambia (up to -40%, RCP8.5) (Ahmed *et al.*, 2015). Maize yields are, however, projected to increase in South Africa (from 5 to 25% depending on the climate scenario and impact model used) (Dube *et al.*, 2013), in Guinea-Bissau (8.9%, RCP8.5) (Ahmed *et al.*, 2015), or in Ethiopia (up to 84%, RCP8.5) but only by considering fertilizer application (Kassie *et al.*, 2015). Across SSA countries, sorghum and millet yields are projected to decline from -45.5% to -4% (RCP8.5), even by considering drought and heat tolerant varieties (Islam *et al.*, 2016). The effect of adaptation is also studied by Srivastava, Gaiser and Ewert (2015) in the Republic of Benin, where yam yields are projected to decline by -30% (A1B or B1) but to increase from 7 to 49% (A1B), depending on the implementation of fertilizer application, irrigation, or late maturing cultivar. According to the climate scenario, but without considering adaptation measures, yields of soybean are projected to rise in South Africa by up to 20%.

In Europe, wheat, soybean, and maize production are projected to decline, on average and respectively, by -12%, -20% and up to -40% for RCP8.5 and when the effect of O_3 pollution is considered. For RCP4.5 and with the effect of O_3 pollution, wheat and rice production are projected to increase, respectively, by 5 and 7% (Tai *et al.*, 2014). Nelson *et al* (2014) project yield declines of wheat and rice by -15%.

In Asia, losses of wheat, maize, soybean, rice, and potatoes are projected to occur under RCP8.5 scenario. However, under lower emission impact scenarios and/or under the effect of CO₂, O₃ pollution, or with the implementation of improved crop varieties, future climate impacts in countries or regions may be positive or negative. For example, in Asian countries, rice yields may decline around -5 to -55% under A2 scenario but are projected to increase around 20 to 26% when the CO₂ fertilisation effect is considered (Li *et al.*, 2015).

In China, rice yield is projected to decline by -12% under A2, -4.3% under B2 (Ju *et al.*, 2013) or -2% under RCP8.5 and with the effect of O₃ pollution (Tai *et al.*, 2014). Tai *et al* (2014) project maize and soybean production gains, respectively, by 5 and 10% under RCP4.5 but that may turn negative under RCP8.5 (by \sim -5% for both crops). Wheat production is projected to decline by nearly -15% (RCP8.5), but to increase up to 15% under RCP4.5 (Tai *et al.*, 2014) or up to 68% under the A1F1 scenario and only if the uncertain effect of CO₂ is considered (Tao and Zhang, 2013).

Wheat production is projected to decline by -35% in South Asia, and up to -10% in Southeast Asia (both with RCP8.5) (Tai *et al.*, 2014). At the country level wheat yield losses are projected, for example, in Pakistan by -18% (A1B) (Shi *et al.*, 2015; Zhu *et al.*, 2013) or, under RCP8.5: around -7% in Pakistan, Bangladesh, India, Nepal even if drought and heat tolerance varieties are considered (Islam *et al.*, 2016). In WSSA, on average and under RCP8.5 and by considering the effect of O₃ pollution, rice production is projected to decline by -1% (Tai *et al.*, 2014). Losses in rice yields are, however, projected to rise around -30% in Pakistan (A1B) or in India (A2) (Banerjee *et al.*, 2014; Zhou *et al.*, 2013).

In USA, high production losses are projected for maize (-50 to -45% under RCP4.5 and RCP8.5) and for soybean (-5 to -10%) (Tai *et al.*, 2014). Wheat yields are projected to decline up to -30% under A2 (Jiang and Koo, 2014) and production to decline up to -10% according to the climate scenario (Rosenzweig *et al.*, 2014; Tai *et al.*, 2014). However, if technological advances are considered and under A2, wheat yields may reach 60% in the country. Rice production is projected to increase, respectively, by 2 to 5% under RCP8.5 and RCP4.5 under O₃ pollution (Tai *et al.*, 2014).

In the CSA region, crop models project production losses of maize and soybean of -25%, and wheat losses of -10% under RCP8.5 under the effect of O_3 pollution (Tai *et al.*, 2014). An average decline of -20% on yields of oil seeds, wheat, and rice is projected, under RCP8.5, in Brazil by Nelson *et al* (2014).

2.3.2 Impacts on food access

2.3.2.1 Observed impacts

Due to a low number of case studies it is difficult to derive conclusions on the observed climate change impacts on the access dimension. Nevertheless, it is reported that local food supply is strongly determined by local weather, and for the cases that food market is barely connected with foreigner markets, food access becomes a serious problem. Brown and Kshirsagar (2015) found that almost 20% of local market prices (wheat, maize and rice) were affected by domestic weather disturbances in 51 developing countries. They estimated 9% of local market prices were associated with international price changes, while 4% with both domestic weather disturbances and variability of international prices affect rural economies is key to define policies to mitigate the effects of climate disturbances and prevent the lack to food access. Avelino *et al.* (2015) reported the impact of temperature range decrease on coffee production due to coffee rust epidemics (caused by a fungus) in Central America and Colombia, with impacts on local profitability, which constrained food access, since coffee is often the only source of income to buy food and supplies for grain cultivation.

2.3.2.2 Projected impacts

On average, future climate scenarios are likely to increase commodity prices and thus, may negatively impact food access, with exception of Japan (and within the analysed papers) (Figure 2.5) (Table A3). According to Nelson *et al* (2014), price of oil seeds, wheat, and rice are projected to increase, on average, by 10% in European countries and Brazil, 15% in China, USA and SSA countries, and 25% in India. They showed that a large part of climate change shock is transferred to the production-side and trade responses. With a negative productivity effect from climate change, prices increase (due to the inelastic nature of global demand) and trigger more intensive management practices, area expansion, reallocation through international trade and reduced consumption, with especially negative effects for the poor in rural areas.

			_	
	[-6.8 ; -1.3]	[10.9 ; 18.0		
HARAN	Wheat Maize Rice Barley Soybean Sug	garcane Sorghum Cassava R	apeseed	
500	[-5.8 ; -0.6]		0.7 ; 24.9]	
ALP.				
DPE	Wheat Maize Rice Barley Sorghum Soy	bean Rapeseed Sugar beet	Sugarcane Sugar beet F	Rapeseed Wheat Maiz
A	[-24.5 ; -0.4]			[1.0 ; 9.0
AL &	Wheat Maize Rice Oil palm Wheat	Maize Rice Barley Sorghum	Soybean Rapeseed Sug	arcane Cassava
I ASIA	[-12.0; -0.1]			[0.2;5.9]
DUTHERN	Wheat Rice Barley Soybean Oil palm	Sugarcane Cassava Sorghur	n Maize Rapeseed	
SI ASIA	[-15.9 ; -0.8]		[0.9 ; 1.9]	
ERN &	Wheat Rice Barley Rapeseed Oil palm	Cassava Coffee Wheat Ma	iize Sorghum Soybean	Sugarcane
AMERICA	[-16.0; -0.1]			[1.7; 19.0]
BEAN	Wheat Rice Oil palm Maize Barley So	ybean Rapeseed Sugarcane	Cassava	
MIERICA	[-1.6 ; -0.6]		[0.5;6.8]	
De la				

igure 2.2 Observed climate change mpacts on food availability.

Percentage of change of crop yield or production) due to historical emperature and precipitation trends in world regions. Crops with negative ind (or production) changes are ndicated inside the red boxes, while he positive ones are inside green boxes. The lowest and highest thange (%) per region is indicated in siguare brackets. The historical period may differ for each estimated mpact but is within the interval 1960-1014. The distribution of the impacts is based on published literature adicated in Table A1.

30



Figure 2.3 Projections on worldwide crop yield change due to climate impacts. Number of assessments (i) on future projections of worldwide crop yield percentage of change in comparison to a baseline, per temporal period. Studies reporting negative changes are 65 (A, left side) and positive changes are 16 (B, right side). Detailed results from literature review are presented in Table A2.

Mosnier *et al.* (2014) projects (under A2) an increase of world crop price index (including cereals, oil crops, sugar crops, tubers, fibers) up to 5%, which for China and Mongolia may rise up to 6 and 38%, respectively. Higher prices in Mongolia are explained by demand increase, which is concentrated on few products, and by less flexibility in trade (i.e. high transportation costs and negative climate change impacts on few trading partners may lead to higher import prices). China has more flexibility to adjust trade partners. On the contrary, for Japan it is foreseen a decrease on crop price index up to -5%, justified by a higher domestic productivity that compensate higher import prices. Zhu *et al* (2013) projected, in Pakistan, a price increase for wheat (18–27%), rice (26–32%), and maize (14–28%) (for B1 and A1) compared to a no-climate-change scenario. Pakistan is expected to become a net food importer (due to its moderate growth in agricultural production, water scarcity and population growth), which will likely be exacerbated by climate change (Zhu *et al.*, 2013). Dube *et al.* (2013) expected soybeans production to remain largely constant, in South Africa, while net imports are expected to considerably increase, leading to a commodity price rise by 60% (for averaged A1B and B1).

The prevalence of malnutrition is a result of a lack to food access (FAO *et al.*, 2014). Dawson, Perryman and Osborne (2016) predicted (under A1B) an increase of 50% or more in the population in risk of undernourishment as a result of climate change in South America and Africa, Australia and Central Asia, with some European areas, South-East Asia, USA and Russia also seeing an increase in population at risk. Hertel and Baldos (2016) (under RCP8.5) highlighted the effect of market integration (in line with environmental policies protecting sensitive lands) on attenuating the increase of undernourishment rate.

SUB-SAHARAN	Cassavaa Groundnutsz MaizeabzRiceaz Sorghumaz Milletaz Yamz Wheatz Maizeabz	E Soybeana, Pama Groundnutsa
AFRICA	[-45.5 ; -1.2]	[0.0; 84.0
	Wheat12 Mai2612 Soybean12 Wheate1 Rice12	
	[-40.0;-10.0] [0.2;12.0]	
CENTRAL &	Wheat,d.2 Maize,d.2 Rice,d.2 Soybean2 Wheat, Maize, Rice, Soybean	
EASTERN ASIA	[-15.0 ; -0.4] [2.0 ; 68]	Legend
STERN & SOUTHERN	Wheatab.1.2 Maizes.1.2 Ricea.2 Soybean2 Potato2 Wheatab.4*Rice2 *Potato2	Climate scenarios a – A1B
(SOUTHEAST ASIA	[-35.0 ; -1.0]	b – B1 c – A2
		d – B2 e – A1F1
NORTHERN & CENTRAL AMERICA	Wheatera Maizera Soybeana, Wheatact Ricea	1- RCP4.5 2- RCP8.5
No.		 No adaptation
CARIBBEAN 8. SOLITH AMEDICA	Wheat12 Maize12 Rice2 Soybean12	 Adaptation COnfortination offort
	[-25.0;-5.0]	
		*Asia

gure 2.4 Future climate change impacts food availability.

nge scenario, the simulation of ptation actions and/or the effect of O₃ inologic advances. Climate change per region are indicated in square act may vary according to the climate re). Adaptation measures include lizer application, irrigation, delayed heat tolerance, and/or other icial Report Emissions Scenarios ES: A1, A1FI, A1B, A2, B1, B2); and in IPCC Representative Concentration mple, in the Sub-Saharan region (or in nate scenarios A1B1, RCP8.5, sidering the effect of CO_2 fertilisation /or without adaptation measures. The ibution of the impacts is based on I0s-2060s), due to future climate ps with negative yield or production es. The lowest and highest changes ckets. For one crop in one region, the CO2 concentrations (see legend in nting, late maturing cultivars, drought narios are indicated in the IPCC or more countries in this region), rice Is are projected to be negative under centage of change of crop responses, ected for the medium-term period ds in world regions. The reference od varies according to the published ly but is within the interval 1961-2010. nges are indicated inside red boxes, e the positive ones are inside green 1ways (RCP: RCP4.5, RCP8.5). For published literature indicated in Table A2. Globally, within segmented markets, the undernourishment rate is likely to increase by 45%, but with market integration it increases only by 27%, since a greater economic integration can work as a food security insurance against the most negative climate impact predictions.

2.4 Contribution for food governance and policy coherence

Food security governance is commonly stated at the transnational level, referring to the institutionalised process of bringing state and non-state actors for a cooperation action to solve problems that affect more than one state or region (De Haen and MacMillan, 2010). However, food security governance has expanded to other scales, national, community and household level, to overcome the barriers and problems that put at risk any dimension of food security (Candel, 2014). Modern food policies face (1) new conceptions, as pointed by McKeon (2015), opposing pathways between those upholding the dominant status quo model of industrial agriculture and those struggling for alternative models emphasising local diversified and resilient food systems, and (2) new drivers like regionalization, consumerism and the culture of choice, climate mitigation and adaptation, sustainability, and the spread and flow of information and technologies. However, trade rules have been defined primarily towards the maximization of commerce rather than to living within planetary boundaries (Lang and Ingram, 2013; Yearley, 2013), meaning the quest for sustainability has not been taken in the food domain.

A fragmentation of governance and leadership and apparent redundancy among big organizations is criticized by Lang and Ingram (2013). The inefficiency of the system makes it unable to define coherent policies across multiple scales to feed 9 billion people in 2050 (Candel, 2014).

Governance schemes must assure the interface with the multitude of issues that influence the short and long-term food security. That interface will be facilitated if, at different scales and across them, policies governing food security include the aspects of sectoral policies affecting any of its dimensions – here we focus on the availability and access dimensions.

Accomplishing this multidimensional policy and governance is a challenge of 21 century towards food security. The reviewed material on the impacts of climate change on food security dimensions, as well as, the selected strategies strengthening each dimension (Table 2.1), may contribute to respond to this challenge, by taking into consideration the high variability of impacts and, consequently, the potential for differentiated adaptation pathways.

Figure 2.5 Future climate change impacts on food access. Percentage of change of crop prices, projected for the mediumterm period (2040s-2060s), due to future climate trends in world regions. The reference period varies according to the published study but is within the interval 1961-2010. Crops with negative price change (i.e. price decrease) are indicated inside the red boxes, while the positive ones	(i.e. price increase) are inside green boxes. The lowest and highest price change (%) per region is indicated in square brackets. Climate change scenarios and adaptation actions are indicated in the legend box (see Fig.2.4 for details on adaptation). The distribution of the impacts is based on published literature indicated in Table A3.
	(Cereals, oil crops, sugar crops, tubers, cotton)c Wheat2 Rice2 I.1.5; 38] I.1.5; 38] I.1.5; 38] Climate scenarios a – A1B b – B1 c – A2 d – B2 d – B2 c – A1F1 1. RCP4.5 2. RCP8.5 2. RCP8.5 (No adaptation • Adaptation effect
Wheat2 Rice2 Soybeana,b [15;60] [15;60] [15] [15] [15]	<pre>(Cereals, oil crops, sugar crops, tubers, cotton)c [~-4] [~-4] [Wheat2 Rice2, Maizeb [14; 26] [14</pre>
SUB-SAHARAN AFRICA EUROPE	CENTRAL & EASTERN ASIA WESTERN & SOUTHERI & SOUTHEAST ASIA NORTHERN & CENTRAL AMERIC/ CENTRAL AMERIC/ CARIBBEAN & SOUTH AMERIC/



Results show that the type and magnitude of climate impacts on food availability is highly variable from place to place and for different food items: observed impacts on the availability dimension are mostly, but not only, negative (section 2.1.1) and projections highlight a majority of negative changes on crop responses, while few positive projections already include adaptation measures, as well as the effect of CO₂ fertilisation (section 2.1.2); it is also foreseen an increase in crop prices among projections but with a high variability (section 2.2.2), as it is the example of on the projected crop price increase in China (in medium future from 6% to 15%), depending on the emission scenario used. The high variability among projections call for mechanisms of governance and policy coherence that consider an adapted monitoring and foresight arrangements, moving away from pure projection and one size fits all solutions, which are not adapted to build intelligence for such complex issues.

Despite the richness and usefulness of the analysed material, we consider that there is the need to assess more studies on the climate change impacts on food access. Furthermore, we considered studies that look at regions from the point of view of production and not consumption, expressing thus a supply point of view. Such approach could be undertaken in follow up studies oriented to the nutritional value of food consumption, including the impacts of climate change on malnourishment. In addition, even if the selected future climate change impacts on food commodities can vary according to the climate change scenarios used, we provide a picture on the present and future hotspots of the impact on different crops – particularly cereals which are relevant in terms of caloric food consumption but also for providing feed to maintain the livestock sector.

The impacts of observed and projected changes in climate raise new factors and possibilities potentially exposing societies to risk, such as changes in spatial production patterns and water scarcity. These factors affect per se food security dimensions, and need to be taken into the design of policies and governance schemes. The spatial patterns of crops yield across the Planet are governed by the current spatial distribution of climate drivers, and may change with future climate and with the adaptive capacity in each region. For example, >60% of the calories in the human diet is highly dependent on just four grains – wheat, maize, rice, and sovbean – and the population that relies most on these grains is rising (Yearley, 2013). These crops are grown in only a handful of countries (Global Food Security Programme UK, 2015), and where observed climate trends are associated to yield losses, and to a lower extend to yield gains, but where future climate (i.e. 2050s) is projected to cause mostly negative impacts: >55% of the global wheat production is grown in the EU, China, South Asia, and the USA, but prospects for changes of wheat response to climate change, points to a decrease up to, respectively, -12, -15, -35, and -30% (for the highest emission scenarios). While in South Asia, even by considering drought and heat tolerant wheat varieties, the projected impact is negative, wheat yields in the USA could increase up to 60% with technological advances. The USA, China, Europe, and CSA produce >65% of the worldwide maize, where future climate change is expected to lead to yield declines up to, respectively, -50, -5, -40, and -25%. On the other hand, in the SSA region even though the observed climate impacts are associated to overall losses of -6%, the region may see its maize production increase by 24% if improved drought varieties are considered. More than 50% of global rice production is grown in Asian countries and (depending on the region and under higher emission scenarios), yield losses may reach -55% (-12% only in China). However, the uncertain CO₂ fertilisation effect may reverse such trend and lead to yield gains up to 26% across Asian countries. Finally, >88% of the worldwide soybean is grown in USA, CSA, and China, where future climate projections point to negative crop responses up to, respectively, -10, -25 and -5% under RCP8.5. On the other hand, for example South Africa may see its soybean yield to increase by 20% even without adaptation.

The potential climate change impacts on the spatial production patterns would potentially have direct consequences on the interconnections of the global food market. For example, the EU imports food and fisheries much more from Latin America (37,4 M€, representing 40% of total imports from the region) than from China (6,7 M€ representing 2% of total imports from the region) (EC, 2015a, 2015b). Prospects for changes of soybean yield for CSA, due to climate change, point to a decrease of 25% in the medium future. This means that Europe may be forced to change the countries food sources, which will be a challenge if a complete cross-regional analysis of the world regions will not be carried. Even though if in high income regions food security might not be an issue, the increase in food demand along with the cascading climate change impacts on trade patterns, could exert pressure on food prices and affect agricultural income in those regions, as it might be the case of the EU (European Environment Agency, 2019). With the negative productivity effect from climate change, the price of oil seeds, wheat, and rice are projected to increase by 10% in European countries and Brazil, 15% in China, USA and SSA countries, and 25% in India. The lack of food access may lead to a projected increase of the undernourishment rate by ~50%, in particular in South America and Africa, and Central Asia, and with some European areas, South-East Asia, USA and Russia also seeing an increase in population at risk.

The other critical factor is water scarcity, since agricultural activities are responsible for consuming, on average, 70% of the fresh water available on the planet, (FAO, 2012) and climate change adds significant uncertainty to the availability of water in many regions. It will impact both rainfed systems through precipitation patterns, and irrigated systems through availability of water at basin level. According to FAO, in 2009, 311 million hectares were equipped with irrigation, 84% of those actually being irrigated, corresponding to 16% of all cultivated land and contributing to 44% of total crop production (HLPE, 2015). Irrigation, as nitrogen fertilisation, is key to climate change adaptation, which requires financial resources that, if not available, will likely imply significant changes in the food trade movements. Moreover, competing uses of water are expected to exacerbate with climate change, as under a business as usual scenario, global water demand is projected to increase by 55% by 2050, with over 40% of the global population living in river basins experiencing severe water stress, especially in North and Southern Africa, and South and Central Asia (HLPE, 2015).

Climate change decision framework and governance bodies, especially regarding adaptation, must contribute to food policies at different scales, as already happens for the global scale regarding recommendations from the Committee for World Food Security (Committee on World Food Security, 2016). On other scales, crossing knowledge of climate change impacts and vulnerability into food policies eventually occurs, although no specific policies or governance schemes make it explicit. For example, at EU countries, as Portugal, strategies on food security and nutrition mostly focus on increasing knowledge on food habits and literacy of citizens to prevent health regimes, and on mechanisms to assure food access for social-economic vulnerable groups (DGS, 2015), taking for sure sufficient level of food availability. Issues relating climate change with food sources, water scarcity at importer countries or potential vulnerability of own food production systems, are not considered at all in food security strategies, representing a serious gap for a country food security.

Finally, aspects of low carbon and efficient use of resources along the food production, distribution and consumption must also be included in modern food security policies, primarily due to the need for a transition from current fossil fuels use to renewable and efficient energy sources, implying technological and cultural changes. Otherwise, adaptation options and practices counteract the aim of climate stabilization, as stated in the Paris Agreement, and sustainable consumption and production, as stated in SDGs (UNEP, 2015).

As pointed by Candel (2014) a sustainable food security asks for the (re)organization of the fragmented governance system by establishing connectivity between policy domains, scale levels, leadership and to clearly allocate responsibilities, costs and benefits. Climate change issues, as shown in this paper, provide the scope and key reasons to accelerate such renovation.

Scope	Description	Reference
Consumer- oriented food policy	To strengthen and train the capacity of communities to cope with adverse climate and to implement a consumer strategy for improving reliance on trade. Strengthen FOOD ACCESS.	(Avelino <i>et al.</i> , 2015; Mosnier <i>et al.</i> , 2014)
Policy definition and coordination	To implement policies to: (a) ensure availability of imports of vital commodities and an effective distribution of food resources from nations and regions and (b) define coordinated goals concerning public health and food security by strengthening collaboration between stakeholders (farmers, agricultural policymakers and air quality managers). Strengthen FOOD AVAILABILITY	(Dube et al., 2013; Tai et al., 2014)
Market	Institutional and infrastructural support in the form of access to governmental funds. Strengthen FOOD ACCESS	(Mosnier <i>et al</i> ., 2014)
access, credit,	Adaptation of the international trade in agricultural	(Dawson et al., 2016; Dube et al.,
and social	production, diversification of trading partners and access to	2013; Hertel and Baldos, 2016;
insurance	international and local markets. strengthen FOOD	Mosnier et al., 2014; Nelson et al.,
	AVAILABILITY and ACCESS	2014; Stevens and Madani, 2016)

Table 2.1 Strategies selected from the literature that strength food availability and access.

Scope	Description	Reference
Agriculture planning and strategy	Definition of strategies to: (a) stabilize crop production and shifts in crop varieties, (b) improve soil quality and fertility and (c) combine investment in agricultural research and increased water-use efficiency in agriculture. Strengthen FOOD AVAILABILITY and ACCESS	(Bu <i>et al.</i> , 2015; Emam <i>et al.</i> , 2015; Lobell and Tebaldi, 2014; Ray <i>et al.</i> , 2015; Stevens and Madani, 2016; Zhu <i>et al.</i> , 2013)

2.5 Conclusion

Climate change poses significant risk of disruption on food security, meaning a risk for the resilience of global to local food systems, and hence a challenge to food policies and, ultimately, to food security governance. The review of current evidence and projections of the impacts of climate change on food availability and access provides inputs for food policy design by gathering the regional differences.

Evidence shows that crop responses to climate change are highly variable, and not only negative, from place to place and for different food items. Future climate trends, on the other hand, are projected to lead mostly to negative changes, while few positive changes on crop responses may occur if adaptation measures and the effect of CO₂ fertilisation are considered. However, even if adaptation measures may decrease or even offset the adverse climate change for certain crops and regions, our review shows that, overall, the current global breadbaskets are projected to be negatively affected, with harmful consequences on the food supply and on its prices through the interconnections of the global food market. In a mediumterm future, under the highest emission scenarios and if no adaptation takes place, wheat yields or production are projected to decline in the EU (up to -12%), China (-15%), South Asia (-35%), and the USA (-30%) where here yields could still increase (+60%) with technological advances. Maize production is projected to decline in the USA (up to -50%), China (-5%), and Europe (-40%). In Asian countries a decline in rice yields (up to -55%) may be reversed (with gains up to 26%) when the still uncertain CO₂ fertilisation effect is considered. Under RCP8.5, the change in soybean responses may be negative in the USA (up to -10%), Caribbean and South America (-25%), and China (-5%). On the other hand, for example South Africa may see its soybean yield to increase by 20% even without adaptation, and Sub-Saharan countries have potential to increase maize production (up to 24%) if improvements on crop adaptation are implemented. Without further adaptation, by 2050s, crop prices are projected to increase in different regions and among different scenarios, for example for aggregated oil seeds, wheat and rice by 15% across Sub-Saharan countries and China, 25% in India, and 10% in Brazil, and between 14 to 32% for maize, wheat and rice in Pakistan which is expected to become a net food importer.

The knowledge acquired with this review is useful for food governance schemes and policy coherence as it draws the attention to the high variability of impacts among regions and for different food items and, consequently, for differentiated adaptation pathways. Studies addressing a cross-regional analysis of the

world regions in terms of water scarcity and spatial patterns change of production and consumption need to be considered for the design of food policies and governance schemes. These factors may imply significant changes in the food trade movements, therefore potentially compromising food security. Finally, climatefood-policies combining adaptation and mitigation is key to answer to regional specific adaptation measures, while adjusting the transition from current fossil fuels use to renewable and efficient energy use.

2.6 References

- 1. Ahmed, K.F., Wang, G., Yu, M., Koo, J., You, L., 2015. Potential impact of climate change on cereal crop yield in West Africa. Climatic Change 133, 321–334. https://doi.org/10.1007/s10584-015-1462-7
- Asseng, S., Ewert, F., Martre, P., Rötter, R.P., Lobell, D.B., Cammarano, D., Kimball, B.A., Ottman, M.J., Wall, G.W., White, J.W., Reynolds, M.P., Alderman, P.D., Prasad, P.V. V., Aggarwal, P.K., Anothai, J., Basso, B., Biernath, C., Challinor, A.J., De Sanctis, G., Doltra, J., Fereres, E., Garcia-Vila, M., Gayler, S., Hoogenboom, G., Hunt, L.A., Izaurralde, R.C., Jabloun, M., Jones, C.D., Kersebaum, K.C., Koehler, A.-K., Müller, C., Naresh Kumar, S., Nendel, C., O'Leary, G., Olesen, J.E., Palosuo, T., Priesack, E., Eyshi Rezaei, E., Ruane, A.C., Semenov, M.A., Shcherbak, I., Stöckle, C., Stratonovitch, P., Streck, T., Supit, I., Tao, F., Thorburn, P.J., Waha, K., Wang, E., Wallach, D., Wolf, J., Zhao, Z., Zhu, Y., 2015. Rising temperatures reduce global wheat production. Nature Climate Change 5, 143–147. https://doi.org/10.1038/nclimate2470
- Avelino, J., Cristancho, M., Georgiou, S., Imbach, P., Aguilar, L., Bornemann, G., Läderach, P., Anzueto, F., Hruska, A.J., Morales, C., 2015. The coffee rust crises in Colombia and Central America (2008–2013): impacts, plausible causes and proposed solutions. Food Security 7, 303–321. https://doi.org/10.1007/s12571-015-0446-9
- Banerjee, S., Das, S., Mukherjee, A., 2014. Adaptation strategies to combat climate change effect on rice and mustard in Eastern India. Mitig Adapt Strateg Glob Change. https://doi.org/10.1007/s11027-014-9595-y
- 5. Brown, M.E., Kshirsagar, V., 2015. Weather and international price shocks on food prices in the developing world. Global Environmental Change 35, 31–40. https://doi.org/10.1016/j.gloenvcha.2015.08.003
- Bu, L., Chen, X., Li, S., Liu, J., Zhu, L., Luo, S., Lee, R., Zhao, Y., 2015. The effect of adapting cultivars on the water use efficiency of dryland maize (Zea mays L.) in northwestern China. Agricultural Water Management 148, 1–9. https://doi.org/10.1016/j.agwat.2014.09.010
- Candel, J., 2014. Food security governance: a systematic literature review. Food Security 6, 585–601. https://doi.org/10.1007/s12571-014-0364-2
- Challinor, A.J., Smith, M.S., Thornton, P., 2013. Use of agro-climate ensembles for quantifying uncertainty and informing adaptation. Agricultural and Forest Meteorology 170, 2–7. https://doi.org/10.1016/j.agrformet.2012.09.007
- Challinor, A.J., Watson, J., Lobell, D.B., Howden, S.M., Smith, D.R., Chhetri, N., 2014. A meta-analysis of crop yield under climate change and adaptation. Nature climate change 4, 287–291. https://doi.org/10.1038/NCLIMATE2153
- Chen, C., Zhou, G.S., Zhou, L., 2014. Impacts of climate change on maize and winter wheat yields in China from 1961 to 2010 based on provincial data. Journal of Integrative Agriculture 13, 1555–1564. https://doi.org/10.1017/S0021859614001154
- Committee on World Food Security, 2016. Making a Difference in Food Security and Nutrition Forty-third Session - CFS 2016/43. Rome, Italy.
- 12. Dawson, T.P., Perryman, A.H., Osborne, T.M., 2016. Modelling impacts of climate change on global food security. Climatic Change 134, 429–440. https://doi.org/10.1007/s10584-014-1277-y
- 13. De Haen, H., MacMillan, A., 2010. Towards global governance of food security. Focus, Rural 21.
- DGS, 2015. Estratégia para a promoção da alimentação saudável em Portugal. Revista da Direção Geral da Saúde - Portugal Saúde em Números 4, 37–41.

- Dube, S., Scholes, R.J., Nelson, G.C., Mason-D, D., Palazzo, A., Sikhalazo Dube, A., Sikhalazo Dube, C., 2013. South African food security and climate change: Agriculture futures. Economics: The Open-Access, Open-Assessment E-Journal 7, 2013–35. https://doi.org/10.5018/economics-ejournal.ja.2013-35
- 16. EC, 2015a. European Union, Trade in goods with China.
- 17. EC, 2015b. European Union, Trade in goods with Latin American Countries.
- Emam, A.R., Kappas, M., Hosseini, S.Z., 2015. Assessing the impact of climate change on water resources, crop production and land degradation in a semi-arid river basin Ammar Ra fi ei Emam, Martin Kappas and Seyed Zeynalabedin Hosseini. Hydrology Research 854–870. https://doi.org/10.2166/nh.2015.143
- 19. European Environment Agency, 2019. Climate change adaptation in the agriculture sector in Europe. Luxembourg. https://doi.org/10.2800/537176
- Fader, M., Von Bloh, W., Shi, S., Bondeau, A., Cramer, W., 2015. Modelling Mediterranean agro-ecosystems by including agricultural trees in the LPJmL model. Geoscientific Model Development 8, 3545–3561. https://doi.org/10.5194/gmd-8-3545-2015
- 21. FAO, 2013. Measuring different dimensions of food security. Rome, Italy.
- 22. FAO, 2012. Greening the Economy with Agriculture. Rome, Italy.
- 23. FAO, 2008. The State of Food Insecurity in the World: High Food Prices and Food Security—Threats and Opportunities. Rome, Italy.
- 24. FAO, IFAD, UNICEF, WFP, WHO, 2018. The State of Food Security and Nutrition in the World 2018. Building climate resilience for food security and nutrition. Licence: CC BY-NC-SA 3.0 IGO. Rome, Italy.
- FAO, IFAD, WFP, 2014. The State of Food Insecurity in the World 2014. Strengthening the enabling environment for food security and nutrition. Rome, Italy.
- Fukuda-Parr, S., Orr, A., 2013. The MDG Hunger Target and the Contested Visions of Food Security, Working Paper Series - The Power of Numbers: A Critical Review of MDG Targets for Human Development and Human Rights. Cambridge, MA.
- Global Food Security Programme UK, 2015. Climate and Global Crop Production Shocks (2015). Final Project Report from the UK–US Taskforce on Extreme Weather and Global Food System Resilience. UK.
- 28. Hertel, T.W., Baldos, U.L.C., 2016. Attaining food and environmental security in an era of globalization. Global Environmental Change 41, 195–205. https://doi.org/10.1016/j.gloenvcha.2016.10.006
- 29. HLPE, 2015. Water for food security and nutrition High Level Panel of Experts, Comiittee on World Food Security. Rome, Italy.
- IPCC, 2019. Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems [P.R. Shukla, J. Skea, E. Calvo Buendia, V. Masson-Delmot.
- Islam, S., Cenacchi, N., Sulser, T.B., Gbegbelegbe, S., Hareau, G., Kleinwechter, U., Mason-D'Croz, D., Nedumaran, S., Robertson, R., Robinson, S., Wiebe, K., 2016. Structural approaches to modeling the impact of climate change and adaptation technologies on crop yields and food security. Global Food Security 10, 63–70. https://doi.org/10.1016/j.gfs.2016.08.003
- Jiang, Y., Koo, W.W., 2014. Estimating the local effect of weather on field crop production with unobserved producer behavior: A bioeconomic modeling framework. Environmental Economics and Policy Studies 16, 279– 302. https://doi.org/10.1007/s10018-014-0079-9
- Ju, H., van der Velde, M., Lin, E., Xiong, W., Li, Y., 2013. The impacts of climate change on agricultural production systems in China. Climatic Change 120, 313–324. https://doi.org/10.1007/s10584-013-0803-7
- Kassie, B.T., Asseng, S., Rotter, R.P., Hengsdijk, H., Ruane, A.C., Van Ittersum, M.K., 2015. Exploring climate change impacts and adaptation options for maize production in the Central Rift Valley of Ethiopia using different climate change scenarios and crop models. Climatic Change 129, 145–158. https://doi.org/10.1007/s10584-014-1322-x

- 35. Krishnamurthy, P.K., Lewis, K., Kent, C., Aggarwal, P., 2015. Climate Impacts on Food Security and Livelihoods in Asia A review of existing knowledge. Sweden.
- 36. Lang, T., Ingram, J., 2013. Food security twists and turns, in: O'Riordan, T., Lenton, T. (Eds.), Addressing Tipping Points for a Precarious Future. Oxford University Press, Oxford.
- Li, T., Hasegawa, T., Yin, X., Zhu, Y., Boote, K., Adam, M., Bregaglio, S., Buis, S., Confalonieri, R., Fumoto, T., Gaydon, D., Marcaida, M., Nakagawa, H., Oriol, P., Ruane, A.C., Ruget, F., Singh, B., Singh, U., Tang, L., Tao, F., Wilkens, P., Yoshida, H., Zhang, Z., Bouman, B., 2015. Uncertainties in predicting rice yield by current crop models under a wide range of climatic conditions. Global Change Biology 21, 1328–1341. https://doi.org/10.1111/gcb.12758
- Lobell, D.B., Tebaldi, C., 2014. Getting caught with our plants down: the risks of a global crop yield slowdown from climate trends in the next two decades. Environmental Research Letters 9, 074003. https://doi.org/10.1088/1748-9326/9/7/074003
- McKeon, N., 2015. Food Security Governance Empowering Communities, Regulating Corporations. Routledge, London and New York.
- Mereu, V., Carboni, G., Gallo, A., Cervigni, R., Spano, D., 2015. Impact of climate change on staple food crop production in Nigeria. Climatic Change 132, 321–336. https://doi.org/10.1007/s10584-015-1428-9
- Moore, F.C., Lobell, D.B., 2015. The fingerprint of climate trends on European crop yields. PNAS 112, 2670– 2675. https://doi.org/10.1073/pnas.1409606112
- Mosnier, A., Obersteiner, M., Havlík, P., Schmid, E., Khabarov, N., Westphal, M., Valin, H., Frank, S., Albrecht, F., 2014. Global food markets, trade and the cost of climate change adaptation. Food Security 1–16. https://doi.org/10.1007/s12571-013-0319-z
- 43. Mouillé, B., Burlingame, U.R., Barbara, C., 2008. The Contribution of Plant Genetic Resources to Health and Dietary Diversity. Rome, Italy.
- Nelson, G.C., Valin, H., Sands, R.D., Havlík, P., Ahammad, H., Deryng, D., Elliott, J., 2014. Climate change effects on agriculture : Economic responses to biophysical shocks. PNAS 111. https://doi.org/10.1073/pnas.1222465110
- Newbery, F., Qi, A., Fitt, B.D.L., 2016. Modelling impacts of climate change on arable crop diseases: progress, challenges and applications. Current Opinion in Plant Biology 32, 101–109. https://doi.org/10.1016/j.pbi.2016.07.002
- Pickering, C., Byrne, J., 2014. The benefits of publishing systematic quantitative literature reviews for PhD candidates and other early-career researchers. Higher Education Research & Development 33, 534–548. https://doi.org/10.1080/07294360.2013.841651
- Ray, D.K., Gerber, J.S., MacDonald, G.K., West, P.C., 2015. Climate variation explains a third of global crop yield variability. Nature Communications 6, 5989. https://doi.org/10.1038/ncomms6989
- Ray, D.K., West, P.C., Clark, M., Gerber, J.S., Prishchepov, A. V, Chatterjee, S., 2019. Climate change has likely already affected global food production. PLOS ONE 14, e0217148.
- Rosenzweig, C., Elliott, J., Deryng, D., Ruane, A.C., Müller, C., Arneth, A., 2014. Assessing agricultural risks of climate change in the 21st century in a global gridded crop model intercomparison. PNAS 111, 4–9. https://doi.org/10.1073/pnas.1222463110
- Rötter, R.P., Höhn, J.G., Fronzek, S., 2014. Projections of climate change impacts on crop production: A global and a Nordic perspective. Acta Agriculturae Scandinavica, Section A — Animal Science 62:4, 166–180. https://doi.org/10.1080/09064702.2013.793735
- 51. Sanchez, B., Rasmusses, A., Porter, J., 2014. Temperatures and the growth and development of maize and rice: a review. Global Change Biology 20, 408–417. https://doi.org/10.1111/gcb.12389
- 52. Shi, P., Tang, L., Lin, C., Liu, L., Wang, H., Cao, W., Zhu, Y., 2015. Modeling the effects of post-anthesis heat stress on rice phenology. Field Crops Research 177, 26–36. https://doi.org/10.1016/j.fcr.2015.02.023
- Srivastava, A.K., Gaiser, T., Ewert, F., 2015. Climate change impact and potential adaptation strategies under alternate climate scenarios for yam production in the sub-humid savannah zone of West Africa. Mitig Adapt Strateg Glob Change. https://doi.org/10.1007/s11027-015-9639-y

- Stevens, T., Madani, K., 2016. Future climate impacts on maize farming and food security in Malawi. Scientific Reports 6, 36241. https://doi.org/10.1038/srep36241
- Stringer, R., 2016. Food Security Global Overview, in: M. Caraher and J. Coveney (Ed.), Food Poverty and Insecurity: International Food Inequalities. Springer International Publishing Switzerland 2016, Nexus Tower, Adelaide, SA 5005, Australia, pp. 11–18. https://doi.org/10.1007/978-3-319-23859-3_2
- 56. Tai, A.P.K., Martin, M.V., Heald, C.L., 2014. Threat to future global food security from climate change and ozone air pollution. Nature Climate Change 4, 817–821. https://doi.org/10.1038/nclimate2317
- Tao, F., Zhang, Z., 2013. Climate change, wheat productivity and water use in the North China Plain: A new superensemble-based probabilistic projection. Agricultural and Forest Meteorology 170, 146–165. https://doi.org/10.1016/j.agrformet.2011.10.003
- Tripathi, A., Kumar, D., Chauhan, D.K., Kumar, N., Singh, G.S., 2016. Agriculture, Ecosystems and Environment Paradigms of climate change impacts on some major food sources of the world : A review on current knowledge and future prospects. Agriculture, Ecosystems and Environment 216, 356–373. https://doi.org/10.1016/j.agee.2015.09.034
- 59. UNEP, 2015. Policy Coherence of the Sustainable Development Goals A Natural Resource Perspective. France.
- Wang, J., Liu, X., Zhang, X., Smith, P., Li, L., Filley, T.R., Cheng, K., Shen, M., He, Y., Pan, G., 2016. Size and variability of crop productivity both impacted by CO2 enrichment and warming-A case study of 4 year field experiment in a Chinese paddy. Agriculture, Ecosystems and Environment 221, 40–49. https://doi.org/10.1016/j.agee.2016.01.028
- Wang, W., Yu, Z., Zhang, W., Shao, Q., Zhang, Y., Luo, Y., Jiao, X., Xu, J., 2014. Responses of rice yield, irrigation water requirement and water use efficiency to climate change in China: Historical simulation and future projections. Agricultural Water Management 146, 249–261. https://doi.org/10.1016/j.agwat.2014.08.019
- Waongo, M., Laux, P., Kunstmann, H., 2015. Adaptation to climate change: The impacts of optimized planting dates on attainable maize yields under rainfed conditions in Burkina Faso. Agricultural and Forest Meteorology 205, 23–39. https://doi.org/10.1016/j.agrformet.2015.02.006
- 63. Wei, T., Cherry, T.L., Glomrød, S., Zhang, T., 2014. Climate change impacts on crop yield: Evidence from China. Science of the Total Environment 499, 133–140.
- 64. Yearley, S., 2013. Full Planet, Empty Plates: The New Geopolitics of Food Scarcity, Lester R Brown, W W. Food Security 5, 285–286. https://doi.org/10.1007/s12571-013-0246-z
- 65. Zhao, C., Liu, B., Piao, S., Wang, X., Lobell, D.B., Huang, Y., Huang, M., Yao, Y., Bassu, S., Ciais, P., Durand, J.-L., Elliott, J., Ewert, F., Janssens, I.A., Li, T., Lin, E., Liu, Q., Martre, P., Müller, C., Peng, S., Peñuelas, J., Ruane, A.C., Wallach, D., Wang, T., Wu, D., Liu, Z., Zhu, Y., Zhu, Z., Asseng, S., 2017. Temperature increase reduces global yields of major crops in four independent estimates. Proceedings of the National Academy of Sciences 114, 9326 LP 9331. https://doi.org/10.1073/pnas.1701762114
- Zhao, J., Guo, J., Mu, J., Xu, Y., 2015. Exploring the dynamics of agricultural climatic resource utilization of spring maize over the past 50 years in Northeast China. Physics and Chemistry of the Earth 87–88, 19–27. https://doi.org/10.1016/j.pce.2015.07.008
- Zhou, Y., Li, N., Dong, G., Wu, W., 2013. Impact assessment of recent climate change on rice yields in the heilongjiang reclamation area of north-east china. Journal of the Science of Food and Agriculture 93, 2698–2706. https://doi.org/10.1002/jsfa.6087
- Zhu, T., Ringler, C., Mohsin Iqbal, M., Sulser, T.B., Arif Goheer, M., 2013. Climate change impacts and adaptation options for water and food in Pakistan: scenario analysis using an integrated global water and food projections model. Water International 38, 651–669. https://doi.org/10.1080/02508060.2013.830682
- 69. Zinyengere, N., Crespo, O., Hachigonta, S., 2013. Crop response to climate change in southern Africa: A comprehensive review. Global and Planetary Change 111, 118–126. https://doi.org/10.1016/j.gloplacha.2013.08.010

3 Exposure of the EU food imports to extreme weather disasters in exporting countries

This study is published in the journal Food Security:

Brás, T.A., Jägermeyr, J., Seixas, J., 2019. Exposure of the EU-28 food imports to extreme weather disasters in exporting countries. Food Security. https://doi.org/10.1007/s12571-019-00975-2

Author contributions:

T.B., J.J., and J.S. designed the research. T.B. compiled and analysed the data and wrote the paper. The paper was discussed and revised by all authors.

Abstract

EU relies on a diversified foreign market, even for crops for which it has a high self-sufficiency. This study contributes to the discussion on the vulnerability of agri-food supply to the impacts of extreme weather disasters (EWD). We focus on the largest import commodities of the EU and we aim to (1) map external dependencies of EU agri-food sector, (2) estimate the impact of EWD on crop production in countries from where the EU receives their imports, and (3) assess the exposure of EU agri-food imports to such impacts. Crop and trade data are acquired through EUROSTAT and FAOSTAT, EWD records from EM-DAT, all between 1961 and 2016. A superposed epoch analysis is used to estimate the impact of EWD on the average national production, yield, and harvested area of selected crops in exporting countries.

The EU imports between 35-100% of its consumption of soybeans, banana, tropical fruits, coffee, and cocoa. Our study reveals a substantial impact of EWD, especially due to droughts and heatwaves, on the production of soybeans, tropical fruits, and cocoa, with import weighted impacts of 3, 8, and 7%, respectively. Floods cause weighted impacts of 7% (soybeans) and 8% (tropical fruits). Coffee production shows gains during cold waves, but the inter-annual variability offsets these effects.

This study provides conclusions that may support EU on the development of adaptation schemes in external supplier countries to secure EU food supply. Such schemes may prioritize provisions contributing for the stability of crop production and incomes in those countries, while dealing with future adverse EWD impacts.

Keywords

Extreme Weather Disasters, Crop Production, Yield, Harvested Area, EU External Suppliers, EU Import Share-Weighted Impacts

3.1 Introduction

Extreme weather events can cause damage to crops and food production systems, and associated price spikes have the potential to destabilize food systems and threaten local to global food security (Lesk *et al.*, 2016; Nelson *et al.*, 2014; Rosenzweig *et al.*, 2014). The severity of an extreme weather event and the vulnerability and exposure of the human and natural systems to it will determine whether it results in a disaster (IPCC, 2012).

In the last four decades, droughts and heatwaves have caused between 1200 and 1800 million tonnes of losses in national maize, rice, and wheat production, respectively (Lesk *et al.*, 2016). Jägermeyr & Frieler (2018) confirm these findings with global crop modeling and show that heatwaves and droughts predominantly affect rainfed rather than irrigated yields.

This first line of evidence suggests that damages are about 10% stronger in developed countries (Europe, North America and Australasia) compared to the developing world (Asia and Africa), where the crop and management diversification across many small fields allows for drought resistance (Lesk *et al.*, 2016). In addition, it is shown that smallholders tend to minimize the risk of crop loss, whereas in higher-income countries the priority is to maximize yield, which can compromise the resistance to droughts. The EWD impacts on specific crops in tropical export-oriented countries and associated implications through trade dependencies have, however, not been explored in that study.

Our study is focused on the exposure of 28-Member States of the European Union (EU) agri-food supply to extreme weather disasters (EWD). The EU is one of the world's largest suppliers and producers of food (EU, 2018). Previously published impacts of EWD on agricultural production within the EU are summarized in Table B1 (in the Appendix B). As a central example, during the 2003 heat wave >10% declines in crop yields were reported in Italy, Germany, Austria, Spain, France, and Portugal (Jägermeyr *et al.*, 2018). Wheat and maize were the most damaged crops, with reductions of 11% (10 Mt) and 21% (9 Mt), respectively (COPA-COGECA, 2003). Impacts were amplified regionally, across the Iberian Peninsula, cereals production fell on average by 40% during the 2004-2005 drought (EEA, 2010).

Extreme weather implications for the European food production system causes higher food import demands, but exporting countries can be affected as well (IPCC, 2014). Consequently, in view of potential future aggravations in global extreme weather event frequency and intensity due to climate change (Hanks &

Craeynest, 2014; IPCC, 2012, 2014), there are growing concerns about Europe's food availability and access not just in terms of its own production (since the EWD can affect crop availability and its prices in the EU), but especially through cascading effects due to trade dependencies. In fact, Europe is the world's biggest importer of food, with about 70% of food EU-external imports from the developing world, regions considered highly vulnerable to climate change (EU, 2018; Hanks *et al.*, 2014; IPCC, 2014). Trade dependencies propagate weather-related food production shocks throughout the global food system (Puma *et al.*, 2015; Rosenzweig *et al.*, 2001) and the reliance of the global food system on trade is expected to become even more substantial (Brooks & Matthews, 2015).

This study sets out to (1) map the external dependency of the EU agri-food sector, (2) estimate the impact of EWD on crop production, yield and harvested area in countries from which the EU receives their imports (also referred as exporting countries or external supplier countries throughout the text), and (3) assess the exposure of the EU agri-food imports to such weather-related shocks. This work does not consider any food price analysis.

	Goal	Data sets (country-based data)	Time series	Source
1.	Mapping dependencies of EU agri-food supply	Agri-food products at EU: production, imports and exports	2005-2014	(FAO, 2017), (EUROSTAT, 2016)
		Agri-food products at EU supplier countries: production		
 Influence of EWD in the agri-food products among supplier countries 	Agri-food products at EU supplier countries: production, harvest area and yield	1961-2016	(FAO, 2017)	
	Influence of EWD in the agri-food products among supplier countries	EWD in EU supplier countries': floods, droughts, heatwaves and cold waves	1964-2013	(EM-DAT, 2018)
		Percentage of irrigated area per agri-food product in each country	Latest available value	(FAO, 2016), (FAO, 2017)
		Koppen Geiger climate classification	2000	(Kottek, Grieser, Beck, Rudolf, & Rubel, 2006)
3.	Exposure of EU agri-food imports due to EWD in supplier countries	Agri-food products at EU supplier countries: production, harvest area and yield	1961-2016	(FAO, 2017)
		EWD in EU supplier countries': floods, droughts, heatwaves and cold waves	1964-2013	(EM-DAT, 2018)
		EU import share per supplier country	2005-2014	(FAO, 2017)

 Table 3.1 List of goals and respective data sources used in this study.

3.2 Methods

3.2.1 Mapping external dependency and sufficiency of EU agri-food supply

The EU imported crop categories, between 2005 and 2014, are selected trough EUROSTAT (EUROSTAT, 2016) and FAOSTAT (FAO, 2017). Datasets used in this study are listed in Table 3.1. Processed food products are not considered for the analysis, as it is difficult to identify the exporting countries providing production statistics of such commodities. From the 48 crop categories imported by EU, we selected the following 12, representing 86% (in quantity) of the total imported: (1) soybeans, (2) maize, (3) wheat and meslin, (4) bananas, (5) rice, (6) cane or beet sugar, (7) coffee, (8) rape or colza seeds, (9) citrus fruit, (10) coccoa, (11) tropical fruits (dates, figs, pineapples, avocados, guavas, mangoes) and (12) apples, pears and quinces. For these crops, the import dependency and self-sufficiency are calculated, according to equations (1) and (2) respectively, by using data on imports, exports and production reported for EU along ten years. The food import dependency means the reliance on imports for a country's food consumption needs, while food self-sufficiency refers to a country's ability to meet its own food requirements from domestic production without imports (Clapp, 2015). For simplification, and due to lack of data, crop reserves are not considered in the equations.

Eq. (1):

$$Id_{crop} = \left(\sum_{i=1}^{10} I_{crop}\right) / \left(\sum_{i=1}^{10} P_{crop} + I_{crop} - E_{crop}\right) * 100$$

.

Eq. (2):

$$Ss_{crop} = \left(\sum_{i=1}^{10} P_{crop}\right) / \left(\sum_{i=1}^{10} P_{crop} + I_{crop} - E_{crop}\right) * 100$$

Where,

Id_{crop} = Crop import dependency (%)

- SS_{crop} = Crop self-sufficiency (%)
- I_{crop} = Crop imports (tonnes)
- P_{crop} = Crop production (tonnes)
- E_{crop} = Crop exports (tonnes)

crop = each of the twelve crops

i = number of years, from 2005 to 2014

By selecting the world exporting countries supplying at least 95% of each crop (in quantity) to EU, we can map the main exporting countries per crop and the geographic distribution of EU import dependency (Fig. 3.1). Figure 3.2a shows that the EU exhibits a self-sufficiency above 70% for rice, citrus, maize, rape and colza seeds, apples, pears, quinces, wheat, and sugar beet, even though these crops are among the 12 most imported in quantity. In fact, wheat, apples, pears, and quinces, show an EU self-sufficiency above 100%, meaning that the region produces more than what it consumes, and the remainder is exported.

For soybeans, bananas, tropical fruits, coffee, and cocoa, the EU self-sufficiency is below 9%, and 35 to 100% is being imported (between 2005 and 2014). The EU import dependency of coffee is even higher than 100% as there are coffee exports, but no production. Soybeans shows a similar picture; demand exceeds by far the internal production mostly due to the livestock sector (Ercin *et al.*, 2016).

Figure 3.2b presents the 41 countries that collectively provide more than 35% of the EU imports for soybeans, banana, tropical fruits, coffee and cocoa. Soybeans is mostly provided by North American and South American countries, banana from Central and South American countries, tropical fruits mostly from Central America, coffee from South America and Asia, and cocoa from the African countries. Those are the five crops and the exporting countries that are considered for further assessment of the impact of EWD on crop production, yield, and harvested (section 3.2.2).

3.2.2 Impact of EWD on crop production in exporting countries relevant for the EU

We use a Superposed Epoch Analysis (SEA), a time series statistical method used in data analysis, to isolate the average response signal of EWD on national crop production, while reducing noise due to extraneous variables, such as human decision making and agronomic management. This methodology is based in Lesk *et al.*, 2016 who estimated national cereal production losses across the globe resulting from reported EWD, and in Jägermeyr *et al.*, 2018 who represented spatially explicit information of growing seasons and surface water constraints in global gridded crop model simulations to quantify, through a SEA, the associated gains in model performance regarding annual fluctuations in national maize and wheat yields. The SEA analysis, also known as compositing, was mainly introduced by Mass *et al.*, 1989.

The SEA is applied to national production, yield, and harvested area from each of the five crops supplied by each exporting country. Crop data are obtained from FAOSTAT, between 1961 and 2016. The cases of banana from Suriname, tropical fruits from Panama and Ghana, coffee from Ethiopia, and cocoa from Togo and Guinea were excluded from the analysis since there is missing data on production, yield and/or harvested area. Therefore, this analysis consideres 37 out of the 41 external supplier countries. Due to an increasing trend in crop production, yield and harvested area, observational data are detrended. The trend is removed by subtracting the linear best-fit function from each time series. The result is a time series with normalised fluctuations from year to year.

Data on EWD is gathered for the same period through The International Disaster Database (EM-DAT, 2018). According to EM-DAT, for a disaster to be entered into the database at least one of the following criteria must be fulfilled: ten or more people reported killed, one hundred or more people reported affected, declaration of a state of emergency or call for international assistance. For this study we consider floods, droughts, heatwaves, and cold waves.



Figure 3.1 Geographic distribution of EU food import dependency.

Data is presented per crop, per world region and per exporting country (ISO3 codes), between 2005 and 2014. For simplicity "tropical fruits" aggregate dates, figs, pineapples, avocados, guavas, mangoes, and "sugar" aggregates sugar beet and sugar cane. Data acquired through (FAO, 2017) and (EUROSTAT, 2016).



Figure 3.2 EU self-sufficiencies per crop category and import shares per external supplier.

(a) EU import dependency and self-sufficiency ratios, and amount of imports and production per crop category (Mt., secondary y-axis). (b) EU import share per exporting country (ISO3 codes) for soybeans, banana, coffee, cocoa, and tropical fruits (i.e. crops for which EU import dependency is higher than 35%).

Tropical fruits include dates, figs, pineapples, avocados, guavas, mangoes. Colors: grey for European and North American countries, green for South American countries, pink and red for Central American countries, blue for Asian countries and yellow for African countries. All data are obtained from the FAO (FAO, 2017) and (EUROSTAT, 2016) and refer to the time period 2005 to 2014. Grey boxes with countries acronyms in red indicate the countries that (due to lack of data in the original data bases) are excluded from estimating the impact of EWD on crop production.

As in Lesk et al., 2016, from each time series of crop production (i.e. one time series per crop and per exporting country) we extract shorter time series using a 7-year window centered on the year of occurrence of an EWD type, with 3 years of data preceding and following the event. For example, if in the period of analysis, ten years of droughts are reported (in non-consecutive years), then we would have ten time series of a 7-years window centered in each drought (which we call "Drought TS"). For production time series, this procedure is implemented four times, one per EWD type. Each 7-year window time series is normalised (year-wise) to the average of the 3 years preceding and following the EWD. We stress that the average of those six adjacent years is calculated only for the years with no EWD of the same type (i.e. non-disaster years). Therefore, whenever there is an EWD in one of the 3 years preceding and following the event, that year is excluded from calculating the mean. Also, for the same reason, the EWD occurring between 1961 and 1963, and 2014 and 2016 are not considered. Whenever an EWD of the same type occurs in multiyears, we average crop production across all EWD years to produce a single disaster year datum, which is then centered in the 3 years preceding and following the event. This procedure results in a reduction in the total number of events since the average of sequential EWD years (of same type) is considered as one event. By centering the time series in EWD years we are strengthening the signal (positive or negative) at the year of the event while also cancelling the noise in the non-disaster years. After implementing those procedures, we obtain a composite which is the mean of all the time series for an EWD type (in the above given example the composite would be the column average of the "Drought TS"). A list of the EWD that took place in the exporting countries supplying the EU with each crop is provided on Tables B2-B4. These are the EWD considered in this study.

The composites are calculated by the following approaches: 1st) by aggregating all time series per EWD type, regardless the crop, and 2nd) by aggregating the time series of the exporting countries supplying the EU with each crop. This is done to enlarge our samples of EWD and to detect whether there is a signal in production data corresponding to when the disasters occurred.

We combine droughts and heatwaves in the same composite and then perform the analysis by aggregated and by individual crops. Since the effect of those events on crop production may be offset, or even enhanced, if the crop is irrigated and/or if grown in a tropical wet climate (characterized by high surface temperatures with plentiful precipitation), we also analyse the effect of droughts and heatwaves by considering only the exporting countries supplying the EU with crops grown in rainfed and non-tropical systems (Table B5).

For that case, only the countries with a percentage of irrigated harvested area higher than 40% are removed from the analysis. The percentage of irrigated area per crop, in each exporting country, is calculated through the ratio of the irrigated harvested area (provided by AQUASTAT (FAO, 2016)) with the total harvested area (provided by FAOSTAT (FAO, 2017)). This is calculated only for the most recent year with information available in AQUASTAT. According to the Koppen-Geiger classification (Kottek *et al.*, 2006), exporting
countries having 'Tropical rainforest climate' (Af) and 'Tropical monsoon climate' (Am) as a dominant climate classification are removed from the analysis.

For simplification an equal weight is attributed to all EWD regardless the EWD type, location, duration, and impact. The above-mentioned procedure is applied to production, yield and harvested area time series, in total 12 time series per crop (i.e. a time series for production, yield, and harvested area considering the impact of floods, the combined droughts and heatwaves, and cold waves).

With the SEA we estimate the associated loss or gain in production, yield, and harvested area of each crop. The assessment of the statistical significance of the averaged normalised mean at the EWD years is performed from bootstrap replicate data sets, which are obtained by resampling (with replacement) the time series of crop production, yield, and harvested area. Bootstrapping resamples a dataset with replacement thousands of times to create simulated datasets. Specifically, per each crop and EWD type, each one of the 7-years' time series is resampled (column-base), while applying the SEA, to create 1000 different composites. The normality of the normalised 1000 means at the EWD years is assessed with the histogram. For all crops we observe a normal distribution, therefore, for simplification, and as an example, histograms showing a normal distribution of the data are presented only for the resampled normalised means of aggregated crop production (Fig. B.1). The normalised mean at the EWD year of the 1000 resamples is considered to be statistically significant for the confidence intervals (CI) of 95%, 90%, 85%, 80%, and not significant for CI below 80%. This technique is well adopted in statistical models linking climate and crop yields (Leng & Huang, 2017). The MATLAB code to create a bootstrap to replicate a data set can be found at: https://www.mathworks.com/help/stats/datasample.html.

3.2.3 Exposure of EU agri-food imports

The averaged impact estimated for each crop and EWD type (section 3.2.2), results from the arithmetic average of the impacts estimated from all the EWD that occurred in external supplier countries. This means that, among all the exporting countries relevant for the EU, only the ones with reported EWD are considered for the estimation of the averaged impact in that crop. To elaborate on the exposure of the EU agri-food imports due to the occurrence of EWD in the crop exporting countries, we estimate the import share-weighted impact of those events on crop production by considering the import share per exporting country. For each crop, the import share-weighted impact of each EWD type is done by: i) calculating the normalised composite of the estimated impact for each exporting country, ii) multiplying the normalised composite by the corresponding import weight to EU. The weighting scheme allows us to draw direct conclusions of the overall exposure of EU agri-food imports to specific EWD types across exporting countries. This analysis is performed only for the statistically significant impacts of EWD on crop production.

3.3 Results

3.3.1 Assessing the impact of EWD in crop production in the exporting countries relevant for the EU

The Superposed Epoch Analysis (SEA) is applied to the 37 countries (Fig. 3.2b) supplying the five crops for which EU had an import dependency above 35% (soybeans, banana, tropical fruits, coffee, and cocoa). This provides a good sample size of EWD (310 floods, 190 droughts and heatwaves and 56 cold waves) to estimate its impacts on crop production, yield, and harvested area, with importance for the EU food supply regarding exporting countries.

The results on the impact of each type of EWD, including its statistical significance, for aggregated and individual crops, are shown in Figure 3.3. By aggregating the five crops (Fig. 3.3, 1st row) the results are the following ones: during years of floods an average loss of -2% and -1% (CI 95%) is observed for crop production and yield, respectively. During years of droughts and heatwaves, an average impact on the aggregated crop production of -1% (CI 80%) is observed, although for yield and harvested area no significant impact is detected (since the CI is below 80%, i.e. not statistically significant (n.s.)). We did not find statistically significant impacts from droughts and heatwaves in rainfed or in non-tropical systems (Fig. B2).

Overall, considering the different EWD, the aggregation across crops results in smaller average impacts as specific crops can have opposing responses under the same EWD type. We therefore present results individually for each crop hereinafter: (a) Soybeans - both production and yields were negatively affected by floods (-7% and -5%, respectively, CI 95%) and droughts and heatwaves (-4% and -3%, respectively, CI 95%). The average impact of these events in production is estimated in a loss of 555 Mt; (b) Banana production and yield declined by 6% (CI 95%) and 10% (CI 95%), respectively during cold waves, while harvested area was found to increase by 5% (CI 95%). Yields were also negatively impacted by floods, by -5% (CI 95%), while the harvested area increased by 3% (CI 75%). Droughts and heatwaves did not have significant impacts on production, yield, or harvested area; (c) Tropical fruits - production was negatively affected by floods (-4%, CI 95%) and droughts and heatwaves (-3%, CI 95%). The overall impact in years of these events represent a loss of nearly 40 Mt. The low relative negative impact in yield is statistically significant for floods (-1%, CI 80%) and for droughts and heatwaves (-2%, CI 90%); (d) Coffee – a positive response to the EWD types analysed here is detected. Both production and yield increase during droughts and heatwaves by 2% (CI 80% and 90%), respectively, as well as, during cold waves by 4% and 3%, respectively (CI 95%). However, we find a substantial decrease in production and yield in the year after the extreme event (by about 7%, respectively). The effect of flood is not statistically significant for production and harvested area, but yield increased by 1% (CI 80%); (e) Cocoa - we detect significant losses during years of droughts and heatwaves by -6% (CI 75%, equivalent to 6 Mt), -2%, and -3% (CI 90%) for production, vield, and harvested area, respectively.

3.3.2 Assessing the exposure of EU agri-food imports

Soybeans, tropical fruits, and cocoa show the largest impact during EWD years, which can have potential implications for the EU agri-food supply. We therefore weight country-level EWD impacts by EU import shares, which highlights the EU exposure (Fig. 3.4). The combined impact from floods, and from droughts and heatwaves in soybeans production was -11% (-7% from floods and -4.3% from droughts and heatwaves). However, the import share-weighted impact was -9%, meaning that the negative impact is higher in exporting countries from which EU has a lower import dependency. For tropical fruits the picture is different, the arithmetic mean production impact of about -7%, caused by both floods and droughts and heatwaves together, more than doubles to about -16% when weighted by import shares. This indicates that most of the crop loss occurs in exporting countries from which EU has a higher import share. The import share-weighted impact of droughts and heatwaves in cocoa production (-7%) is slightly higher comparing with the average impact in exporting countries (-6%).

Banana and coffee are crops for which there is not a potential implication for the EU agri-food supply. Cold waves negatively impacted banana production (-6%) but those events took place only in Brazil and Belize (Table B5), which together represent only 3% of the EU import share of that crop and thus the weighted banana exposure is marginal. Coffee production increased, on average, during years with cold waves and droughts and heatwaves with an overall gain of nearly 6%. This overall impact slightly decreases to 4% (mostly due to cold waves) when considering the share of EU imports per external supplier countries. This could be explained with the fact that nearly 70% of the cold waves took place in a group of exporting countries representing a lower share of EU coffee imports (8%). Therefore, the weighted coffee gain decreases comparing with the overall gain.

3.4 Discussion

The 12 crops most imported by EU are provided by a diversified foreign market since, for most external suppliers, the dependency on imports is below 10%. Seven of those crops are largely grown in the EU, with a self-sufficiency above 70%. For the other five crops (i.e. soybeans, banana, tropical fruits, coffee, and cocoa) more than 35% of what is consumed in EU is produced in 41 exporting countries.

The SEA revealed significant negative impacts from EWD on soybeans, banana, tropical fruits, and cocoa in exporting countries. Despite a diversified external market, the impacts from EWD in soybeans, tropical fruits, and cocoa, have the potential to negatively affect the EU imports of these crops. For banana the EU import share-weighted impact is negligible. Coffee production shows gains during cold waves but consistent loss in the following year with large inter-annual variability, in general, offsets these effects (see discussion below).

The estimated loss in soybeans production represents an EU import share-weighted impact of -9%, and this negative impact is higher in exporting countries from where EU has a lower import dependency. Nevertheless, such impact may imply a potential decrease on the crop availability in the EU market. Since soybeans is a common substitute of wheat and maize, any fluctuation on its production, and consequently on its prices, may influence the demand and supply chain of the other commodities as well (Ercin *et al.*, 2016).

The impact of floods in soybeans crops have been reported for many areas of the United States of America and the world (Sullivan *et al.*, 2001), and vary according to the crop growth stage during the flood, the duration of the flooding or if in presence, or not, of a flood-tolerant soybean variety (Wu *et al.*, 2017). Such factors were not, however, considered during this first national-level analysis and would be useful for further risk assessments. Flooding can cause physical injuries and anaerobic stress to soybean crops, which in turn can result in a poor vegetative growth and in a low photosynthetic activity, leading to yield loss (Tewari & Arora, 2016). Our estimation on the impact of droughts and heatwaves in soybeans production is in line with Siebers *et al.* (2015) who, by using infrared heating technology in an open-air field experiment, as a way to impose heatwaves on soybeans, showed that short high-temperature stress events resulted in losses in crop production in the Midwest, in the USA.

We found that cold waves and floods lead to increased banana harvested area, indicating that these events might not have been harmful for the entire area, or that the impact was offset as a result of farmer decision when faced by beneficial economic influences such as governmental subsidies (lizumi & Ramankutty, 2015). During years of droughts and heatwaves, no significant impact is observed in banana production, yield or harvested area. Most of the exporting countries that are banana growers are under the influence of a wet tropical climate or use irrigation, which are factors that can offset the impact during those events. As demonstrated by Jägermeyr *et al.*, 2018, at the global scale, heat wave and drought events predominantly affect rainfed over irrigated yields and in case water demand is fulfilled (through irrigation, or as a result of a humid climate), the additional available radiation during those years can offset losses, or even be beneficial for crop growth. This might also contribute to the observed gains in coffee production during droughts and heatwaves.

For tropical fruits, there is a high exposure of EU imports to the impact of EWD. The adverse effect of floods is significant for crop production, yield and (in a less extend) harvested area. This indicates a potential trend for complete crop failure during years with floods. Nonetheless, one year after floods, there are no changes on average production and harvested area, meaning that the crop potentially recovers from the impact.

Cocoa production is substantially affected by droughts and heatwaves, with import share-weighted impact of nearly -7%. This comes with a lagged effect and even higher observed losses in the first year after the event. Such multi-year impact of droughts and heatwaves might affect the recovery of perennial crops and soil moisture, but also changes in planting habits (see discussions in Lesk *et al.* (2016) on cereals). Since the EU completely relies on cocoa imports to satisfy its consumption, a weighted loss of 7% in cocoa production may have consequences to market speculations and may result in economic volatility.

This study assesses EWD impacts on crops selected according to the EU import-dependency ratio. This includes staple crops (such as soybean), which are relevant for caloric consumption in the EU, but also cash crops (such as tropical fruits, coffee, and cocoa). Production anomalies of these crops can therefore potentially reduce caloric availability to some extent in the EU but are not expected to fundamentally impair EU food supply. Import-induced market volatility and resulting market speculations, however, can lead to price spikes. This can have significant adverse effects on food access and, therefore, on food security -- especially for the poor -- which has the potential to exacerbating social unrest.

In order to guarantee the imports of cocoa, tropical fruits, and soybeans, the EU could assist on adaptation schemes in exporting countries, for example by establishing partnerships for research and innovation in crop tolerance to extreme weathers, and by supporting the definition and implementation of disaster risk reduction and management actions, while also supporting the implementation of fair and ethical food policies. This would also be helpful to promote the stability on the production of such crops and, consequently, the stability of incomes in exporting countries, contributing for local food security.

Our study includes assumptions and limitations that include the following: The presented impacts from EWD on crop production, yield, and harvested area are based on a first-order approach at national level with limited data availability. The effect of extreme weather disaster can be much stronger locally, especially in large countries where only part of the cultivated area is being affected. Not all the weather events with impact on agriculture are reported or classified as natural disasters recorded in EM-DAT. Information on the effects of local extreme events are tracked in local statistics only and not available at the international level (Kocur-Bera, 2018). We also did not attribute weights to the magnitude and duration of EWD as there is no such data available, meaning that we treated all events listed in the same way. Moreover, since we aggregated data for each crop from many external supplier countries, it could result in the attenuation of the impact of those events, i.e., losses in one country could be offset by gains among the others. The EWD were not selected based on the crop growth stage, and we did not consider the type of crops varieties in each country (i.e., if tolerant or not to a type of an EWD).

Future research could take advantage of data on EWD that occur in a medium to local scale. It could also be improved if benefited from a detailed georeferenced information on the agro-climatic zones from crop growing regions and on the major agricultural systems (i.e. if irrigated or rainfed).







Production

Yield

Harvested Area

Figure 3.3 Composite figure of EWD influences on observed crop production, yield and harvested area.

The mean event impact (%) refers to the offset in year 0. The number of EWDs composited (n) and the statistical significance of the mean impact is indicated in the legend. Dashes along the y axis indicate the 25th and 75th percentile of the observations. Statistical significance is based on 1000 bootstrap samples and considered not significant above 0.20 (n.s.). The legend thus reads: event type, mean event Results are shown for all crops aggregated, and individually for soybeans, banana, tropical fruits, coffee and cocoa (rows). EWDs include floods (blue), droughts and heatwaves (red) and cold waves (purple). impact, number of observations, level of significance.



Figure 3.4 Impacts of EWD on external EU suppliers and on EU imports per crop category.

Averaged production (yellow marker) and import-weighted (red marker) impact of EWD (%) on soybeans, banana, tropical fruits, coffee and cocoa. The range of impacts in supplier countries is illustrated by yellow bars. Weighted impact is calculated according to the EU crop import shares from supplier countries. Results are shown only for the statistically significant impacts of EWD on crop production among EU suppliers.

3.5 Conclusion

This study highlights the Extreme Weather Disasters (EWD) impacts on specific crops in export-oriented countries and presents the larger implications of such impacts through trade dependencies based on the import share per external supplier country. The focus is on the EU agri-food sector, for which we mapped the external dependency and assessed its potential exposure to EWD. This was done by estimating the overall impact of EWD on production, yield, and harvested area in exporting countries. To the best of our knowledge this is the first study to perform it.

The EU imports between 35-100% of its consumption of soybeans, banana, tropical fruits, coffee, and cocoa, which are grown in 41 countries. Floods, droughts and heatwaves significantly decreased the overall averaged production of soybeans (11%) and tropical fruits (7%), while cocoa production decreased (6%) during years with droughts and heatwaves.

Despite a diversified external market, such losses represent a substantial negative exposure of EU imports to EWD, namely from floods, that cause import share-weighted impacts of -7% (soybeans) and -8% (tropical fruits), while droughts and heatwaves of -3% (soybeans), -8% (tropical fruits), and -7% (cocoa). Since the impacts from floods in tropical fruits, and from droughts and heatwaves in cocoa, have a significant negative impact on the respective crop production, these events potentially imply negative consequence for EU imports. This can potentially lead to market speculations and to higher volatility in commodity prices in the food industries.

To stabilize the EU food imports, the European Union could support the implementation of adaptation schemes in external supplier countries. Improved crop production stability would be associated with important co-benefits regarding the stability of local incomes in exporting countries, and therefore contributing to local food security.

3.6 References

- 1. Brooks, J., & Matthews, A. (2015). Trade Dimensions of Food Security. OECD Food, Agriculture and Fisheries Papers, 77. Retrieved from http://dx.doi.org/10.1787/5js65xn790nv-en
- 2. Clapp, J. (2015). The State of Agricultural Commodity Markets 2015–16. Rome, Italy. https://doi.org/I5222E/1/12.15
- COPA-COGECA. (2003). Assessment of the impact of the heat wave and drought of the summer 2003 on agriculture and forestry - Fact sheets of the Committee of Agricultural Organizations in the European Union and the General Committee for Agricultural Cooperation in the European. Retrieved January 14, 2019, from http://docs.gip-ecofor.org/libre/COPA_COGECA_2004.pdf
- 4. EEA. (2010). Mapping the Impacts of Natural Hazards and Technological Accidents in Europe: An Overview of the Last Decade. Luxembourg.
- 5. EM-DAT. (2018). The Emergency Events Database Universit'e catholique de Louvain (UCL) -CRED, D. Guha-Sapir. Brussels, Belgium. Retrieved January 5, 2019, from www.emdat.be/database

- Ercin, A. E., Chico, D., & Chapagain, A. K. (2016). Dependencies of Europe's economy on other parts of the world in terms of water resources, Horizon2020 - IMPREX project, Technical Report D12.1. Retrieved from https://waterfootprint.org/media/downloads/EUs_vulnerability_to_water_scarcity_and_drought.pdf
- EU. (2018). Monitoring Agri-trade Policy. MAP 2018–1. Agri-food trade in 2017: another record year for EU agrifood trade. Retrieved from https://ec.europa.eu/info/sites/info/files/food-farmingfisheries/news/documents/agricultural-trade-report_map2018-1_en.pdf
- 8. EUROSTAT. (2016). EUROSTAT. Retrieved January 15, 2016, from http://appsso.eurostat.ec.europa.eu/nui/show.do?dataset=DS-016894&lang=en
- 9. FAO. (2016). AQUASTAT Main Database. Food and Agriculture Organization of the United Nations (FAO).
- 10. FAO. (2017). FAOstat, Food and Agricultural Organization. Retrieved January 5, 2019, from http://www.fao.org/faostat/en/
- 11. Hanks, K., & Craeynest, L. (2014). The EU'S 2030 Energy and Climate Change Package Fit for a food and energy-secure world? Oxfam International. Retrieved from www.oxfam.org
- lizumi, T., & Ramankutty, N. (2015). How do weather and climate influence cropping area and intensity? Global Food Security, 4, 46–50. https://doi.org/http://dx.doi.org/10.1016/j.gfs.2014.11.003
- IPCC. (2012). Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation. A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change [Field, C.B., V. Barros, T.F. Stocker, D. Qin, D.J. Dokken, K.L. Ebi, M.D. Cambridge, UK, and New York.
- IPCC. (2014). Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part B: Regional Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Barros, V.R., C.B. Field, D.J. Dokken, M.D. Mastrandre. Cambridge, United Kingdom and New York, NY, USA.
- Jägermeyr, J., & Frieler, K. (2018). Spatial variations in cultivars pivotal to understand global fluctuations in maize and wheat yields. Science Advances, (i), 1–11. https://doi.org/10.1126/sciadv.aat4517
- Kocur-Bera, K. (2018). A safe space of rural areas in the context of the occurrence of extreme weather events— A case study covering a part of the Euroregion Baltic. Land Use Policy, 71, 518–529. https://doi.org/https://doi.org/10.1016/j.landusepol.2017.11.013
- Kottek, M., Grieser, J., Beck, C., Rudolf, B., & Rubel, F. (2006). World Map of the Köppen-Geiger climate classification updated. Meteorol. Z., 15, 259–263. https://doi.org/DOI: 10.1127/0941-2948/2006/0130
- Leng, G., & Huang, M. (2017). Crop yield response to climate change varies with crop spatial distribution pattern. Scientific Reports, 7, 1463. https://doi.org/10.1038/s41598-017-01599-2
- Lesk, C., Rowhani, P., & Ramankutty, N. (2016). Influence of extreme weather disasters on global crop production. Nature, 529(7584), 84–87. https://doi.org/10.1038/nature16467
- Mass, C. F., & Portman, D. A. (1989). Major Volcanic Eruptions and Climate: A Critical Evaluation. Journal of Climate, 2, 566–593. https://doi.org/https://doi.org/10.1175/1520-0442(1989)002<0566:MVEACA>2.0.CO
- Nelson, G. C., Valin, H., Sands, R. D., Havlík, P., Ahammad, H., Deryng, D., & Elliott, J. (2014). Climate change effects on agriculture : Economic responses to biophysical shocks. PNAS, 111(9). https://doi.org/10.1073/pnas.1222465110
- 22. Puma, M. J., Bose, S., Chon, S. Y., & Cook, B. I. (2015). Assessing the evolving fragility of the global food system. Environmental Research Letters, 10(024007). https://doi.org/10.1088/1748-9326/10/2/024007
- Rosenzweig, C., Iglesias, A., Yang, X. B., Epstein, P. R., & Chivian, E. (2001). Climate change and extreme weather events - Implications for food production, plant diseases, and pests. GLOBAL CHANGE & HUMAN HEALTH, NASA Publications, 2(2), 90–104.
- Rosenzweig, C., Jones, J. W., Hatfield, J. L., Ruane, A., Thornburn, K. J., Antle, J. M., ... Winter, J. M. (2014). The Agricultural Model Intercomparison and Improvement Project (AgMIP): Protocols and pilot studies, 35. https://doi.org/10.1016/j.agrformet.2012.09.011

- Siebers, M. H., R. Yendrek, C., Drag, D., Locke, A. M., Acosta, L. R., Leakey, A. D. B., ... Ort, D. R. (2015). Heat waves imposed during early pod development in soybean (Glycine max) cause significant yield loss despite a rapid recovery from oxidative stress. Global Change Biology, 21(8), 3114–3125. https://doi.org/10.1111/gcb.12935
- 26. Sullivan, M., VanToai, T., Fausey, N., Beuerlein, J., Parkinson, R., & Soboyejo, A. (2001). Evaluating On-Farm Flooding Impacts on Soybean. Crop Science, 41(1), 93. https://doi.org/10.2135/cropsci2001.41193x
- Tewari, S., & Arora, N. K. (2016). Environmental Stresses in Soybean Production Soybean Production Volume
 (M. Miransari, Ed.), Environmental Stresses in Soybean Production. Isfahan, Iran: AbtinBerkeh Scientific Ltd. Company, Isfahan, Iran. https://doi.org/https://doi.org/10.1016/B978-0-12-801535-3.00002-4
- Wu, C., Pengyin, C., Hummer, W., Zeng, A., & Klepadlo, M. (2017). Effect of Flood Stress on Soybean Seed Germination in the Field. American Journal of Plant Sciences, 8, 53–68. https://doi.org/http://dx.doi.org/10.4236/ajps.2017.81005

4 Drought and heatwave crop losses tripled over the last five decades in Europe

This study is under review in the journal Environmental Research Letters:

Brás, T.A., Seixas, J., Carvalhais, N., Jägermeyr, J., Drought and heatwave crop losses tripled over the last five decades in Europe.

Author contributions:

T.B. and J.J. conceived the study with contributions from N.C. and J.S.; T.B. and J.J. performed the analysis with valuable contributions from N.C.; T.B. and J.J. wrote the manuscript. All authors discussed and commented on the manuscript.

Abstract

Extreme weather disasters (EWD) can jeopardize domestic food supply and disrupt commodity markets. However, historical impacts on European crop production associated with droughts, heatwaves, floods, and cold waves are not well understood — especially in view of potential adverse trends in the severity of impacts due to climate change. Here, we combine observational agricultural data (FAOSTAT) with an extreme weather disaster database (EM-DAT) between 1961 and 2018 to evaluate EWD responses in European crop production. Using a compositing approach (superposed epoch analysis), we show that historical droughts and heatwaves reduced European cereal yields on average by 9 and 7.3%, respectively, associated with a wide range of responses (inter-quartile range +2 to -23%; +2 to -17%). Non-cereal yields declined by 3.8 and 3.1% during the same set of events. Cold waves led to cereal and non-cereal yield declines by 1.3 and 2.6%, while flood impacts were marginal and not statistically significant. Production losses are largely associated with yield declines, with no significant changes in harvested area. While all four event frequencies significantly increased over time, the severity of aggregated heatwave and drought impacts on crop production roughly tripled over the last 50 years, from -2.2 (1964-1990) to -7.3% (1991-2015). Both the trend in frequency and severity can possibly be explained by different reporting schemes and underlying climate change impacts.

Keywords

Extreme Weather Disasters; Historical Impacts; Climate Change; Crop Responses; European Bioclimatic Zones

4.1 Introduction

The European Union with 28 Member States (EU) is one of the world's major food producers and exporters. EU cropland expands across four main bioclimatic zones (Kottek *et al* 2006) (Table C1), from the hotsummer Mediterranean climate (Csa) to the Subarctic climate (Dfc). The 173 million hectares of the EU agricultural area (i.e., 39% of the EU's total land area) (EUROSTAT, 2019) is used for growing a variety of crops. About 65% of the cultivated area is allocated to cereals (mostly wheat, rye, barley, maize, millet and sorghum), followed by oil crops, olives, vegetables and grapes, roots and tubers, sugar and orchards (Fig. C1a) (FAO 2019a). Cereals and vegetables are the food commodities with the highest production by weight (FAO 2019b) accounting for nearly 30% (26 billion EUR) of the total EU food exports, while maintaining domestic staple food supply.

The EU food system has been affected by a number of extreme weather disasters (EWD; Fig. C1b), which caused significant crop production losses (EM-DAT 2018, Hanel *et al* 2018, Russo *et al* 2015). Most recently, the 2018 heatwave and drought lead to overall cereal production 8% lower than the previous five-year average (DG AGR 2018). These losses caused fodder shortages for livestock and triggered sharp commodity price increases. Soft wheat and barley prices jumped by 34 and 48%, respectively (DG AGR 2018, EC 2018). The 2003 heatwave and drought led to >10% cereal yield declines (particularly wheat and maize) with largest losses in Italy, Germany, Austria, Spain, France, and Portugal (Jägermeyr and Frieler 2018, COPA-COGECA 2003, EUROSTAT 2020).

Depending on human and economic losses, extreme weather events may result in an extreme weather disaster (EWD). The International Disaster Database (EM-DAT) recognizes EWDs if at least 10 people die, 100 or more people are injured, made homeless, or required immediate assistance, or if a country declared a state of emergency, or called for international assistance (EM-DAT 2018). The reporting of EWDs therefore depends on the vulnerability and exposure of human and natural capital, which can confound crop impact analyses.

Extreme weather events significantly influence the year-to-year variability in crop yields at various spatial scales (Ray *et al* 2015, Vogel *et al* 2019, Jägermeyr and Frieler 2018). Climate change is expected to further increase the frequency, intensity, spatial extent, and duration of extreme weather events (IPCC 2012, Russo *et al* 2015, Diffenbaugh *et al* 2017). Future agriculture adaptation challenges are therefore not only linked to changes in the long-term average climate, but particularly to changing weather extremes and interannual

fluctuations in general. In view of large uncertainties associated with long-term average climate change impacts — in some regions crop yields might even benefit (i.e., through the lengthening of the growing season and through the influence of higher CO₂ concentration (Mueller *et al* 2015, Deryng *et al* 2014, Rosenzweig *et al* 2014)) — adverse effects of extreme events on crop production are of increasing concern (Christidis *et al* 2015, Glotter and Elliott 2016, Hov *et al* 2013). However, the historical impacts of extreme weather events on the production of different crops and in different regions remain insufficiently understood. While climate model projections agree that the frequency and severity of extreme weather events are expected to increase under unabated climate change, as of yet there is little quantitative evidence of increasing trends in crop production losses due to such events based on observational records (Lesk *et al* 2016).

Here, we use observational crop statistics from the EU member states (FAO 2019a) in combination with the EM-DAT record (EM-DAT 2018) for a standardised account of historical EWDs to evaluate associated impacts on crop production, yield and harvested area. We consider droughts, heatwaves, floods and cold waves from 1961 to 2018 and separate impacts for different bioclimatic regions. We consider all crops currently grown in the EU as listed by the UN's Food and Agricultural Organization (FAO). Crops are mainly aggregated to two main groups — cereals (CER) and non-cereals (Non-CER) — to avoid limitations due to sample size and to facilitate the evaluation of individual event types for two different time windows. Yet, further analyses consider 12 crop groups individually (Table C2).

We use a Superposed Epoch Analysis (SEA) to estimate the impact of EWDs for different crop groups, climate regions, and time periods. SEA is a statistical method to isolate the average response signal of different events, while reducing noise due to extraneous factors (Lesk *et al* 2016, Jägermeyr and Frieler 2018, Brás *et al* 2019). The SEA analysis is based on detrended national crop statistics, and the statistical significance is tested based on a bootstrapping approach (see Methods for details).

This study addresses the following research questions: (1) How large are historical crop losses associated with different EWD types in Europe? (2) Has the frequency and impact of EWDs increased over the past 50 years? (3) In what climate regions are the EWD impacts most severe?

4.2 Methods

We use national crop production, yield and harvested area obtained from FAOSTAT (2019a) and national EWD occurrence including droughts, heatwaves, floods and cold waves from the EMDAT International Disaster Database (EM-DAT 2018), all from 1961- 2018. Table C3 provides a list of all EWDs considered in this study: 32 droughts, 61 heatwaves, 399 floods and 99 cold waves across the 28 EU countries (Fig. C1b and Table C1). The number of events evaluated for crop impacts is smaller as FAO production, yield, and harvested area data is not available in all countries and years included in EMDAT. The composite analysis

(Fig. 4.1 and 4.2) further decreases the event number as the first and last three years are omitted (before 1964 and after 2015), and because events listed in consecutive years are averaged to a single event datum (see below).

A total of 129 food crops are cultivated in the EU, which we group into the following categories: cereals, including wheat, barley, maize and other cereals, and non-cereals including oil crops, olives, vegetables, grapes, roots and tubers, sugar, orchards, treenuts, citrus, soft fruits and others (Table C2). In case of occasional zero values which we interpret as missing values, in the FAO data record, all other variables (yield, harvested area, production) are set to missing to ensure the same number of records for each variable. All missing values, as well as, countries with reported crop data of less than 10 years are excluded from the analysis.

The averaged EWD impact on crop production, yield and harvested area is estimated through a Superposed Epoch Analysis (SEA), a statistical method is used to isolate the average crop response signal to each EWD type at national level, while reducing noise due to extraneous factors, such as human decision making and agronomic management (Lesk *et al* 2016, Jägermeyr and Frieler 2018, Brás *et al* 2019).

From the national crop data time series, we extract 7-year windows centred on each year of an EWD occurrence, with three years of data preceding and following the event. Each 7-year window is normalised to the average of those 6 adjacent years but by excluding any year coinciding with another EWD of the same type. In order to always have a complete 7-year window, we disregard all the events between 1961-1963 and 2016-2018, in order to normalise each event impact with the average of the 6 adjacent non-disaster years. For calculating the composite signal for two distinct time periods, we consider EWDs between 1964-1990 (crop data 1961-1993) and 1991-2015 (crop data 1988-2018).

If an EWD of the same type occurs again in a subsequent year, we average the data across all years with successive EWD occurrence (e.g., multi-year drought) to produce a single disaster year datum, which is then surrounded by the 6 adjacent years. This procedure results in a reduction in the total number of events since the average of sequential EWD years of same type is considered as one event. After normalisation, we calculate the composite vector, which is the column-based mean of all 7-year windows for a specific EWD type, crop category or climate zone. The composite vector thus always consists of seven elements. We detrend the composite vector by subtracting its linear regression line and subsequently add the composite vector mean. The fourth element of the detrended composite vector is the event signal: the average normalised EWD impact. To calculate the detrended composite signal across different crops — and for droughts and heatwaves together, as pointed out below — 7-year windows are grouped together to calculate the mean composite signal.

The statistical significance of the EWD composite signal is assessed based on bootstrap replicates, obtained by resampling the full 7-year windows. Specifically, each of the 7-year windows is resampled with

replacement (column-based) 1000 times before normalising each year with the average of the 6-adjacent non-disaster years and before calculating the average composite vector to create 1000 different composite vectors, which will then be detrended. Resampling with replacement means that a particular observation from the original data set could appear multiple times in a given bootstrap sample (which has the same number of elements in each original data set). We repeat this process to obtain 1000 detrended composite signals. This represents an empirical bootstrap distribution of the mean impact during EWD years, which we use to test the normality hypothesis and to derive confidence interval. We test if the empirical bootstrap distribution is statistically different from the normal distribution using the Kolmogorov-Smirnov test with a significance level of 0.05 (Öner & Deveci Kocakoç, 2017). If data approximates a normal distribution, we assess the statistical significance of the mean event impact, which is the deviation of the detrended composite signal from 1 in year 0. To test the null hypotheses (i.e., the detrended composite signal equals 1), we first calculate the confidence interval (CI) of the empirical bootstrap distribution for different significance levels. If both end points of the CI are smaller (or larger) than 1 and if the composite signal lies within the CI, it is considered statistically significant at the respective significance level, i.e. 5%, 10% and 20%, and not significant if ≥20%. For further details, see Brás et al. (2019), Leng & Huang (2017) and Wong & Easton (1980).

We first calculate the detrended composite signal of droughts, heatwaves, floods and cold waves for production, yield and harvested area data using the entire time series from 1964-2015, separating the two main crop categories cereals and non-cereals. In a second step we calculate the composite signal for two time slices (i.e. 1964-1990, 1991-2015) for cereals, non-cereals and for both categories aggregated. To improve statistical significance, droughts and heatwaves are grouped to evaluate the composite signal i) separately for the first and second time slices, ii) for the 12 crop categories individually, and iii) in each Koeppen-Geiger climate zone. The analysis by climate zone is done by aggregating all countries according to its dominant Koeppen-Geiger classification (Table C1).

Since the FAO crop data contain many more non-cereal crop categories than cereal categories, we calculate the average cereal and non-cereal signal, respectively, in each country for each EWD, before aggregating both. This way cereals and non-cereals receive the same weight when combined in the overall composite signal (Fig. 4.2).

In addition to the composite signal of multiple events, we evaluate the trend in EWD frequency (Fig. 4.4), and the trend across normalised crop production anomalies over time (1961-2018) for each event type (Fig. 4.3 and Fig. C4). This is done by first calculating the sum of annual cereal production at country level. We then detrend each country-level cereal time series by subtracting its second order polynomial; these anomalies are then normalised by dividing with its standard deviation. Normalised anomalies are calculated separately for cereal and non-cereal crops, and also stratified by individual climate zone. The statistical significance of time trends (for both event frequency and production anomalies) is assessed by fitting a

linear regression and testing its slope parameter for significance using the *t*-test. Significance levels are classified according to the following thresholds: *** if *p*-value < 0.05, ** if *p*-value < 0.1, * if *p*-value < 0.20, and n.s. (not significant) if *p*-value >= 0.20.

4.3 Results

4.3.1 EU crop response to extreme weather disasters

Between 1961 and 2018, the EM-DAT record lists a total of 591 events across the 28 European countries (Fig. C1b), specifically 32 droughts, 61 heatwaves, 399 floods, and 99 cold waves (Table C3). On average, droughts and heatwaves reduced EU cereal yields by 9% (inter-quartile range: +2 to -23%, 28 events) and 7.3% (+2 to -17%, 47 events), respectively (Fig. 4.1). The same events reduced non-cereal yields by 3.8% (+6 to -13%) and 3.1% (+4 to -12%), respectively. Cold waves led to cereal and non-cereal yield declines by 1.3% (+7 to -9%, 60 events) and 2.6% (+6 to -11%), while flood impacts on yields were not statistically significant for cereals, and marginal (-0.4%) for non-cereal crops. Yield observations are not indicating a lagged yield level response in the year following reported EWDs, except heatwaves, which are followed by a year with increased cereal yield levels (Fig. 4.1). Due to FAO crop data availability and methodological requirements, the number of events evaluated for crop impact is lower than the original EM-DAT list (see Methods).

Changes in crop production are largely driven by yield declines, with comparatively small and not statistically significant changes in harvested area (Fig. 4.1). During flood and cold wave years, non-cereal harvested area decreased by 1.8%, which generally indicates the abandoning of areas hardest hit (lizumi and Ramankutty 2015).

Overall, cereals — covering two thirds of European cropland — show consistently larger losses associated with droughts and heatwaves compared to non-cereal crops. This can be explained by generally widespread irrigation in non-cereal crops. Combined drought and heatwave production responses for cereals are: wheat (-11.3%), barley (-12.1%) and maize (-12.5%). Non-cereals include: oil crops (-8.4%), olives (-6.2%), vegetables (-3.5%), roots and tubers (-4.5%), sugar beet (-8.8%), among others (Table 4.1). We combine droughts and heatwaves into a single event category to overcome limitations due to sample size in order to assess the statistical significance of the EWD impact for these individual crop groups, but also during shorter time periods (Section 4.3.2), and different bioclimatic regions (Section 4.3.3).

$\begin{array}{c ccccccccccccccccccccccccccccccccccc$			~		~
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Harvested area		3 -2 -1 0 1 2 -1.2% (n.s., n = 28) -1.2% (n.s., n = 47) -0.2% (n.s., n = 141) -0.0% (n.s., n = 60)		 3 -2 -1 0 1 2 3 Year from event 9 -0.5% (n.s., n = 28) -0.8% (*, n = 47) -1.8% (***, n = 141) 1.8% (***, n = 60)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	U	1.10 1.00 0.90 0.80		1.10 1.05 1.00 1.00 0.95 0.95 0.85	cold
on n = 28) n = 47) n = 47)	b Yield	1.10 0.90 0.90 0.90 0.90 0.90 0.90 0.90	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	1.10 1.05 0.95 0.90 0.90	-3 -2 -1 0 1 2 3 Year from event -3.8% (***, n = 28) -3.1% (***, n = 47) -0.4% (*, n = 141) -0.2.6% (***, n = 60) Idhts - heatwaves floods -
Production -2 -1 0 -2 -1 0 -8.5% (*** -0.4% (n.s. -0.4% (n.s. -2.1% (n.s. -2.1% (*** -2.1% (*** -3.9% (*** -3.9% (*** -4.1% (***)	Production		$\begin{array}{cccccccccccccccccccccccccccccccccccc$		 3 -2 -1 0 1 2 3 Year from event 9% (***, n = 28) -3.9% (***, n = 47) -2.1% (***, n = 141) -4.1% (***, n = 60)
	ø	0.90		1.10 1.05 1.00 0.95 0.95 0.95	
0.95 · · · · · · · · · · · · · · · · · · ·		Normalised composite		Normalised composite	

Figure 4.1 Impacts of Extreme Weather Disasters on European crops.

included (n). Dashes along the y-axis indicate the 25^{th} and 75^{th} percentile of the observations. Statistical significance is based on 1000 bootstrap samples (see and cold waves (grey). The composite record between 1964 and 2015, based on 7-year time windows of country-level data centred on the respective event. The (*** if alpha < 0.05, ** if alpha < 0.10, * if alpha \ge 0.2) and the number of events production (first column), yield (second column) and harvested area (third column) are shown for cereal (CER, first second row), individually for: droughts the composite signal from 1 in year 0, each plot, along with its significance level *alpha* < 0.20, n.s. for not significant if Composite impacts in terms of crop row) and non-cereal crops (Non-CER, (red); heatwaves (orange); floods (blue); analysis includes all EWDs in the EM-DAT mean event impact (%) is the deviation of highlighted in the legend box underneath Methods).

4.3.2 Crop impact and frequency of extreme weather disasters over time

The impact of droughts and heatwaves on European crop production roughly tripled between the first (1964-1990) and second (1991-2015) half of the observation record: from -2.2 to -7.3%. While cereals show larger absolute losses in both time periods (increasing from -3.6 to -9.8%), non-cereals' impacts increase more than fivefold from -0.9 to -4.8% (Fig. 4.2). This aggravating signal in cereals is largely driven by more severe yield losses: cereal yield declines doubled from -4.4 to -8.9%. For non-cereal crops, however, yield declines changed less substantially (from -3.2 to -3.7%), but additional harvested area declines (1.8 to -1.4%) cause steep changes in overall production (Fig. 4.2e,f). Importantly, while these numbers reflect the average impact across all recorded events, Figure 4.2 also illustrates that the most severe events become disproportionally more severe. For example, the 25 percentile of production impacts decreased from -8.1 to -13.5%, whereas the 75 percentiles only changed from 4.1 to 0.7% (Fig. 4.2a,b).

For floods (Fig. C2) and cold waves (Fig. C3) the results draw a slightly more complex picture. While we find somewhat less severe production declines for both event types among more recent observation, for cold waves this signal is driven by much less affected harvested area despite increasing yield losses (Fig. C3d,f). For floods on the other hand, the production signal is driven by less severe yield impacts in the second time period (Fig. C2c,d), which is in line with an overall positive trend across flooding yield declines presented next.

Observations show a negative trend in normalised anomalies of cereal production over time for all evaluated event types except floods (Fig. 4.3). Even though the drought category comes with the lowest number of cases, the trend is statistically significant (at the 0.05 level) and indicates increasing annual cereal production losses by more than 3%, the steepest decline among the four EWD types. For heatwaves the trend line is more marginal and not significant. Flood events indicate a slightly positive trend cereal production anomalies that is also not statistically significant. Cold waves on the other hand show a surprisingly steep and significant negative trend. No significant trends are found for non-cereal crops (Fig. C4).

After all, droughts, heatwaves, floods, and cold waves became more frequent over the last five decades, all following statistically significant trends (Fig. 4.4). Results indicate an annual increase in event frequency of 1% (droughts), 6% (heatwaves), 29% (floods), and 10% (cold waves). The number of reported droughts and heatwaves increased from 13 in the first half of the observation period to 62 in the second half (Fig. 4.2). Similarly, there were 38 floods and 4 cold waves on record in the first half, and 103 and 56 in the second half, respectively (Fig. C2 and C3).

70



Figure 4.2 Drought and heatwave crop responses in the first and second half of the observation record. The composite impact of cereal (CER), non-cereal (Non-CER) and both categories aggregated (Combined) is shown for production (1st row), yield (2nd row) and harvested area (3rd row), and is separated for the time slices 1964-1990 (1st column) and 1991-2015 (2nd column). Droughts and heatwaves are aggregated to avoid limitations due to sample size. Significance levels are as in Figure 4.1. Similar plots for floods and cold waves are shown in Figures C2 and C3.

4.3.3 Severity of extreme weather disasters across different climate regions

Between 1964 and 2015, the average cereal yield response to both droughts and heatwaves combined, shows largest relative losses (-12.8%) in warm-summer humid continental climates (Köppen-Geiger zone Dfb, see Table C1) covering eastern European countries such as Romania, Slovakia, Estonia, and Austria (Table 4.1). The response in temperate oceanic climates (Cfb; remaining central European countries plus France and the United Kingdom) is -6.6% and in hot-summer Mediterranean climates (Csa; Portugal, Spain, Italy, Greece, Cyprus and Malta) cereal yield declines by -6.9%. Overall production declines are mostly driven by yield changes with comparatively small (and mostly not significant) changes in harvested area. While countries in the Csa climate zone show smallest average production losses for wheat and not

significant impact for maize, they show largest losses for barley (as well in yield and harvested area), even though barely is most commonly grown in central and northern European countries (Table 4.1).

Non-cereal crops also show largest yield and production losses in the Cfb and Dfb climate zones, namely staple crops such as vegetables, sugar, soft fruits, roots and tubers (Table 4.1). Olives, a relevant cash crop in the EU also show production losses in the Cfb region (-13.2%), driven by declines in yield (-11.3%) and harvested area (-2.8%). We did not find significant signals among countries in the subarctic climate zone (Dfc; Sweden and Finland).

While floods do not show a significant effect on cereal yield at overall European level (Fig. 4.1), in Cfb countries, barley (largely grown in temperate central and northern EU countries) exhibit significant yield declines by 3.4 which is offset by a positive response in maize (largely grown in drier Mediterranean countries) by 5.3% (Table C4). Years with flood events are likely to have a generally wetter growing season, which might benefit overall maize growth especially in more semi-arid climates. Cold waves have a negative effect on crop production especially across continental Dfb climates: wheat -11.1%; barley -15.4%; maize - 7.8%; oil crops -15.9%; vegetables -4.6%; grapes -9%; treenuts -26.6%, largely associated with yield declines (Table C5). But the response in Cfb countries is largely positive for cereals, which could be explained by faster achievement of vernalization requirements of winter crops in colder years (Jägermeyr *et al* 2020).

4.4 Discussion

Here we use observational data to systematically evaluate European crop responses to droughts, heatwaves, floods, and cold waves included in the EM-DAT disaster data base. While the frequency of reported EWDs increased for all four event types, results suggest that impacts associated with droughts and heatwaves on European crop production roughly tripled over the observation period starting in 1964. Even though there are several issues linked to using disaster events as a metric for extreme weather event impact analysis, our findings support the hypothesis that climate change is among the factors driving increased crop losses due to extreme weather events in the historical data record.

Especially droughts show increasing crop losses over the last five decades, most prominently for cereal production. These findings are in line with evidence reported by the Intergovernmental Panel on Climate Change (IPCC) that Southern Europe is experiencing more intense and longer droughts (Bocchiola *et al* 2013). Lesk *et al.*(2016) also find increasing drought-related crop losses for cereals between 1964-2007 at the global level. The IPCC (2012) and other more recent studies (Pfleiderer *et al* 2019, Christidis *et al* 2015, Stott 2016, Coumou and Rahmstorf 2012) find that heatwaves are becoming more severe in most parts of Europe. Our results indicate only a slightly negative and not significant trend in the heatwave response, which might be explained by the fact that expanding irrigation helps to attenuate adverse heatwave impacts

especially among Central European and Mediterranean countries. Irrigation can largely mitigate adverse heatwave impacts by cooling surface temperatures and thus reducing direct heat damage, but also resulting water stress impacts through maintaining increased soil moisture requirements (Jägermeyr and Frieler 2018, Vogel *et al* 2019, Leng 2017, Leng and Hall 2019, Troy *et al* 2015). According to AQUASTAT statistics (FAO 2016), nearly 28% of European cereal area is under irrigation, predominantly in Cfb and Csa regions. An additional factor that can help explain the missing significance in the heatwave trend line is that the EM-DAT time series is substantially shorter for heatwaves (starting in 1985) than for the other events (droughts start in 1976, floods in 1965, cold waves in 1971).

We evaluate the impact of each event individually, not integrated over time. An increase in event frequency thus does not affect the composite severity signal in this analysis. Observational evidence, however, shows an increase in the frequency of extreme weather events in Europe, especially for heatwaves, and most strongly in the Mediterranean region (IPCC 2012). Our findings also indicate a strong increase in the frequency of EWD for droughts, heatwaves, floods, and cold waves. The recently published report of the United Nations Office for Disaster Risk Reduction (UNDRR) (2020) supports our findings showing a sharp increase in worldwide heatwaves (+232%), droughts (+29%), and floods (+134%) over the last 20 years. While the mortality rate of these events decreased, they are associated with a significant increase in economic damage and the number of people affected. The increase in event numbers may partially be explained by better recording and reporting, yet much of it is attributed to a significant rise in the number of climate-related disasters (UNDRR/CRED 2020).

An extreme weather event can become an EWD if a specific human or economic damage occurs. The EM-DAT data base is a standardised record of large EWD and thus commonly used for advancing the understanding of their impact, but the linkage to capital loss weakens the direct linkage to the weather signal. The increased EWD frequency is therefore a confound signal of an increased extreme weather event number, and increased capital exposure and vulnerability to such events.

Climate change is leading to fewer extremely cold days and nights on average (EASAC 2013). On the other hand, climate change is also expected to increase general weather variability, for example through more stationary atmospheric wave pattern that can cause intensified heatwaves, but also cold snaps (Kornhuber *et al* 2019, Mann *et al* 2018). We expect that the increasing trend in cold wave events found in the EM-DAT record (Fig. 4.4d) is likely a combination of increased event reporting and underlying climate change. The increasing frequency of flooding events is in line with other studies (e.g. Kundzewicz *et al*.(2017)).

Additional limitations associated with using a national EWD record for agricultural impact analysis include the following aspects. (1) affected areas in a specific country accounting for the EWD damage might not coincide with the crop production areas and is therefore not always representative for the agriculture sector, which is especially important in large countries such the U.S. or Russia. (2) related to the confounded frequency trend, not all extreme weather events causing crop production losses are reported in EM-DAT. Therefore, the number of extreme weather events will be higher than the associated EWD reported. (3) reported EWDs are not necessarily occurring during the crop growing period, but anytime within the calendar year, which likely contributes to an underestimation of the overall impact signal. (4) no weights are attributed to individual EWDs accounting for the magnitude or duration of events. These points are reflected in the wide range of impacts shown in the 25th and 75th percentiles (Fig. 4.1 and 4.2) and are discussed in Brás el al. (2019).

The aggregation of data to the European level can mask more severe regional impacts as losses in one region can be offset by gains in others, such as seen for cold waves in Table C5. But the limited number of events and countries on record hamper finer-grained analyses in many cases as the composite impact signal becomes statistically insignificant without a sufficient number of cases (Tables C4-C5). In follow-up studies some of these limitations could be overcome by using spatially explicit and index-based event metrics focused on actual cropland areas and different agricultural system. That said, quantifying EWD impacts as conducted here is a different, equally important contribution to understanding food system vulnerabilities.

Droughts led to higher European cereal yield and production losses than heatwaves, while for non-cereal crops the impacts were similar between both events. The geographic difference in EWD impacts with larger losses in the Warm-summer humid continental climates (i.e. Dfb region) and smaller losses in Southern Europe (i.e. Csa regions) but also in countries with a temperate oceanic climate (i.e. Cfb) can be possible explained by the share of cropping area under irrigation.

In Csa countries, 87 and 9% of the area for maize and wheat production is irrigated, respectively. In the Cfb region maize is irrigated to 19% and in the Dfb to 2%, and wheat is generally not irrigated (FAO 2016) (AQUASTAT records from 2003-2011). As an example for non-cereals, Olives are irrigated to 20% in Csa regions, and only to 4% in Cfb regions (FAO 2016). In general terms, the area under irrigation could be expanded in Europe as a measure to alleviate exposure to extreme weather events. But substantial investments would be required (Elliott *et al* 2014), energy consumption would increase (Daccache *et al* 2014) and the cost of crop production and consequently food prices would potentially be affected. Importantly, the evaluation of the irrigation potential must be guided by water sustainability standards such as the European Water Framework Directive (European Comission 2000). Moreover, traditional and sustainable water management practices such as conservation tillage, organic mulching, and water harvesting for supplemental irrigation during dry spells are shown to offer large and synergistic opportunities to buffer impacts of extreme weather in both rainfed and irrigated systems (Jägermeyr 2020, Rosa *et al* 2018, Jägermeyr *et al* 2016, Rost *et al* 2009).

floods (c), and cold waves (d) listed in the EMDAT record (EM-DAT 2018) until 2018 statistics available). Cereal production is according to the Koeppen-Geiger climate Figure 4.3 Cereal production anomalies shown as the sum of all cereal production in a specific country. Countries are colored Dfc Subarctic (see Table C1). The straight line the regression line; its slope parameter and significance level are shown If *p-value* < 0.1; * if *p-value* < 0.2; n.s. for not significant if ≥ 0.2). The number of events (n) is indicated in the title. A similar plot for nonyears with droughts (a), heatwaves (b), currently the last year with FAO yield zone: Cfb - Temperate oceanic, Csa - Hot-Normalised anomalies are shown for all - Warmin the top-left corner (*** if *p-value* < 0.05; ** cereal production is shown in Figure C4. humid continental, and summer Mediterranean, Dfb during years of reported EWDs. ndicates summer



75



Figure 4.4 Number of annually reported EWDs in Europe.

The number (n) of country-level droughts (a), heatwaves (b), floods (c) and cold waves (d) in the EM-DAT record (EM-DAT 2018) are shown between 1961 and 2018. The solid line indicates the regression line, its slope parameter and significance level are shown in the top-left corner (*** if p-value < 0.05; ** if p-value < 0.1; * if p-value < 0.2; n.s. for not significant if \geq 0.2). The dashed line represents the 20-year moving average.

Table 4.1 Composite drought and heatwave impacts by crops and climate region.

Observed production, yield and harvested area impacts (%) associated with droughts and heatwaves (combined to overcome limitations due to sample size) are separated for the EU level and (%). Average event impacts are shown (red if negative, green if positive) if statistically significant (*** if alpha < 0.05, ** if alpha < 0.10, * if alpha < 0.2, n.s. for not significant if alpha ≥ 0.2); n.n. if for each Koeppen-Geiger (KG) region (Kottek et al 2006) between 1964 and 2015. Results are shown for major crop categories, ordered by the respective fraction of the total EU cropping area empirical bootstrapped distribution of the normalised mean is not normal. Statistical significance is based on 1000 bootstrap samples (see Methods). Column n indicates the number of events. Blank cells mean that a crop is not grown in the respective KG region. Impacts for floods and cold waves are shown in Tables C4 and C5.

mate	c	14	14	14	14	14	14	14		14	11	14	14	13	∞		13	14
nid continental cli	Harvested area	-3.8 *	-5.8 ***	1.3 n.s.	-2.5 n.n.	-4.5 n.s.	-1.5 n.s.	-3.1 n.s.		1.1 n.s.	0.1 n.n.	-2.4 *	-2.5 n.s.	4.9 ***	9.8 ***		-10.5 ***	-14.6 ***
-summer hun	Yield	-12.8 ***	-12 ***	-13.7 ***	-17.6 ***	-12 ***	-5.9 ***	* 9-		-8.4 ***	4.5 n.s.	-9.2 ***	-11.4 ***	5.5 **	-7.5 ***		-4.2 ***	-8.7 ***
Dfb - Warm	Production	-15.6 ***	-17 ***	-13.2 ***	-18.6 ***	-15.2 ***	-7.4 ***	-13 ***		-7.7 ***	4.7 n.s.	-10.6 ***	-11.6 ***	15.1 n.n.	1.8 n.s.		-14.2 ***	-21.2 ***
te	c	28	28	28	23	28	28	28	27	28	28	28	23	28	28	28	28	28
diterranean clima	Harvested area	-0.3 n.s.	1.4 n.s.	-4.5 **	1 n.s.	0 n.s.	0.1 n.s.	1.7 n.s.	-2.3 ***	0.5 n.s.	0.3 n.s.	0.2 n.s.	1.9 n.s.	2.1 ***	-1.8 **	-2.4 ***	-2.1 *	-2.8 n.n.
summer Me	Yield	-6.9 ***	-11 ***	-15.5 ***	-3.4 ***	-5 ***	-1.6 ***	-2.9 n.s.	-0.2 n.s.	-2.6 ***	-1.8 n.s.	1.3 *	-3.4 ***	-1.1 n.s.	1.8 n.s.	-1.7 n.s.	-2.9 ***	0.9 n.s.
Csa - Hot-	Production	-6.8 ***	-9.5 ***	-19.2 ***	-2.5 n.s.	-4.5 **	-1.4 ***	-5.5 n.s.	-3 n.s.	-1.3 ***	-1.6 n.s.	2.4 n.s.	-2.5 n.s.	0.9 n.s.	0.6 n.s.	-3.2 **	-4.2 n.n.	-4 n.n.
	c	33	33	33	30	33	33	33	12	33	32	33	32	33	26	11	33	32
ceanic climate	Harvested area	1.1 n.s.	-3.3 ***	1.7 *	-0.8 n.s.	2 *	-1.3 **	-1.9 n.s.	-2.8 ***	-0.9 n.s.	-0.7 n.n.	0.8 n.s.	-3.4 ***	-2.2 ***	-2.1 n.s.	-3.1 n.s.	0.2 n.s.	-3.5 **
Temperate o	Yield	-6.6 ***	-7.3 ***	-7.4 ***	-16.4 ***	-4.9 ***	-4.5 ***	-7.6 ***	-11.3 *	-3.5 ***	-0.4 n.s.	-11.1 ***	-11.8 ***	-1.8 n.s.	-2.8 *	-8.7 ***	-4.9 ***	-4.4 **
Cfb -	Production	-6.4 ***	-10.5 ***	-5.7 ***	-17.2 ***	-4.2 ***	-5.4 ***	-8.2 ***	-13.2 **	-4.1 ***	-0.9 n.s.	-11 ***	-14.2 ***	-3.8 ***	-4.2 *	-11.4 ***	-5.3 ***	-5.5 n.n.
	c	75	75	75	67	75	75	75	39	75	71	75	69	74	62	39	74	74
28	Harvested area	-0.3 n.s.	-2 ***	-0.7 n.s.	-0.6 n.s.	0.1 n.s.	-0.7 *	-1 n.s.	-2.4 ***	0 n.s.	-0.2 n.s.	0.1 n.s.	-1 n.s.	0.8 n.s.	-1.1 n.s.	-2.6 ***	*** **	-5.5 ***
pean Union-2	Yield	-7.9 ***	-9.6 ***	-11.6 ***	-12.2 ***	-6.3 ***	-3.4 ***	-5.7 ***	-3.6 n.s.	-3.9 ***	-0.2 n.s.	-4.8 ***	-8.2 ***	-0.6 n.s.	-0.4 n.s.	-3.4 ***	-4 ***	-3.1 **
Euro	Production	-8.3 ***	-11.3 ***	-12.1 ***	-12.5 ***	-6.4 ***	-3.9 ***	-8.4 ***	-6.2 **	-3.5 ***	-0.3 n.s.	-4.5 ***	-8.8 ***	0.9 n.s.	-1 n.s.	-5.1 ***	-6.9 ***	-8.2 ***
	% cropping area	65.0	29.5	14.1	10.2	11.2	35.0	13.4	5.5	4.5	3.7	2.3	1.7	1.6	1.1	0.6	0.4	0.2
	Crops	CEREALS	Wheat	Barley	Maize	Other cereals	NON-CEREALS	Oil crops	Olives	Vegetables	Grapes	Roots & tubers	Sugar	Orchards	Treenuts	Citrus	Soft fruits	Other crops

This study highlights that droughts and heatwaves are particularly harmful for cereal production, with a loss twice as high as for non-cereal crops, especially in Mediterranean and Eastern European countries, but also in Central Europe with similar relative losses in both crop categories. Production losses of wheat in Central and Eastern Europe, as well as of barley in the Mediterranean region, are largely associated to yield declines but also to a reduction in harvested area, which is an indicator for partial crop failure (lizumi and Ramankutty 2015). On the other hand, barley production in Cfb is associate to yield declines but also to an increase in the harvested area, suggesting that farmers may have offset production losses by expanding the harvested area. This is an observed behaviour incentivised by crop insurances and governmental subsidies (lizumi and Ramankutty 2015).

Cereals are especially relevant in terms of caloric food consumption (providing > 60% of the energy intake (FAO 1997)), but also for providing feed to maintain the livestock sector. In 2014, the EU represented 13% of global cereal production (Knox *et al* 2016), contributing 24% of global cereal exports (FAO 2019a) (mainly originated from Dfb and Cfb climate zones, while countries in the Csa climate zone only produce 81% of their cereal demand resulting in a net import of cereals (FAO 2019a)). The EU contributes for example almost 50% of the global sugar production (Knox *et al* 2016), 70% of the world olive oil exports (International Olive Council 2018), but also to nearly 50% of the world's wine (Wine Institute 2017). The size and trend of extreme event impacts on both cereal and non-cereal production is of concern as it can cause ripple effects in the global food trade system and affect food prices and food availability worldwide (e.g. Puma *et al*.(2015), Jägermeyr *et al*.(2020)). Such cascading effects are particularly relevant in already food insecure regions.

Future projections suggest an increase in summer dryness in most parts of Europe, with longer and more intense heatwaves and droughts (EASAC 2013, IPCC 2012, Christidis *et al* 2015). Especially the Mediterranean region is likely to experience severe multi-year droughts (Guerreiro *et al* 2017). The historical agricultural losses associated to EWD illustrated in this study, especially for droughts, are therefore expected to further increase in the future.

4.5 Conclusion

Agricultural impacts associated with droughts, heatwaves, floods, and cold waves are not well understood across larger spatial scales, especially in view of potential adverse trends due to climate change. Here, we use a superposed epoch analysis to estimate average per-disaster crop losses across Europe due to reported extreme weather disasters from 1964-2015. While the frequency of all four event types significantly increases over time, our results suggest that the average crop production impact of droughts and heatwaves has tripled over the last fifty years. Even though using a weather disaster record for crop impact analyses has limitations, it offers a unique and standardized metric indicating that climate change is already driving increasing crop losses in observational records. Our study contributes to the discussion of strategies and priorities in view of improving food system resilience.

4.6 References

- 1. Bocchiola D, Nana E and Soncini A 2013 Impact of climate change scenarios on crop yield and water footprint of maize in the Po valley of Italy Agricultural Water Management 116 50–61 Online: http://dx.doi.org/10.1016/j.agwat.2012.10.009
- 2. Brás T A, Jägermeyr J and Seixas J 2019 Exposure of the EU-28 food imports to extreme weather disasters in exporting countries *Food Security* Online: http://link.springer.com/10.1007/s12571-019-00975-2
- 3. Christidis N, Jones G S and Stott P A 2015 Dramatically increasing chance of extremely hot summers since the 2003 European heatwave *Nature Climate Change* 5 46–50 Online: https://doi.org/10.1038/nclimate2468
- COPA-COGECA 2003 Assessment of the impact of the heat wave and drought of the summer 2003 on agriculture and forestry - Fact sheets of the Committee of Agricultural Organizations in the European Union and the General Committee for Agricultural Cooperation in the European Online: http://docs.gip-ecofor.org/libre/COPA_COGECA_2004.pdf
- 5. Coumou D and Rahmstorf S 2012 A decade of weather extremes *Nature Climate Change* 2 491–6 Online: https://doi.org/10.1038/nclimate1452
- 6. Daccache A, Ciurana J S, Rodriguez Diaz J A and Knox J W 2014 Water and energy footprint of irrigated agriculture in the Mediterranean region *Environmental Research Letters* 9 12
- Deryng D, Conway D, Ramankutty N, Price J and Warren R 2014 Global crop yield response to extreme heat stress under multiple climate change futures *Environmental Research Letters* 9 034011 Online: http://stacks.iop.org/1748-9326/9/i=3/a=034011?key=crossref.dbe03166b591fb71af81975aae0cc1ee
- DG AGR 2018 Short-term outlook for EU agricultural markets in 2018 and 2019 (Brussels) Online: https://ec.europa.eu/info/sites/info/files/food-farming-fisheries/farming/documents/short-term-outlook-autumn-2018_en.pdf
- Diffenbaugh N S, Singh D, Mankin J S, Horton D E, Swain D L, Touma D, Charland A, Liu Y, Haugen M, Tsiang M and Rajaratnam B 2017 Quantifying the influence of global warming on unprecedented extreme climate events *Proceedings* of the National Academy of Sciences 114 4881 LP – 4886 Online: http://www.pnas.org/content/114/19/4881.abstract
- 10. EASAC 2013 Trends in extreme weather events in Europe: implications for national and European Union adaptation strategies (Halle, Germany: German National Academy of Sciences Leopoldina 2013) Online: www.easac.eu
- 11. EC 2018 COMMODITY PRICE DASHBOARD Online: https://geotee.gr/lnkFiles/20181009134118_4.pdf
- Elliott J, Deryng D, Müller C, Frieler K, Konzmann M, Gerten D, Glotter M, Flörke M, Wada Y, Best N, Eisner S, Fekete B M, Folberth C, Foster I, Gosling S N, Haddeland I, Khabarov N, Ludwig F, Masaki Y, Olin S, Rosenzweig C, Ruane A C, Satoh Y, Schmid E, Stacke T, Tang Q and Wisser D 2014 Constraints and potentials of future irrigation water availability on agricultural production under climate change *Proceedings of the National Academy of Sciences* 111 3239 LP – 3244 Online: http://www.pnas.org/content/111/9/3239.abstract
- 13. EM-DAT 2018 The Emergency Events Database Universit'e catholique de Louvain (UCL) -CRED, D. Guha-Sapir. Brussels, Belgium Online: www.emdat.be/database
- 14. European Comission 2000 Directive 2000/60/EC Official Journal of the European Communities L327 1–82
- 15. EUROSTAT, 2019. Agri-environmental indicator irrigation [WWW Document]. eurostat Statistics Explained. URL https://ec.europa.eu/eurostat/statistics-explained/index.php/Agri-environmental_indicator_-_irrigation
- 16. EUROSTAT 2020 Selling prices of crop products (absolute prices) annual old codes data from 1969 to 2005 Online: http://appsso.eurostat.ec.europa.eu/nui/show.do?dataset=apri_ap_hcouta&lang=en
- 17. FAO 2016 AQUASTAT Main Database. Food and Agriculture Organization of the United Nations (FAO)
- 18. FAO 2019a FAOSTAT Online: http://www.fao.org/faostat/en/
- FAO 2019b The State of the World's Biodiversity for Food and Agriculture ed J B & D Pilling (Rome, Italy: FAO Commission on Genetic Resources for Food and Agriculture Assessments) Online: http://www.fao.org/3/CA3129EN/CA3129EN.pdf

- 20. FAO 1997 The State of the World's Plant Genetic Resources for Food and Agriculture Online: http://www.fao.org/3/aw7324e.pdf
- 21. Glotter M and Elliott J 2016 Simulating US agriculture in a modern Dust Bowl drought *Nature Plants* 3 16193 Online: https://doi.org/10.1038/nplants.2016.193
- 22. Guerreiro S B, Kilsby C and Fowler H J 2017 Assessing the threat of future megadrought in Iberia INTERNATIONAL JOURNAL OF CLIMATOLOGY 37 5024–5034
- 23. Hanel M, Rakovec O, Markonis Y, Máca P, Samaniego L, Kyselý J and Kumar R 2018 Revisiting the recent European droughts from a long-term perspective *Scientific Reports* 8 9499 Online: https://doi.org/10.1038/s41598-018-27464-4
- Hov Ø, Cubasch U, Fischer E, Höppe P, Iversen T, Gunnar Kvamstø N, Kundzewicz W. Z, Rezacova D, Rios D, Duarte Santos F, Schädler B, Veisz O B and Zer U 2013 Extreme Weather Events in Europe: preparing for climate change adaptation (Oslo)
- Iizumi T and Ramankutty N 2015 How do weather and climate influence cropping area and intensity? Global Food Security 4 46–50 Online: http://dx.doi.org/10.1016/j.gfs.2014.11.003
- International Olive Council 2018 World Olive Oil Figures Online: https://www.internationaloliveoil.org/wpcontent/uploads/2020/04/HO-W901-29-11-2019-P.pdf
- IPCC 2012 Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation. A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change [Field, C.B., V. Barros, T.F. Stocker, D. Qin, D.J. Dokken, K.L. Ebi, M.D (Cambridge, UK, and New York)
- Jägermeyr J 2020 Agriculture's Historic Twin-Challenge Toward Sustainable Water Use and Food Supply for All Frontiers in Sustainable Food Systems 4 35 Online: https://www.frontiersin.org/article/10.3389/fsufs.2020.00035
- Jägermeyr J and Frieler K 2018 Spatial variations in cultivars pivotal to understand global fluctuations in maize and wheat yields Science Advances 1–11 Online: http://advances.sciencemag.org/content/4/11/eaat4517/tab-pdf
- 30. Jägermeyr J, Gerten D, Schaphoff S, Heinke J, Lucht W and Rockström J 2016 Integrated crop water management might sustainably halve the global food gap *Environmental Research Letters* 11
- 31. Jägermeyr J, Robock A, Elliott J, Müller C, Xia L, Khabarov N, Folberth C, Schmid E, Liu W, Zabel F, Rabin S S, Puma M J, Heslin A, Franke J, Foster I, Asseng S, Bardeen C G, Toon O B and Rosenzweig C 2020 A regional nuclear conflict would compromise global food security *Proceedings of the National Academy of Sciences* 117 7071 LP 7081 Online: http://www.pnas.org/content/117/13/7071.abstract
- 32. Knox J, Daccache A, Hess T and Haro D 2016 Meta-analysis of climate impacts and uncertainty on crop yields in Europe Environmental Research Letters 11 113004 Online: http://dx.doi.org/10.1088/1748-9326/11/11/113004
- Kornhuber K, Osprey S, Coumou D, Petri S, Petoukhov V, Rahmstorf S and Gray L 2019 Extreme weather events in early summer 2018 connected by a recurrent hemispheric wave-7 pattern *Environmental Research Letters* 14 54002 Online: http://dx.doi.org/10.1088/1748-9326/ab13bf
- Kottek M, Grieser J, Beck C, Rudolf B and Rubel F 2006 World Map of the Köppen-Geiger climate classification updated Meteorol. Z. 15 259–63 Online: http://koeppen-geiger.vu-wien.ac.at/present.htm
- 35. Kundzewicz Z W, Pińskwar I and Brakenridge G R 2017 Changes in river flood hazard in Europe: a review *Hydrology Research* 49 294–302 Online: https://doi.org/10.2166/nh.2017.016
- Leng G 2017 Recent changes in county-level corn yield variability in the United States from observations and crop models Science of The Total Environment 607–608 683–90
- 37. Leng G and Hall J 2019 Crop yield sensitivity of global major agricultural countries to droughts and the projected changes in the future *Science of the Total Environment* 654 811–821 Online: https://doi.org/10.1016/j.scitotenv.2018.10.434
- Leng G and Huang M 2017 Crop yield response to climate change varies with crop spatial distribution pattern Scientific Reports 7 1463 Online: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC5431105/

- Lesk C, Rowhani P and Ramankutty N 2016 Influence of extreme weather disasters on global crop production Nature 529 84–7 Online: http://dx.doi.org/10.1038/nature16467
- Mann M E, Rahmstorf S, Kornhuber K, Steinman B A, Miller S K, Petri S and Coumou D 2018 Projected changes in persistent extreme summer weather events: The role of quasi-resonant amplification *Science Advances* 4 eaat3272 Online: http://advances.sciencemag.org/content/4/10/eaat3272.abstract
- Mueller B, Hauser M, Iles C, Rimi R H, Zwiers F W and Wan H 2015 Lengthening of the growing season in wheat and maize producing regions Weather and Climate Extremes 9 47–56 Online: http://www.sciencedirect.com/science/article/pii/S2212094715000183
- Öner M and Deveci Kocakoç İ 2017 JMASM 49: A Compilation of Some Popular Goodness of Fit Tests for Normal Distribution: Their Algorithms and MATLAB Codes (MATLAB) *Journal of Modern Applied Statistical Methods* 16 547–75 Online: http://digitalcommons.wayne.edu/jmasm/vol16/iss2/30
- Pfleiderer P, Schleussner C-F, Kornhuber K and Coumou D 2019 Summer weather becomes more persistent in a 2 °C world Nature Climate Change 9 666–71 Online: https://doi.org/10.1038/s41558-019-0555-0
- 44. Puma M J, Bose S, Chon S Y and Cook B I 2015 Assessing the evolving fragility of the global food system *Environmental* Research Letters 10
- 45. Ray D K, Gerber J S, MacDonald G K and West P C 2015 Climate variation explains a third of global crop yield variability *Nature* Communications 6 5989 Online: http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=4354156&tool=pmcentrez&rendertype=abstract
- 46. Rosa L, Rulli M C, Davis K F, Chiarelli D D, Passera C and D'Odorico P 2018 Closing the yield gap while ensuring water sustainability *Environmental Research Letters* 13 104002 Online: http://dx.doi.org/10.1088/1748-9326/aadeef
- 47. Rosenzweig C, Elliott J, Deryng D, Ruane A C, Müller C and Arneth A 2014 Assessing agricultural risks of climate change in the 21st century in a global gridded crop model intercomparison *PNAS* 111 4–9
- 48. Rost S, Gerten D, Hoff H, Lucht W, Falkenmark M and Rockström J 2009 Global potential to increase crop production through water management in rainfed agriculture *Environmental Research Letters* 4 99
- Russo S, Sillmann J and Fischer E M 2015 Top ten European heatwaves since 1950 and their occurrence in the coming decades *Environmental Research Letters* 10
- 50. Stott P 2016 How climate change affects extreme weather events *Science* 352 1517 LP 1518 Online: http://science.sciencemag.org/content/352/6293/1517.abstract
- 51. Troy T J, Kipgen C and Pal I 2015 The impact of climate extremes and irrigation on US crop yields *Environmental* Research Letters 10
- 52. UNDRR/CRED 2020 The human cost of disasters: an overview of the last 20 years (2000-2019) Online: https://www.undrr.org/publication/human-cost-disasters-overview-last-20-years-2000-2019
- Vogel E, Donat M G, Alexander L V, Meinshausen M, Ray D K, Karoly D, Meinshausen N and Frieler K 2019 The effects of climate extremes on global agricultural yields *Environmental Research Letters* 14 054010 Online: http://dx.doi.org/10.1088/1748-9326/ab154b
- 54. Wine Institute 2017 Wine Institute World statistics. World Wine Consumption by Country Online: https://www.wineinstitute.org/resources/statistics
- Wong C K and Easton M C 1980 An Efficient Method for Weighted Sampling Without Replacement SIAM Journal of Computing 9 111–113

5 General discussion and conclusions

5.1 Key contributions

The research carried out in this dissertation brought the attention to the needs of better quantifying and assessing the impacts from extreme weather disasters (EWD) on food production, and for the implications they may represent to the EU food import dependencies. EWD can significantly exacerbate annual variability in crop yields, and, consequently, the fluctuations in food availability and in food prices (Jägermeyr and Frieler, 2018; Lesk et al., 2016; Nelson et al., 2014; Ray et al., 2015; Vogel et al., 2019). Therefore, by means of the interconnections of the world food system, these events have the potential to immediately, or indirectly, threaten local to global food security (Puma *et al.*, 2015; Rosenzweig *et al.*, 2001). The challenges for agriculture are associated to changes in the long-term average climate, and also to the occurrence of EWD, which are usually more impactful and generally more uncertain. However, there are major data gaps of the extent that EWD impact agriculture (FAO, 2015), as national and international disaster loss databases rarely report damage or losses in the sector. In particular, there is still little quantitative evidence of increasing trends in crop losses associated to disasters, and also on its implications to food trade (Lesk *et al.*, 2016; Puma *et al.*, 2015).

The European Union with 28 Member States (EU) is a major player in the global food market, and also a world leader in the fight against climate change (Bas-Defossez et al., 2018; Berkhout et al., 2018; Ciscar et al., 2018; Tai et al., 2014). Thus, the way the EU addresses the challenges of agriculture, sustainability and healthy diets has implications at the global level. My motivation with this dissertation is to slightly contribute for such discussion, while advancing the knowledge on the impacts of EWD in the EU food system. Specifically, this dissertation contributes to answering the following research questions: (1) What is the exposure of the EU food imports to extreme weather disasters?; (2) What are the impacts of extreme weather disasters in the EU food production?, and (3) What are the trends on EU crop losses during extreme weather disasters years?

We took advantage of records of EWD – droughts, heatwaves, floods, and cold waves – provided by EM-DAT (2018), which is the most comprehensive global database of natural and technological disasters occurred from 1900 to present (Lesk et al., 2016; Park et al., 2019). Records of EWD were combined with observational agricultural (FAO, 2017) and trade data (only in Chapter 3) (EUROSTAT, 2017), between 1961 and 2018 (2016 in Chapter 3). Through the use of time series statistical analysis based on superposed epoch analysis (SEA), we estimated the impact of EWD on the average production, yield, and harvested area of selected staple and cash crops. The SEA was implemented to estimate the impacts of EWD in agriculture across Non-EU suppliers, and to estimate the exposure of EU agri-food imports to EWD impacts based on the import share per supplier country (in Chapter 3). Also, the SEA was implemented at the EU level and per bioclimatic region to estimate the EWD responses in multiple crop categories (in Chapter 4). At the EU level, we also evaluated the trend across crop production anomalies over time, per disaster type.

Several innovative scientific contributions were achieved with this dissertation (Table 5.1):

- Statistically significant losses due to historical droughts and heatwaves (soybeans, tropical fruits, and cocoa), floods (soybeans and tropical fruits) and cold waves (banana production) were estimated in countries from which EU has a reasonable import share;
- Despite a diversified external market, the EWD crop losses in Non-EU suppliers represent a substantial and negative exposure to EU food imports. Production losses of soybeans, tropical fruits, and cocoa associated to droughts and heatwaves but also floods, lead to an overall decline, up to 16%, in the EU import-weighted share of each crop;
- 3. At the EU level, on average, droughts reduced European cereal yields by 9%, and heatwaves by 7%;
- 4. Major losses are found for cereals, but also vegetables and oil crops in the Eastern countries, while smaller losses are estimated in Southern but also Central European countries;
- The severity of aggregated heatwave and drought impacts on crop production roughly tripled over the last 50 years, from -2 (1964-1990) to -7% (1991-2015);
- Results suggest that droughts are significantly becoming more severe over time as, in every new year with a drought, the EU cereal production losses increase by 3%;
- 7. Results indicate an annual increase in event frequency on droughts (1%), heatwaves (6%), floods (29%), and cold waves (10%), in Europe;
- Even though using weather disaster records as a metric for extreme weather event impact analysis has limitations, it offers a unique and standardized approach suggesting that, at the EU level, climate change is already driving increasing crop losses in observational records.

In particular, the following outcomes are provided to each Research Question and are consolidated in Table 5.1, along with the main methodological limitations.

RQ#1: What is the exposure of the EU food imports to extreme weather disasters?

The published paper Brás *et al.* (2019) (in Chapter 3) highlights the EWD impacts on the EU largest import commodities – soybean, banana, tropical fruits, cocoa, and coffee – grown in Non-EU food suppliers, and presents the larger implications of such impacts through trade dependencies based on the import share per supplier (Fig. 5.1). It includes staple crops – which are relevant for caloric consumption in the EU – but also cash crops, most of them not grown in the region itself due to its natural conditions. More than 90% of the EU consumption of those crops is produced in 41 Non-EU countries. We found that, despite a diversified external market, the EWD impacts on crops grown in Non-EU suppliers represent a substantial and negative exposure to EU food imports. Specially

droughts and heatwaves lead to a decline in EU import-weighted shares for soybeans, tropical fruits, and cocoa, but also floods for soybeans and tropical fruits. For example, cocoa production is substantially affected by droughts and heatwaves (-5%), with import share-weighted impact of -7%. Such impact difference means that major losses took place in exporting countries from where EU has a higher import dependency. Since the EU completely relies on cocoa imports to satisfy its consumption, such weighted loss in cocoa production may have consequences to market speculations and may result in economic volatility.

Regarding soybeans – that in the EU is mostly used for animal feed (EU, 2018) – overall production losses from droughts and heatwaves, and floods of -11% across Non-EU exporters, represent a total EU import share-weighted impact of -9%. This means that major relative losses took place in exporting countries from where EU has a lower import dependency. Nevertheless, such impact may imply a potential decrease on the crop availability in the EU market. Since soybean is a common substitute of wheat and maize, any fluctuation on its production, and consequently on its prices, may influence the demand and supply chain of other commodities as well (Ercin *et al.*, 2016). We argue that production anomalies of these crops can potentially reduce caloric availability to some extent in the EU but are not expected to fundamentally impair EU food supply. Import-induced market volatility and resulting market speculations, however, can lead to price spikes in the EU and this can have significant adverse effects on food access (see section 5.2 for further discussion).

RQ#2: What are the impacts of extreme weather disasters in the EU food production?, and RQ#3: What are the trends on EU crop losses during extreme weather disasters years?

While the answer to RQ#2 quantifies the impact (e.g. degree of change) in crop responses due to EWD, the answer to RQ#3 evaluates the trend of such impacts over time.

A key outcome of Chapter 4 refers that European crop losses tripled over the last five decades (from -2.2 (1964-1990) to -7.3% (1991-2015)) due to aggregated droughts and heatwaves. At the European level, major averaged production losses, associated to such events, are found for cereals, but also for sugar, oil crops, and olives (Fig. 5.2a). In a regional assessment (Fig. 5.2b) we found that higher crop losses associated to droughts and heatwaves are estimated in Eastern Europe, while smaller losses are found in Mediterranean and Central European countries. The geographic difference in EWD impacts can be possibly explained by the share of cropping area under irrigation. It can largely mitigate adverse impacts from droughts and heatwaves by cooling surface temperatures and through maintaining increased soil moisture requirements (Jägermeyr and Frieler, 2018; Leng, 2017; Leng and Hall, 2019; Troy *et al.*, 2015; Vogel *et al.*, 2019). In general terms, the area under irrigation could be expanded in Europe as a measure to alleviate exposure to droughts and heatwaves, but substantial investments would be required (Elliott *et al.*, 2014). It would correspondingly impact water resources availability and energy consumption for crop irrigation (Daccache *et al.*, 2014; Fader *et al.*, 2016). This might have an impact in the cost of crop production and on food prices. Additionally, if this additional energy is to be provided by fossil fuels, irrigation demand will also correspond to a rise of greehouse gas emissions (GHG), which is not aligned with EU climate goals. Notably, the evaluation of the irrigation potential must be guided by water sustainability standards such as the *European Water Framework Directive* (European Comission, 2000; Iglesias and Garrote, 2015). Specific measures, such as traditional and sustainable water management practices like conservation tillage, organic mulching, and water harvesting for supplemental irrigation during dry spells, as well as, using drought and heat tolerant crop varieties, are shown to offer large and synergistic opportunities to buffer impacts of extreme weather in both rainfed and irrigated systems (Iglesias and Garrote, 2015; Islam et al., 2016; Jägermeyr, 2020; Jägermeyr et al., 2016; Rosa et al., 2018; Rost et al., 2009).

Most importantly, and especially for droughts, we found an increasing trend on the annual cereal production losses: on average, in every new year with a drought, EU cereal production losses increase by 3%, thus suggesting that these events are becoming more severe. Even though we used records of disaster events as a standardized metric for extreme weather event impact analysis (which has limitations listed in Table 5.1), our results suggest that climate change is already driving increased crop losses in observational records.

At the EU level, results indicate an annual increase in event frequency on droughts (1%), heatwaves (6%), floods (29%), and cold waves (10%), which is in line with the recently published report of the United Nations Office for Disaster Risk Reduction (UNDRR) (2020) showing a sharp increase in worldwide EWD over the last 20 years. The increase in the events frequencies may partially be explained by better recording and reporting, yet much of it is attributed to a significant rise in the number of climate-related disasters (UNDRR/CRED, 2020). In view of future projections in summer dryness in most parts of Europe, with longer and more intense heatwaves and droughts (IPCC, 2019), agricultural losses as estimated in this study may increase in the future.

With this dissertation, we found that specially droughts and heatwaves, have potential to negatively impact the EU food imports, namely for crops not grown in the EU (such as cocoa and tropical fruits), but also staple crops such as soybean and for which EU only produces about 10% of its consumption. At the EU level, and across its bioclimatic regions, significant averaged production losses, particularly from droughts and heatwaves, are estimated for staple crops such as cereals and vegetables. These are the food commodities with the highest production by weight (FAO, 2019a) accounting for nearly 30% of the total EU food exports. In the EU more than 60% of the cereals consumed are used for animal feed and nearly 30% are used for human consumption. Thus, a decline in soybean and cereals availability is particularly relevant because more than 60% of the calories in the human diet is highly dependent on these crops (i.e. wheat, maize, soybeans, and also rice), and their consumption is increasing with the growing demand for animal protein-based diets. More than 50% of EU citizens are overweight, 5% are at risk of undernutrition and around 8% live in food poverty (Bas-Defossez et al., 2018). Thus, to some extent, potential losses in the availability of staple crops can contribute to food insecurity, especially for the most vulnerable groups, potentially exacerbating social unrest. Potential actions to overcome such challenges are addressed in section 5.2.


Figure 5.1 Potential exposure of EU food imports to EWD in Non-EU suppliers.

For each food commodity and EWD type (i.e. droughts and heatwaves, floods and cold waves), it is indicated the EU import dependency (%) (2nd column), the spatial location of Non-EU suppliers (3rd column), and the percentage of change in crop production in Non-EU suppliers and in the EU imports (4th column). Results consider the EWD reported in EM-DAT (2018) from 1964 to 2013, and are shown only for crop production anomalies, across Non-EU suppliers, with significance levels above 0.10 (see Chapter 3 for details).



Figure 5.2 Drought and heatwaves impacts on crop production in the EU and in its bioclimatic regions.

Mediterranean climate (Csa, light green), Warm-summer humid continental climate (Dfb, purple)). Results are shown only for the production anomalies with significance region (b). Bioclimatic regions are according to Koeppen-Geiger (KG) (Kottek et al., 2006) classification (c): Temperate oceanic climate (Cfb, dark green), Hot-summer Averaged impacts on crop production (%) from droughts and heatwaves, reported in EM-DAT (2018) from 1964-2015, are indicated for the EU (a), and per bioclimatic levels above 0.10 (see Chapter 4 for details).

Context Question#: What are the current and future hotspots of climate change impacts on food production?

Even though we cannot picture the potential future EWD impacts on the four main produced crops – wheat, maize, soybeans, and rice –, we expect that their production is compromised under a future average climate, in the EU and worldwide (Fig. 5.3). The literature review carried out in Chapter 2 highlights that, by 2050, under the higher emission climate scenario, and even considering crop adaptation in few countries, the global breadbaskets – USA, China, Europe, Southern Asia, and Southern America – may see a decline in crop production, from -5% to -55%. This will come with harmful consequences on the worldwide food supply and on its prices through the interconnections of the global food market. Particularly, the European wheat and maize production may decline by 12 and 40%, respectively. The EU may then be forced to import these crops from other regions to satisfy its current levels of grain consumption. But world regions growing crops that EU highly imports, may also see their production decline.

For example, the EU has an import dependency of 60% of the soybean grown in southern America, where future production losses are estimated in 25% (Fig. 5.3). Southern and southeast Asia may have declines up to 55% on rice production, a region that currently satisfies 20% of the EU consumption. Nearly 29% of the soybean consumed in the EU is grown in the USA and Canada, and where average losses up to 10% are projected by 2050. Few positive changes on crop responses may occur if adaptation measures and the effect of CO₂ fertilisation are considered. In fact, other countries or regions may see their crop production increase (see section 2.2). Still, while few aspects of climate change may bring localized benefits for agriculture, the impacts from EWD may offset those gains.



Figure 5.3 Projected climate change impacts on crop production by 2050 under RCP8.5. Results are shown only for crops currently grown in the EU and on regions from where EU is mostly dependent (see Fig. 3.1 in Chapter 3). The projected climate change impacts are based in Figure 2.2 in Chapter 2.

General goal	To contribute to a better understa on its implications through trade of	anding of the effects of extreme we dependences, while taking advanta	ather disasters on food production and ge of observational records.
	Chapter #2	Chapter #3	Chapter #4
Title	Review on the climate change impacts on food availability and access	Exposure of the EU food imports to extreme weather disasters in exporting countries	Drought and heatwave crop losses tripled over the last five decades in Europe
Research question	What are the current and future hotspots of climate change impacts on food production? (context question)	RQ#1: What is the exposure of the EU food imports to extreme weather disasters?	RQ#2: What are the impacts of extreme weather disasters in the EU food production? RQ#3: What are the trends on EU crop losses during extreme weather disasters years?
Methods	Systematic literature review of scientific literature.	Time series statistical analysis based on superposed epoch analysis (SEA). EU import exposure based on the import share per supplier.	Time series statistical analysis based on SEA and normalised anomalies trends.
Main findings	 In a medium-term future, under the higher emission scenarios and even with crop adaptation, global breadbaskets may see a decline on crop production. Specifically, EU may see its wheat and maize production decline by 12 and 40% correspondingly. 	 Droughts and heatwaves caused production losses to soybeans (4%), tropical fruits (3%), and cocoa (5%) with import weighted impacts of 3, 8, and 7%, respectively. Floods caused weighted impacts of 7% (soybeans) and 8% (tropical fruits). Cold waves negatively impacted banana production but those events took place in countries which together represent only 3% of the EU import share of that crop. Coffee production shows gains during cold waves, but the inter-annual variability offsets these effects. 	 The severity of aggregated heatwave and drought impacts on EU crop production roughly tripled over the last 50 years Droughts are leading to cereal production losses by more than 3% per year. Frequencies of droughts, heatwaves, floods, and cold waves significantly increased over time in the EU. Using a weather disaster record for crop impact analyses offers a unique and standardized metric indicating that climate change is already driving increasing crop losses in observational records.
Methodological limitations	 This study would benefit from more studies assessing climate change impacts in access dimension, and from studies oriented to the nutrition value of food consumption. 	 The effect of EWD can be mu countries where only part of the Not all the weather events with classified as natural disasters in the effects of local extreme even Thus, the number of events with reported. No weights were attributed to the there is no such data available, in the same way. Moreover, si from many countries, it could no those events, i.e., losses in one the others. The EWD were not selected agricultural systems, and we did in each country (i.e., if tolerant no such data available. 	Ich stronger locally, especially in large e cultivated area is being affected. In impact on agriculture are reported or recorded in EM-DAT, as information on ents are tracked in local statistics only. ill be higher than the associated EWD the magnitude and duration of EWD as meaning that we treated all events listed nce we aggregated data for each crop esult in the attenuation of the impact of country could be offset by gains among d based on the crop growth stage, d not consider the type of crops varieties or not to a type of an EWD) as there is

Table 5.1 Revisiting the research design including the main findings to the research questions.

5.2 Contributions at the policy level

With the expectation of EWD becoming more frequent and intense in the future (IPCC, 2019), the results achieved by this dissertation raise questions and possibilities – on supply and consumption sides of the food system. A set of plausible future actions, under climate-proof food policies, may potentially contribute to better deal with the exposure to EWD, not only in the EU but worldwide. Even though this dissertation does not directly contribute to the United Nations Sustainable Development Goals (SDG) 2 and 12 – Zero Hunger, and Responsible Consumption and Production –, the bellow discussed contributions at the policy level have potential to tackle SDG targets.

Contribution#1: Diversification of EU external market while "exporting" resilience to Non-EU exporting countries

Food and trade policies may be re-designed in view of a higher investment in EU food imports. To secure imports of cocoa, tropical fruits, and soybeans, but also other food commodities whose production in the EU has been significantly affected from EWD, the EU could diversify even more its external market. Concurrently, the EU can assist on adaptation schemes in exporting countries, for example by establishing partnerships for research and innovation in crop tolerance to extreme weathers, and by supporting the definition and implementation of disaster risk reduction and management actions, while also supporting the implementation of fair and ethical food policies. This would also be helpful to promote the stability on the production of such crops and, consequently, the stability of incomes in exporting countries, contributing for local food security.

Contribution#2: EU as a key player in dietary shift

Another line of action refers to the relation between crop production and the dietary patterns. The world grows about 2.5 times more cereals, and overproduces fats and sugars, while production of fruits and vegetables is about 20% of what would be needed (according to USA dietary guidelines) (Benton, 2019). The food system thus supplies excessive calories by means of standardised diets, whose major ingredients (i.e. wheat, maize, rice, and soybean) are grown in monocultures and in geographically concentrated. This implies that agricultural systems are less resilient to temporal fluctuations in climate and that the world's capacity in coping with geographical risk is limited.

As an alternative, if we produce and consume more fruits and vegetables – while choosing less water intensive crops – as well as, move our diets to plant-based protein, worldwide it would be feasible to eat a nutritionally balanced diet, while reducing pressure on land, and also reduce GHG emissions (Benton, 2019; KC *et al.*, 2018). This would also potentially reduce monocultures and increase agricultural resilience to climate variability, including EWD. In the EU, vegetables occupy about 4.5% of the total cropping area, whereas cereals 65% (Fig.C1a). By reducing grain production for livestock feed – and even though by 2050 the EU population may decrease 1.3% (EUROSTAT, 2020) – such measures could contribute to alleviate the pressure on natural resources from a growing world population.

Another action to reduce the cropping area used for cereals production, highlighted by Santos (2010) is taxing, at the global level, cereals used in animal feed. If such tax is implemented at a sufficiently high rate, it would turn meat sufficiently expensive that the wealthier moderate their consumption, and simultaneously, would make cereals cheaper for food.

An additional and complement line of action in view an increase in vegetables production, while reducing the direct EWD impacts on crops, is a greater focus on urban agriculture for high-value, nutritious crops grown in the urban environment and peri urban fringe, in line with the Milan Urban Food Policy Pact (Edmondson *et al.*, 2020; 2015).

To support such actions, EU agricultural policies could be more driven by nutritional needs and not by economic growth considerations. Governments could also promote preventive healthcare, so people consume less fats, sugar and grains (by means of animal protein), which would thus reduce pressure on natural resources (Benton, 2019).

5.3 Final remarks and future research

This dissertation identifies the effects of droughts, heatwaves, floods, and cold waves in the EU food availability, in terms of its food import dependency and own crop production, while expanding the analysis to different crops and geographical regions. The results achieved are a flagship for policymakers, and food-related stakeholders, to potential develop adaptation and disaster risk management interventions relevant to agricultural and trade policies. In view of such potential contribution, a number of conceptual improvements shall be implemented to this research, and further research shall be explored:

- 1. From a conceptual point of view, the research carried out in this dissertation would benefit if it includes an econometric analysis, as well as, all crop categories imported by the EU, and additional categories of EWD. An assessment of crop prices variability, during EWD years, would support the identification of potential price spikes and also impacts-induced market volatility (i.e. in the EU and in Non-EU supplier countries). In addition, to provide information that supports an economic decision i.e. weather if, due to EWD impacts on crop production the EU shall invest on imports or, rather, on own production –, the assessment has to include all crops that are grown in the EU but also the ones imported, and preferably, expanded by bioclimatic region. Moreover, the analysis could be extended to other types of EWD such as storms, that are becoming more frequent, but perhaps not hail due to its very localized nature. Future research could take advantage of data on EWD that occur in a medium to local scale. It could also be improved if benefited from a detailed georeferenced information on the agro-climatic zones from crop growing regions and on the major agricultural systems (see additional methodological limitations in Table 5.1).
- 2. To better contribute to specific crop adaptation actions, it would be useful to identify and assess the main climate drivers of EWD, in particular droughts and heatwaves. The degree of change in temperature,

precipitation, or in the aridity index, during droughts and heatwaves, would allow, for example, to identify if such events are becoming hotter and drier. Such assessment would also allow to identify which European regions are becoming more arid. This work is currently under preparation and is foreseen to be submitted to a peer review journal in April 2021.

- 3. Droughts and heatwaves lead to an increase on crops' evapotranspiration and, thus, to higher irrigation requirements. It is not, however, yet quantified to what degree irrigation can compensate or alleviate drought and heatwave impacts in the EU agriculture. To answer this question, one could take advantage from EWD records combined with gridded crop data simulated by crop-based models. This work is currently being developed by using crop yield and potential irrigation water withdrawals datasets, provided by the Inter-Sectoral Impact Model Intercomparison Project (ISIMIP).
- 4. Any increase in irrigation demand will affect energy consumption since energy is required to withdrawal water and to distribute it for crop irrigation. Thus, one may ask how much energy is needed for crop irrigation during droughts and heatwaves. The Food and Agriculture Organization from the United Nations makes available annual records of the energy used in crop irrigation, at the country level (FAO, 2019b). However, those records are only reported per country, on every five or 10 years, and are repeated until the next reporting. Thus, such data lacks variability, in addition to missing records from a significant number of countries. The energy demand for crop irrigation has therefore to be modelled. One possible approach is to follow an already published methodology developed for the Mediterranean region (Daccache *et al.*, 2014), while taking advantage of more recent datasets (i.e. historical crop yield and irrigation water withdrawals, as well as, the share of irrigation system and its efficiency per system and crop type, the global shares of irrigation water sources, and the global data set of water table depth).
- 5. Finally, two very important open questions: (1) How EWD will potentially impact European crop yields in the future?, and (2) How much water and energy may be invested to attenuate such impact? Answering to these questions implies an understanding of the explanatory power of predictors i.e. climate variables (e.g. temperature and precipitation) and energy demand to crop yield anomalies. Evaluating the link of EWD with statistical meteorological definitions will also enable a forecasting of future impacts under climate change scenarios (Lesk et al., 2016). One could also take advantage of a selected number of remote sensing vegetation indexes used for crop yield predictions (such as (Burke and Lobell, 2017; Panda *et al.*, 2020)). Quantifying crop anomalies, and understanding their climate drivers is a prerequisite to assess vulnerabilities and design adaptation measures and corresponding costs for example through an investment in irrigation to increase the resilience of food systems.

5.4 Outputs

This section includes a list of the main outputs of this work.

Peer-reviewed publications

The first author of the papers was the leading responsible for the development of the investigation, that was supported by the co-authors, mainly regarding the design of the research, discussion of the results and revision of the manuscripts.

- 1. Brás, T.A., Seixas, J. Review on the climate change impacts on food availability and access. *Under Review* in *Global and Planetary Change*
- 2. Brás, T.A., Jägermeyr, J., Seixas, J., 2019. Exposure of the EU-28 food imports to extreme weather disasters in exporting countries. *Food Security*. https://doi.org/10.1007/s12571-019-00975-2
- 3. Brás, T.A., Seixas, J., Carvalhais, N., Jägermeyr, J., Drought and heatwave crop losses tripled over the last five decades in Europe. *Under Review* in *Environmental Research Letters*

Presentations in scientific conferences, workshops, and projects

- Oral Communication, Brás, T.A.; Seixas J.; Assessing the impact of climate extremes and energy use in crop production – EU28 agri-food suppliers, 18-20 Sep.. Fifth Annual International Conference on Sustainable Development., New York, USA, Columbia University, 2017
- Oral Communication, Brás, T.A.; Jägermeyr, J.; Carvalhais N.; Seixas, J.; How extreme weather disasters affect Food Security - Seeking an Integrated Approach with EO, AIR Centre Workshop on Discovering Exploratory EO Use-Cases in the Atlantic, December at TERINOV, Terceira Island, Azores, Portugal, 2019
- Project participation: ERA4CS Joint Call on Researching and Advancing Climate Services Development
 project entitled "CLIM2POWER Translating climate data into power plants operational guidance". My
 participation included the estimation of future irrigation water demand in Europe and in the larger
 Portuguese and Spanish watersheds, by considering the representative concentration pathways (RCP)
 8.5, and by taking advantage of the ISIMIP global data set on potential irrigation water withdrawals.
- Project participation: EU Climate KIC Pioneers programme, in Bulgaria during October-November 2018. Under this project, I selected Bulgaria as a case study to access historical irrigation water and energy

requirements. This project counted with the collaboration of the Faculty of Mathematics and Informatics, and the Faculty of Hydraulic Engineering at UACEG, in Sofia.

- European Space Agency Phi-Week Bootcamp 2019: Solving a Big Energy Challenge Using EO Data, Italy, December 2019, in which my team won the first prize.
- Short-term course: Climate KIC course on Earth observation Big data for climate change, Poland, Warsaw, May 2019
- MATLAB course at Instituto Superior Técnico, Lisbon, 2016 and 2017
- ArcGIS online tutorials, 2018

5.5 References

- 1. Bas-Defossez, F., Allen, B., Weigelt, J., Marechal, A., Meredith, S., Lorant, A., 2018. Feeding Europe: Agriculture and sustainable food systems, Policy Paper produced for the IEEP Think2030 conference. Brussels.
- 2. Benton, T.G., 2019. Using scenario analyses to address the future of food. EFSA Journal 17, e170703. https://doi.org/10.2903/j.efsa.2019.e170703
- Berkhout, P., Achterbosch, T., Berkum, S. van, Dagevos, H., Dengerink, J., Duijn, A.P. van, Terluin, I., 2018. Global implications of the European Food System: A food systems approach. Wageningen Economic Research, Wageningen. https://doi.org/10.18174/448884
- 4. Brás, T.A., Jägermeyr, J., Seixas, J., 2019. Exposure of the EU-28 food imports to extreme weather disasters in exporting countries. Food Security. https://doi.org/10.1007/s12571-019-00975-2
- Burke, M., Lobell, D.B., 2017. Satellite-based assessment of yield variation and its determinants in smallholder African systems. Proceedings of the National Academy of Sciences 114, 2189 LP – 2194. https://doi.org/10.1073/pnas.1616919114
- Ciscar, J.C., Ibarreta, D., Soria, A., Dosio, A., A.Toreti, Ceglar, A., Fumagalli, D., Dentener, F., Lecerf, R., Zucchini, A., Panarello, L., Niemeyer, S., Pérez-Domínguez, I., Fellmann, T., Kitous, A., J.Després, Christodoulou, A., Demirel, H., Alfieri, L., Dottori, F., Vousdoukas, M.I., Mentaschi, L., Voukouvalas, E., C.Cammalleri, Barbosa, P., Micale, F., Vogt, J.V., Barredo, J.I., Caudullo, G., Mauri, A., Rigo, D. de, Libertà, G., Durrant, T.H., Vivancos, T.A., San-Miguel-Ayanz, J., Gosling, S.N., Zaherpour, J., Roo, A. De, Bisselink, B., J.Bernhard, L., B., Rozsai, M., Szewczyk, W., Mongelli, I., Feyen, L., 2018. Climate impacts in Europe: Final report of the JRC PESETA III project, EUR 29427 EN. Publications Office of the European Union, Luxembourg. https://doi.org/10.2760/93257
- 7. Daccache, A., Ciurana, J.S., Rodriguez Diaz, J.A., Knox, J.W., 2014. Water and energy footprint of irrigated agriculture in the Mediterranean region. Environmental Research Letters 9, 12. https://doi.org/10.1088/1748-9326/9/12/124014
- Edmondson, J.L., Cunningham, H., Densley Tingley, D.O., Dobson, M.C., Grafius, D.R., Leake, J.R., McHugh, N., Nickles, J., Phoenix, G.K., Ryan, A.J., Stovin, V., Taylor Buck, N., Warren, P.H., Cameron, D.D., 2020. The hidden potential of urban horticulture. Nature Food 1, 155–159. https://doi.org/10.1038/s43016-020-0045-6
- 9. Elliott, J., Deryng, D., Müller, C., Frieler, K., Konzmann, M., Gerten, D., Glotter, M., Flörke, M., Wada, Y., Best, N., Eisner, S., Fekete, B.M., Folberth, C., Foster, I., Gosling, S.N., Haddeland, I., Khabarov, N., Ludwig, F., Masaki, Y., Olin, S.,

Rosenzweig, C., Ruane, A.C., Satoh, Y., Schmid, E., Stacke, T., Tang, Q., Wisser, D., 2014. Constraints and potentials of future irrigation water availability on agricultural production under climate change. Proceedings of the National Academy of Sciences 111, 3239 LP – 3244. https://doi.org/10.1073/pnas.1222474110

- 10. EM-DAT, 2018. The Emergency Events Database Universit'e catholique de Louvain (UCL) -CRED, D. Guha-Sapir. Brussels, Belgium [WWW Document]. URL https://www.emdat.be/index.php (accessed 1.5.19).
- 11. Ercin, A.E., Chico, D., Chapagain, A.K., 2016. Dependencies of Europe's economy on other parts of the world in terms of water resources, Horizon2020 IMPREX project, Technical Report D12.1.
- 12. EU, 2018. Monitoring Agri-trade Policy. MAP 2018–1. Agri-food trade in 2017: another record year for EU agri-food trade.
- 13. European Comission, 2000. Directive 2000/60/EC. Official Journal of the European Communities L327, 1–82.
- EUROSTAT, 2020. Population projections in the EU [WWW Document]. eurostat Statistics Explained. URL https://ec.europa.eu/eurostat/statistics-explained/index.php?title=People_in_the_EU_-_population_projections&oldid=497115#Population_projections (accessed 11.26.20).
- 15. EUROSTAT, 2017. EU trade in food.
- Fader, M., Shi, S., Von Bloh, W., Bondeau, A., Cramer, W., 2016. Mediterranean irrigation under climate change: More efficient irrigation needed to compensate for increases in irrigation water requirements. Hydrology and Earth System Sciences 20, 953–973. https://doi.org/10.5194/hess-20-953-2016
- 17. FAO, 2019a. The State of the World's Biodiversity for Food and Agriculture. FAO Commission on Genetic Resources for Food and Agriculture Assessments, Rome, Italy.
- 18. FAO, 2019b. FAOSTAT [WWW Document]. URL http://www.fao.org/faostat/en/ (accessed 7.30.19).
- FAO, 2017. FAOstat, Food and Agricultural Organization [WWW Document]. URL http://www.fao.org/faostat/en/ (accessed 1.5.19).
- FAO, 2015. The Impact of Natural Hazards and Disasters on Agriculture and Food and Nutrition Security A Call for Action to Build Resilient Livelihoods. Rome, Italy.
- 21. Iglesias, A., Garrote, L., 2015. Adaptation strategies for agricultural water management under climate change in Europe. Agricultural Water ManagementManagement 155, 113–124. https://doi.org/10.1016/j.agwat.2015.03.014
- IPCC, 2019. Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems [P.R. Shukla, J. Skea, E. Calvo Buendia, V. Masson-Delmot.
- Islam, S., Cenacchi, N., Sulser, T.B., Gbegbelegbe, S., Hareau, G., Kleinwechter, U., Mason-D'Croz, D., Nedumaran, S., Robertson, R., Robinson, S., Wiebe, K., 2016. Structural approaches to modeling the impact of climate change and adaptation technologies on crop yields and food security. Global Food Security 10, 63–70. https://doi.org/10.1016/j.gfs.2016.08.003
- 24. Jägermeyr, J., 2020. Agriculture's Historic Twin-Challenge Toward Sustainable Water Use and Food Supply for All . Frontiers in Sustainable Food Systems .
- 25. Jägermeyr, J., Frieler, K., 2018. Spatial variations in cultivars pivotal to understand global fluctuations in maize and wheat yields. Science Advances 1–11. https://doi.org/10.1126/sciadv.aat4517
- Jägermeyr, J., Gerten, D., Schaphoff, S., Heinke, J., Lucht, W., Rockström, J., 2016. Integrated crop water management might sustainably halve the global food gap. Environmental Research Letters 11. https://doi.org/10.1088/1748-9326/11/2/025002
- 27. KC, K.B., Dias, G.M., Veeramani, A., Swanton, C.J., Fraser, D., Steinke, D., Lee, E., Wittman, H., Farber, J.M., Dunfield, K., McCann, K., Anand, M., Campbell, M., Rooney, N., Raine, N.E., Acker, R. Van, Hanner, R., Pascoal, S., Sharif, S.,

Benton, T.G., Fraser, E.D.G., 2018. When too much isn't enough: Does current food production meet global nutritional needs? PLOS ONE 13, e0205683.

- Kottek, M., Grieser, J., Beck, C., Rudolf, B., Rubel, F., 2006. World Map of the Köppen-Geiger climate classification updated. Meteorol. Z. 15, 259–263. https://doi.org/DOI: 10.1127/0941-2948/2006/0130
- Leng, G., 2017. Recent changes in county-level corn yield variability in the United States from observations and crop models. Science of The Total Environment 607–608, 683–690. https://doi.org/https://doi.org/10.1016/j.scitotenv.2017.07.017
- Leng, G., Hall, J., 2019. Crop yield sensitivity of global major agricultural countries to droughts and the projected changes in the future. Science of the Total Environment 654, 811–821. https://doi.org/https://doi.org/10.1016/j.scitotenv.2018.10.434
- Lesk, C., Rowhani, P., Ramankutty, N., 2016. Influence of extreme weather disasters on global crop production. Nature 529, 84–87. https://doi.org/10.1038/nature16467
- 32. Milan Urban Food Policy Pact, 2015. Milan Urban Food Policy Pact [WWW Document]. URL https://www.milanurbanfoodpolicypact.org/text/ (accessed 11.26.20).
- Nelson, G.C., Valin, H., Sands, R.D., Havlík, P., Ahammad, H., Deryng, D., Elliott, J., 2014. Climate change effects on agriculture : Economic responses to biophysical shocks. PNAS 111. https://doi.org/10.1073/pnas.1222465110
- Panda, S.S., Ames, D.P., Panigrahi, S., 2020. Application of Vegetation Indices for Agricultural Crop Yield Prediction Using Neural Network Techniques. Remote Sensing 2, 2010. https://doi.org/10.3390/rs2030673
- Park, C.S., Vogel, E., Larson, L.M., Myers, S.S., Daniel, M., Biggs, B.-A., 2019. The global effect of extreme weather events on nutrient supply: a superposed epoch analysis. The Lancet Planetary Health 3, e429–e438. https://doi.org/10.1016/S2542-5196(19)30193-7
- Puma, M.J., Bose, S., Chon, S.Y., Cook, B.I., 2015. Assessing the evolving fragility of the global food system. Environmental Research Letters 10. https://doi.org/10.1088/1748-9326/10/2/024007
- Ray, D.K., Gerber, J.S., MacDonald, G.K., West, P.C., 2015. Climate variation explains a third of global crop yield variability. Nature Communications 6, 5989. https://doi.org/10.1038/ncomms6989
- Rosa, L., Rulli, M.C., Davis, K.F., Chiarelli, D.D., Passera, C., D'Odorico, P., 2018. Closing the yield gap while ensuring water sustainability. Environmental Research Letters 13, 104002. https://doi.org/10.1088/1748-9326/aadeef
- Rosenzweig, C., Iglesius, A., Yang, X.B., Epstein, P.R., Chivian, E., 2001. Climate change and extreme weather events - Implications for food production, plant diseases, and pests. GLOBAL CHANGE & HUMAN HEALTH, NASA Publications 2, 90–104. https://doi.org/10.1023/A:1015086831467
- Rost, S., Gerten, D., Hoff, H., Lucht, W., Falkenmark, M., Rockström, J., 2009. Global potential to increase crop production through water management in rainfed agriculture. Environmental Research Letters 4, 99. https://doi.org/10.1088/1748-9326/4/4/044002
- Santos, J.L., 2010. The Environmental Crisis and the Future of Agriculture, in: Environment at the Crossroads Aiming for a Sustainable Future. Carcanet Press Ltd., pp. 61–79.
- Tai, A.P.K., Martin, M.V., Heald, C.L., 2014. Threat to future global food security from climate change and ozone air pollution. Nature Climate Change 4, 817–821. https://doi.org/10.1038/nclimate2317
- Troy, T.J., Kipgen, C., Pal, I., 2015. The impact of climate extremes and irrigation on US crop yields. Environmental Research Letters 10. https://doi.org/10.1088/1748-9326/10/5/054013

- 44. UNDRR/CRED, 2020. The human cost of disasters: an overview of the last 20 years (2000-2019).
- 45. Vogel, E., Donat, M.G., Alexander, L. V, Meinshausen, M., Ray, D.K., Karoly, D., Meinshausen, N., Frieler, K., 2019. The effects of climate extremes on global agricultural yields. Environmental Research Letters 14, 054010. https://doi.org/10.1088/1748-9326/ab154b

6 Appendix

6.1 Appendix A

Review on the climate change impacts on food availability and access

Table A1 Observed impacts of climate drivers on food availability indicators by crops and by region

CSA – Caribbean & South America, EU – European Union, WSE - Western & Southern Europe, ENE – Eastern & Northern Europe. "Climate driver specification" refers to changes on P – precipitation, T – temperature, CO₂ – carbon dioxide, O₃ – ozone and Climate variability. "Impact specification" describes the impact of the climate driver on includes: Naf – Northern Africa, SSA – Sub-Saharan Africa, WSSE – Western & Southern & Southeast Asia, CEA – Central & Eastern Asia, NCA – Northern & Central America, crop yield or production (green, red, or orange text indicates positive, negative or no impact, respectively). IT – Italy, PT – Portugal, ES – Spain, FR – France, NL – Netherlands, Notes: Data is organised as follows: "Crop, Region: [location (*if appropriate*), (time scale); Climate driver specification: impact specification] Literature reference". "Region"

IE – Germany, GR – Greece, BE – B	ielgium, LU – Luxembourg, IE - It	reland.		
AFRICA	EUROPE	ASIA	AMERICA	GLOBAL
Wheat; Naf: (1974-2008); T & P:	Wheat; EU: [O3: yield -15% to -	Wheat; CEA: [China, (1961-2010), winter	Wheat; NCA: [O3: yield -15% to -	Wheat; Glob: [(1979–2008);
yield 12.0% (Ray et al., 2019)	30%] (Broberg, Feng, Xin, &	wheat; Average T increase: production -	30%] (Broberg et al., 2015)	Climate variability: explained
Wheat; SSA: (1974-2008); T & P:	Pleijel, 2015)	5 to -8%; T range decrease: production -	Wheat; NCA: [México; T variability	production fluctuations of 9
yield -2.3% (Ray et al., 2019)	Wheat; EU: [IT, LU, GR, BE, FR,	2 to -9%; P change: contributes to +1 to	(season-to-season T and radiation	MT/y] (Ray, Gerber,
Maize; Naf: (1974-2008); T & P:	ES, NL, PT, DE, UK, IE, (1989-	+6% in production at national scale]	variability): influence on yield inter-	MacDonald, & West, 2015)
yield -4.3% (Ray et al., 2019)	2009); T & P: yield (median) ~-	(Chen, Zhou, & Zhou, 2014)	annual variability] (Rosenzweig et	Wheat; Global: [(1981–2010); T
Maize; SSA: (1974-2008); T & P:	15%, -7%, -6%, -4%, -3%, -2.5%,	Wheat; CEA: [China, (1980–2008); T: yield	al., 2014)	increase (warming): yield -6%
yield -5.8% (Ray et al., 2019)	-2.5%, -2%, 0%, +4%, +9%]	+1.3%; P: yield -0.34%] (Wei, Cherry,	Wheat; NCA: [México ; CO ₂ increase	per degree of warming (with a
Rice; Naf: (1974-2008); T & P: yield	(Moore & Lobell, 2015)	Glomrød, & Zhang, 2014)	[550 ppm]: yield +19.6%	50% probability of between
-1.3% (Ray et al., 2019)	Wheat; WSE: (1974-2008); T &	Wheat; CEA: [China ; O3: yield -15% to -	(±2.8%STD)] (Rosenzweig et al.,	-4.2% and -8.2% loss)] (Asseng
Rice; SSA: (1974-2008); T & P: yield	P: yield -8.7% (Ray et al., 2019)	30%] (Broberg et al., 2015)	2014)	et al., 2015)
- 3.1% (Ray et al., 2019)	Wheat; ENE: (1974-2008); T &	Wheat; CEA: (1974-2008); T & P: yield	Wheat; NCA: (1974-2008); T & P:	Wheat; Glob: (1974-2008); T &
Barley; Naf: (1974-2008); T & P:	P: yield -2.1% (Ray et al., 2019)	4.5 % (Ray et al., 2019)	yield -1.3% (Ray et al., 2019)	P: yield -0.9% (Ray et al., 2019)
yield -6.8% (Ray et al., 2019)	Maize; EU: [IT, PT, ES, FR, NL,	Wheat; WSEA: (1974-2008); T & P: yield -	Wheat; CSA: (1974-2008); T & P:	Maize; Glob: [(1979–2009);
Barley; SSA: (1974-2008); T & P:	DE, GR, BE (1989-2009); T & P:	0.9% (Ray et al., 2019)	yield -1.6% (Ray et al., 2019)	Climate variability: explained
yield -0.6% (Ray et al., 2019)	yield (median) ~-8%, -6%, -	Maize; CEA: [China, (1980–2008); P: yield	Maize; NCA: (1974-2008); T & P:	production fluctuations of 22
Sorghum; Naf: (1974-2008); T & P:	2.5% , +1%, +4%, +5%, +9%,	- 0.14%] (Wei et al., 2014)	yield 0.5 % (Ray et al., 2019)	MT/y] (Ray et al., 2015)
yield 17.9% (Ray et al., 2019)	+ 12 %] (Moore & Lobell, 2015)			

66

Climate variability: explained production fluctuations of 3 Rice; Glob: (1974-2008); T & P: Barley; Glob: (1974-2008); T & Sorghum; Glob: (1974-2008); T Climate variability: explained production fluctuations of 2 MT/y] (Ray et Soybean; Glob: (1974-2008); T Oil palm; Glob: (1974-2008); T & P: yield -0.5% (Ray et al., Sugarcane; Glob: (1974-2008); & P: yield -13.4% (Ray et al., Rapeseed; Glob: (1974-2008); T Glob: (1974-2008); T P: yield -7.9% (Ray et al., 2019) **& P: yield 2.1**% (Ray et al., Soybean; Glob: [(1979–2009); **& P: yield 3.5%** (Ray et al., **& P: yield 0.5%** (Ray et al., T & P: yield 1.0% (Ray et al., Glob: [(1979–2009); vield -0.3% (Ray et al., 2019) MT/y] (Ray et al., 2015) GLOBAL al., 2015) soybean 2019) 2019) 2019) 2019) 2019) 2019) Rice; Vlaize; CSA: (1974-2008); T & P: Barley; CSA: (1974-2008); T & P: Sorghum; CSA: (1974-2008); T & P: <u>ن</u>ە Rapeseed; CSA: (1974-2008); T & P: Rice; NCA: (1974-2008); T & P: yield Rice; CSA: (1974-2008); T & P: yield <u>نہ</u> . تە . تە Rapeseed; NCA: (1974-2008); T & P: . تە Oil palm; CSA: (1974-2008); T & P: Sugarcane; CSA: (1974-2008); T & P: Cassava; NCA: (1974-2008); T & P: Sugarcane; NCA: (1974-2008); T & ے (2013 and 2014); T range decrease: averaged production -16% in 2013 Sorghum; NCA: (1974-2008); T & Soybean; NCA: (1974-2008); T & Soybean; CSA: (1974-2008); T & Oil palm; NCA: (1974-2008); T & Cassava; CSA: (1974-2008); T & ø compared with 2011–12) and -Coffee; NCA: [Central America, Barley; NCA: (1974-2008); T P: yield 1.7% (Ray et al., 2019) yield -2.5% (Ray et al., 2019) yield -7.2% (Ray et al., 2019) **yield -0.4%** (Ray et al., 2019) **yield -0.6%** (Ray et al., 2019) **yield -<mark>2.9%</mark> (**Ray et al., 2019) yield 5.4% (Ray et al., 2019) yield 4.0% (Ray et al., 2019) yield 4.3% (Ray et al., 2019) yield 0.0% (Ray et al., 2019) yield 3.3% (Ray et al., 2019) yield 6.8% (Ray et al., 2019) yield 2.5% (Ray et al., 2019) yield 0.5% (Ray et al., 2019) **/ield 2.7%** (Ray et al., 2019) -0.7% (Ray et al., 2019) -0.1% (Ray et al., 2019) AMERICA Maize; WSEA: (1974-2008); T & P: yield Maize; CEA: [China, (1980–2008); T: yield Maize; CEA: (1974-2008); T & P: yield 2014); T increase: yield -4.8% (J. Wang et western China), (1980s-2010); T increase (1961-2010); T increase (warming trend): Vaize; CEA: [China (Loess Plateau, northdecreased] (Zhao, Guo, Mu, & Xu, 2015) 0.27 day/y) and yield decline (32.0- 49.3 scale (however reduced yield in the west and central regions)] (Chen et al., 2014) 1990s); Minimum daily T increase: yield +3.6% /1°C rise during growing season] Rice; CEA: [China, (1980–2008); T: yield production -3 to -4% at national scale / Maize; CEA: [China, (1961-2010); P: no /laize; CEA: [China (Liaoning province), shortened rice growth duration (0.15 -Average T increase/ T range decrease: production from 0 to +6% at national (1961-2010), spring maize; T increase averaged -0.067 t ha-1 per year (p < Rice; CEA: [China, Yellow River Basin, Rice; CEA: [north-east China, (1980s-Rice; CEA: [southeast China, (2010significant effect in production at a national scale] (Chen et al., 2014) Maize; CEA: [China, (1961-2010); warming trend): yield potential climate warming): production (g/ha)] (W. Wang et al., 2014) (Zhou, Li, Dong, & Wu, 2013) +**0.4**%] (Wei et al., 2014) -12%] (Wei et al., 2014) ASIA 0.01)] (Bu et al., 2015) 5.1% (Ray et al., 2019) 1% (Ray et al., 2019) al., 2016) Sorghum; WSE: (1974-2008); T Soybean; WSE: (1974-2008); T Sugarcane; WSE: (1974-2008); T Rice; WSE: (1974-2008); T & P: Sorghum; ENE: (1974-2008); T & Rapeseed; ENE: (1974-2008); T & P: yield 2.7% (Ray et al., 2019) Maize; WSE: (1974-2008); T & P: Maize; ENE: (1974-2008); T & P: Rice; ENE: (1974-2008); T & P: FR, NL, PT, UK, IE, (1989-2009); -5%, -5%, -4.5%, -2.5%, -2.5%, -Barley; ENE: (1974-2008); T & P: & P: yield -18.2% (Ray et al., & P: yield -21.2% (Ray et al., Soybean; ENE: (1974-2008); T & Rapeseed; WSE: (1974-2008); T & P: yield -11.4% (Ray et al., & P: yield 3.1% (Ray et al., 2019) +4%] Barley; EU: [GR, IT, LU, BE, DE, T & P: yield (median) ~-8%, -6%, 2%, -2% ,-1%] (Moore & Lobell, Sugar beet; EU: [IT, GR, ES,NL, FR, BE, DE, UK (1989-2009); **T & P**: P: yield -9.5% (Ray et al., 2019) P: yield -3.8% (Ray et al., 2019) /ield -24.5% (Ray et al., 2019) yield -3.2% (Ray et al., 2019) yield -0.4% (Ray et al., 2019) **yield -9.1%** (Ray et al., 2019) yield -6.3% (Ray et al., 2019) +1%,+1.5%, +2.5%,+3%, -10% (Moore & Lobell, 2015) EUROPE vield ~-12.5%, 2015) 2019) 2019) 2019) Cassava; Naf: (1974-2008); T & P: Sorghum; SSA: (1974-2008); T & P: Soybean; Naf: (1974-2008); T & P: Soybean; SSA: (1974-2008); T & P: Rapeseed; SSA: (1974-2008); T & P: Oil palm; SSA: (1974-2008); T & P: Cassava; SSA: (1974-2008); T & P: Sugarcane; Naf: (1974-2008); T & P: Sugarcane; SSA: (1974-2008); T & P: yield 10.9% (Ray et al., 2019) yield 24.9% (Ray et al., 2019) yield 18.0% (Ray et al., 2019) yield -1.6% (Ray et al., 2019) yield -5.1% (Ray et al., 2019) yield -3.9% (Ray et al., 2019) yield 0.0% (Ray et al., 2019) yield 1.7% (Ray et al., 2019) **yield 0.7%** (Ray et al., 2019) AFRICA

GLOBAL	
AMERICA	10% in 2014 (compared with 2012– 13)] (Avelino et al., 2015)
ASIA	Rice; CEA: [China, (1980–2008); P: yield "0%] (Wei et al., 2014) Rice; CEA: (1974-2008); T & P: yield 0.9% (Ray et al., 2019) Barley; CEA: (1974-2008); T & P: yield - 0.8% (Ray et al., 2019) Barley; WSEA: (1974-2008); T & P: yield - 0.9% (Ray et al., 2019) Barley; WSEA: (1974-2008); T & P: yield - 0.9% (Ray et al., 2019) Sorghum; CEA: (1974-2008); T & P: yield 4.9% (Ray et al., 2019) Sorghum; WSEA: (1974-2008); T & P: yield 4.9% (Ray et al., 2019) Sorghum; WSEA: (1974-2008); T & P: yield 5.9% (Ray et al., 2019) Soybean; CEA: (1974-2008); T & P: yield 3.2% (Ray et al., 2019) Rapeseed; CEA: (1974-2008); T & P: yield 5.9% (Ray et al., 2019) Rapeseed; WSEA: (1974-2008); T & P: yield 5.9% (Ray et al., 2019) Oil palm; WSEA: (1974-2008); T & P: yield 1.5.9% (Ray et al., 2019) Oil palm; WSEA: (1974-2008); T & P: yield 5.9% (Ray et al., 2019) Oil palm; WSEA: (1974-2008); T & P: yield 1.5.9% (Ray et al., 2019) Oil palm; WSEA: (1974-2008); T & P: yield 5.9% (Ray et al., 2019) Oil palm; WSEA: (1974-2008); T & P: yield 1.5.9% (Ray et al., 2019) Sugarcane; WSEA: (1974-2008); T & P: yield 1.5.9% (Ray et al., 2019) Sugarcane; WSEA: (1974-2008); T & P: yield 1.5.9% (Ray et al., 2019) Sugarcane; WSEA: (1974-2008); T & P: yield 1.5.9% (Ray et al., 2019) Sugarcane; WSEA: (1974-2008); T & P: yield 1.5.9% (Ray et al., 2019) Sugarcane; WSEA: (1974-2008); T & P: yield 1.5.9% (Ray et al., 2019) Sugarcane; WSEA: (1974-2008); T & P: yield 1.5.9% (Ray et al., 2019) Sugarcane; WSEA: (1974-2008); T & P: yield 1.5.9% (Ray et al., 2019) Sugarcane; WSEA: (1974-2008); T & P: yield 1.5.9% (Ray et al., 2019) Sugarcane; WSEA: (1974-2008); T & P: yield 1.5.6% (Ray et al., 2019) Cassava; WSEA: (1974-2008); T & P: yield 1.2% (Ray et al., 2019) Cassava; WSEA: (1974-2008); T & P: yield 1.2% (Ray et al., 2019) Cassava; WSEA: (1974-2008); T & P: yield 1.2% (Ray et al., 2019) Cassava; WSEA: (1974-2008); T & P: yield 1.2% (Ray et al., 2019) Cassava; WSEA: (1974-2008); T & P: yield 1.2% (Ray et al., 2019) Cassava; WSEA: (1974-2008); T & P: yield 1.2% (Ray e
EUROPE	
AFRICA	

Table A2 Future projected impacts of climate drivers on food availability indicators by crops and by region

avg – average, CC – climate change, T –temperature; HCI – high climate index (high temperature increase, high sensitivity of crops to global warming, and a CO₂ fertilisation change; time scale; scenario; location; other (when adequate)] Literature reference". Food indicators (FI) are Y – Yield, P – Production, S - Suitable area and S_CD - Suitable Central & Eastern Asia, NCA – Northern & Central America, CSA – Caribbean & South America, EU – European Union, WSE - Western & Southern Europe, ENE – Eastern & Notes: Variations always refer to reference scenarios, which may differ from a study to another. Date is organised as follows: "Crop; Region: [food indicator; projection area due crop disease. A projected change on FI is represented by (+) when the change is positive, (-) when it decreases and "cte" when there is no change. Three-time scales IPCC Special Report Emissions Scenarios (SRES): A1, A1FI, A1T, A1B, A2, B1, B2; IPCC Representative Concentration Pathways (RCP): RCP2.6, RCP4.5, RCP6, RCP8.5; w CO₂ fert. or w/o CO₂ fert. - means with the effects of CO₂ fertilisation or without the effects of CO₂ fertilisation; w tech adaptation – by including technological adaptation measures; effect at the lower end of published estimates). "Region" includes: Naf – Northern Africa, SSA – Sub-Saharan Africa, WSSE – Western & Southern & Southeast Asia, CEA – are considered: NF - near time future (with projections ranging from 2020s-2030s), MF - medium time future (from 2040s-2060s) and LF - longtime future (from 2070s-2100).

thern Europe.	AFRIC	Maize; SSA [rain	P; [5; 25]%; MF	avg[CSIRO (A1B,	MIROC(A1B, B1)	Africa] (Dube et	Maize; SSA [Y ; (; NF ; (RCP4.5, R(Malawi] (Steven	Madani, 2016)	Maize; SSA [Y ; 8	RCP8.5; Guinea-F	(Ahmed, Wang, \	You, 2015)	Maize; SSA [Y; 2 [,]	RCP8.5 w adapt.	tolerance)] (Islan	2016)	Maize; SSA [Y ; 8	(RCP4.5 & RCP8.	Nitrogen applica	Ethiopia] (Kassie	2015)	Soybean; SSA [Y	; avg[CSIRO (A1B	MIROC(A1B, B1)	
	H	fed maize;		B1);	l; South	al., 2013)	5.4; 4.6)%	CP8.5);	s&		.9% ; MF ;	3issau]	'u, Koo, &		1%; MF;	(drought	n et al.,		4% ; MF ;	5) w	tion;	et al.,		; 20% ; MF	, B1);	l; South	
	EUROPE	Wheat; EU [P ; ~5% ; MF ;	RCP4.5 w O3 cont.] (Tai,	Martin, & Heald, 2014)	Wheat; EU [winter wheat; Y	; 0.2%/year ; (NF&MF) ;	A1FI; Denmark] (Martín,	Olesen, & Porter, 2014)	Rice; EU [P;~7%; MF;	RCP4.5 w O3 cont.] (Tai et	al., 2014)	Rice; EU [P;~12%; MF;	RCP8.5 w O3 cont.] (Tai et	al., 2014)													
	VISV	Wheat; CEA [P ; ~15% ; MF ; RCP4.5 w 03	cont.; China] (Tai et al., 2014)	Wheat; CEA [Y; (38, 19)% & (68, 23)% & (87,	34) % ; (NF) & (MF) & (LF) ; (A1Fl w CO2 fert.,	A1FI w/o CO2 fert.); China] (Tao & Zhang,	2013)	Wheat; WSSE [irrigated; P ; (2, 6)% ; MF ;	(A1B, B1); Iran] (Emam, Kappas, & Hosseini,	2015)	Maize; CEA [P ; ~5% ; MF ; RCP4.5 w O3	pollution; China] (Tai et al., 2014)	Rice; Asia [Y ; ~[20, 26]% ; MF ; A2 w CO2	fert.] (Li et al., 2015)	Soybean; CEA [P ; ~10% ; MF ; RCP4.5 w O3	pollution; China] (Tai et al., 2014)	Potatoes; Asia [Y; ~[3;10]%; MF; RCP8.5 w	adapt. (drought and heat tolerance)] (Islam	et al., 2016)								
	AMERICA	Wheat; NCA [P; 24%; MF; A1B;	USA] (Dawson, Perryman, &	Osborne, 2016)	Wheat; NCA [P;~10%; MF;	RCP4.5 w O3 cont.; USA] (Tai et	al., 2014)	Wheat; NCA [Y ; [20;60]% ; MF ;	A2 w tech advances; USA] (Jiang	& Koo, 2014)	Rice; NCA [P ; ~5% ; MF ; RCP4.5	w O3 cont.; USA] (Tai et al., 2014)	Rice; NCA [P; ~2%; MF; RCP8.5	w O3 cont.; USA] (Tai et al., 2014)													
	GLOBAL	Wheat; Glob [Y; (9.8, 16.9)% & (9.0,	34.3)% ; (MF)&(LF) ; (RCP8.5,	RCP2.6) w mean CC + extreme T	around crop anthesis (HSA) + direct	CO2 fert.] (Deryng, Conway,	Ramankutty, Price, & Warren, 2014)	Soybean; Glob [Y ; (9.5, 11.1)% &	(7.1, 15.3)%; (MF)&(LF); (RCP8.5,	RCP2.6) w mean CC + extreme T	around crop anthesis (HSA) + direct	CO2 fert.] (Deryng et al., 2014)															

Ξ	AFRICA	EUROPE	ASIA	AMERICA	GLOBAL
	Groundnuts; SSA [P; 73%; MF; avg [CSIRO (A1B, B1); MIROC (A1B, B1)]; South Africa] Africa] Africa] Vams; SSA [γ ; [7 ; 49]% ; MF ; A1B w adap.; Republic of Bénin] (Srivastava, Gaiser, & Ewert, 2015) Groundnuts; SSA [γ ; ~ [-3;15]% ; MF; RCP 8.5 w adapt. (drought and heat tolerance)] (Islam et al., 2016)				
cte	Soybean; SSA [P ; 0% ; MF ; avg[CSIRO (A1B, B1); MIROC(A1B, B1)]; South Africa] (Dube et al., 2013)		Rice; WSSE [P ; 0% ; MF ; (RCP8.5&RCP4.5) w O3 pollution] (Tai et al., 2014) Soybean; WSSE [P ; 0% ; MF ; RCP4.5 w O3 pollution] (Tai et al., 2014)	Rice; NCA [P ; 0% ; MF ; (RCP8.5&RCP4.5) w O3 pollution] (Tai et al., 2014)	Rice; Glob [P ; 0% ; MF ; RCP8.5 w O3 pollution] (Tai et al., 2014)
÷	; LF ; A1B; Algeria] (Chourghal, Lhomme, Huard, & Aidaoui, 2015) Maize; SSA [Y : -20% ; MF ; (RCP4.5 & RCP8.5); Ethiopia] (Kassie et al., 2015) Maize; SSA [Y ; (-1.2; 1)% & (-3; 0.2; (MF) & (LF) ; (RCP4.5, RCP8.5); Malawi] (Stevens & Madani, 2016) Maize; SSA [Y ; [-20; 0]% ; MF ; A1B w 380 ppm C02; Nigeria] (Mereu, Carboni, Gallo, Cervigni, & Spano, 2015) Maize; SSA [Y ; [-40.1; -	A1B; UK] (Dawson et al., 2016) Wheat; EU [P; ~~12%; MF; RCP8.5 w O3 cont.] (Tai et al., 2014) Maize; EU [P; (~~40%, - 30%); MF ; (RCP8.5, RCP4.5) w O3 pollution] (Tai et al., 2014) Soybean; EU [P; (~~20%, - 10%); MF ; (RCP8.5, RCP4.5) w O3 pollution] (Tai et al., 2014) (Tai et al., 2014) Oil seeds, Wheat, Rice; EU [Y; ~~15%; MF; RCP8.5] (Nelson et al., 2014)	cont.; China] (Tai et al., 2014) Wheat; CEA [irrigated; Y ; (-6.7, -2.2)% ; MF ; (A2, B2); China] (Ju, van der Velde, Lin, Xiong, & Li, 2013) Wheat; WSSE [P ; ~-35%, -30% ; MF ; (RCP8.5, RCP4.4) w O3 cont.; South Asia] (Tai et al., 2014) Wheat; WSSE [P ; (~-10%, -5%) ; MF ; (RCP8.5, RCP4.4) w O3 cont.; Southeast Asia] (Tai et al., 2014) Wheat; WSSE [P ; (~-10%, -5%) ; MF ; (A1B, B1); Iran] (Emam et al., 2015) Wheat; WSSE [Y ; -18% ; MF ; CSIRO-A1B; Pakistan] (Zhu, Ringler, Mohsin Iqbal, Sulser, & Arif Goheer, 2013) Wheat; Asia [Y; ~-7% ; MF ; RCP8.5; Bangladesh, India, Nepal, Pakistan] /[Y ; ~-2%	RCP8.5 w O3 cont.; USA] (Tai et al., 2014) Wheat; NCA [Y ; <-30% ; MF ; A2; USA] (Jiang & Koo, 2014) Wheat; NCA [Y ; -10% ; MF ; A2&B1 Mexico] (Rosenzweig et al., 2014) Wheat; CSA [P ; ~-10%, -5% ; MF ; (RCP8.5, RCP4.5) w O3 cont.] (Tai et al., 2014) Maize; NCA [P ; (~-50%, -45%) ; MF ; (RCP4.5, RCP8.5) w O3 pollution; USA] (Tai et al., 2014) Maize; CSA [P ; (~-25%, -20%) ; MF ; (RCP8.5, RCP4.5) w O3 pollution] (Tai et al., 2014)	(Rosenzweig et al., 2014) Wheat; Glob [Y ; [-15; +5]%; LF ; RCP8.5 w CO2fert.] (Müller et al., 2015) Wheat; Glob [P ; -40% ; MF ; A1B] (Dawson et al., 2016) Wheat; Glob [P ; ~-15%, -5% ; MF ; (RCP8.5, RCP4.4) w O3 cont.] (Tai et al., 2014) Wheat; Glob [Y ; -5.2%, -2.5% ; NF ; RCP8.5 (20, 10 year period)] (Lobell & Tebaldi, 2014) Wheat; Glob [Y ; [-15; +5]% i. LF ; RCP8.5 w CO2fert.] (Müller et al., 2015) Wheat; Glob [Y ; (-10.1, -4.5)% & (-24.1, -2.9)%; (MF)&(LF); (RCP8.5,
]				-	103

FI AFRICA	EUROPE	ASIA	AMERICA	GLOBAL
Gambia (Ahmed et al.,		; MF; RCP8.5 w adapt. (drought and heat	Soybean; NCA [P; (~-10%, -5%) ;	around crop anthesis (HSA) w/o
2015)		tolerance)] (Islam et al., 2016)	MF; (RCP8.5, RCP4.5) w O3	direct CO2 fert.] (Deryng et al.,
Maize; SSA [Y ; <-8% ; MF ;		Maize; CEA [P ; (~-6%) ; MF ; (RCP8.5,	pollution; USA] (Tai et al., 2014)	2014)
(RCP4.5, RCP8.5); Burkina		RCP4.5) w O3 pollution; China] (Tai et al.,	Soybean; CSA [P ; (~-25%, -20%) ;	Maize; Glob [Y ; [-45; -20]% ; LF ; A2]
Faso] (Waongo, Laux, &		2014)	MF; (RCP8.5, RCP4.5) w O3	(Rosenzweig et al., 2014)
Kunstmann, 2015)		Maize; CEA [irrigated; Y ; (-11.9, -0.4)% ; MF ;	pollution] (Tai et al., 2014)	Maize; Glob [P ; -50% ; MF ; A1B]
Maize; SSA [Y ; ~-6% ; MF ;		(A2, B2); China] (Ju et al., 2013)	Oil seeds, Wheat, Rice; NCA [Y;	(Dawson et al., 2016)
RCP8.5] Angola, Benin,		Maize; WSSE [P ; (~-8%, -5%); MF ; (RCP8.5,	~-20% ; MF ; RCP8.5; USA]	Maize; Glob [P; (~-30%,-20%); MF;
Ethiopia, Ghana, Kenya,		RCP4.5) w O3 pollution; South Asia] (Tai et	(Nelson et al., 2014)	(RCP8.5, RCP4.5) w O3 pollution]
Malawi, Mozambique,		al., 2014)	Oil seeds, Wheat, Rice; CSA [Y ; ~-	(Tai et al., 2014)
Uganda, United Republic of		Maize; WSSE [P ; ~-35% ; MF ;	20% ; MF ; RCP8.5; Brazil]	Maize; Glob [Y ; [-5.3; -2.6]% ; NF ;
Tanzania, Zambia,		(RCP8.5&RCP4.5) w O3 pollution; Southeast	(Nelson et al., 2014)	RCP8.5] (Lobell & Tebaldi, 2014)
Zimbabwe; (Islam et al.,		Asia] (Tai et al., 2014)		Maize; Glob [Y ; [-35; -10]% ; LF ;
2016)		Maize; WSSE [Y ; -21% ; MF ; MIROC-A1B;		RCP8.5 w CO2fert.] (Müller et al.,
Rice; SSA [Y ; [-20; 0]% ; MF		Pakistan] (Zhu et al., 2013)		2015)
; A1B w 380 ppm CO2;		Rice; CEA [P; ~-2%; MF; RCP8.5 w O3		Maize; Glob [Y ; (-7.4, -3.1)% & (-
Nigeria] (Mereu et al.,		pollution; China] (Tai et al., 2014)		12.8, -2.9)% ; (MF)&(LF) ; (RCP8.5,
2015)		Rice; CEA [irrigated; Y ; (-12.4, -4.3)% ; MF ;		RCP2.6)] (Deryng et al., 2014)
Millet; Africa [Y ; -6% ; LF ;		(A2, B2); China] (Ju et al., 2013)		Maize; Glob [Y ; (-10.5, -4.7)% & (-
(A1B & A2); Africa & India]		Rice; CEA [Y ; -18.9% ; LF ; (A2 & B2); China]		22.0, -4.4)% ; (MF)&(LF) ; (RCP8.5,
(Berg, De Noblet-Ducoudré,		(W. Wang et al., 2014)		RCP2.6) w mean CC + extreme T
Sultan, Lengaigne, &		Rice; WSSE [P; ~-1%; MF; RCP8.5 w O3		around crop anthesis (HSA) w/o
Guimberteau, 2013)		pollution] (Tai et al., 2014)		direct CO2 fert.] / [Y ; (-7.4, -3.1)% &
Sorghum & Millet; SSA [Y ;		Rice; WSSE [Y; (<-20)% & (<-27.8)% ;		(-12.8, -2.9)% ; (MF)&(LF) ; (RCP8.5,
[-30; 0]% ; MF; A1B w 380		(NF)&(MF) ; A2; India] (Banerjee, Das, &		RCP2.6) w mean CC + extreme T
ppm CO2; Nigeria] (Mereu		Mukherjee, 2014)		around crop anthesis (HSA) + direct
et al., 2015)		Rice; WSSE [Y ; [-31;-18]% ; MF ; (MIROC-		CO2 fert.] (Deryng et al., 2014)
Sorghum & Millet; SSA [Y ;		A1B,CSIRO-A1B); Pakistan] (Zhu et al., 2013)		Rice; Glob [Y ; [-30; -20]%; LF ; A2]
[-41; 0]% ; LF ; CMIP3 (A2,		Rice; Asia [Y ; ~[-55, -5]% ; MF ; A2] (Li et al.,		(Rosenzweig et al., 2014)
A1B, B1), CMIP5(RCP 4.5,		2015)		Rice; Glob [Y ; [-20; -5]% ; LF ; RCP8.5
6.0 and 8.5); Senegal, Mali,		Soybean; CEA [P ; ~-5% ; MF ; RCP8.5 w O3		w CO2fert.] (Müller et al., 2015)
Burkina Faso and Niger]		pollution; China] (Tai et al., 2014)		Soybean; Glob [Y ; [-60; -30]% ; LF ;
(Sultan et al., 2013)		Soybean; WSSE [P ; ~-5%,-2% ; MF ; RCP8.5		A2] (Rosenzweig et al., 2014)
Sorghum & Millet; SSA [Y ;		w O3 pollution; South Asia, Southeast Asia]		Soybean; Glob [P ; -50% ; MF ; A1B]
[-45.5; -4]% ; MF ; RCP8.5;		(Tai et al., 2014)		(Dawson et al., 2016)
Senegal, Gambia, Mali,		Oil seeds, Wheat, Rice; CEA [Y ; ~-10% ; MF ;		Soybean; Glob [P ; ~-15%, -8% ; MF ;
lvory Coast, Togo, Ghana,		RCP8.5; China] (Nelson et al., 2014)		(RCP8.5, RCP4.5) w O3 pollution]
Sierra Leone, Guinea-Bissau		Oil seeds, Wheat, Rice; WSSE [Y ; ~-20% ; MF		(Tai et al., 2014)
] (Ahmed et al., 2015)		; RCP8.5; India] (Nelson et al., 2014)		

E	AFRICA	EUROPE	ASIA	AMERICA	GLOBAL
	Sorghum & Millet; SSA [Y ;		Oil seeds, Wheat, Rice; WSSE [Y ; ~-10%; MF		Soybean; Glob [Y ; [-30; 0]% ; LF ;
_	~-5% ; MF ; RCP8.5 w		; RCP8.5] (Nelson et al., 2014)		RCP8.5 w CO2fert.] (Müller et al.,
	adapt. (drought and heat		Potatoes; Asia [Y ; ~-1% ; MF ; RCP8.5] (Islam		2015)
	tolerance); Burkina Faso,		et al., 2016)		Soybean; Glob [Y ; (–6.9 , 1.9)% &
_	Eritrea, Ethiopia, Mali,				(-26.0, 0.9)% ; (MF)&(LF) ; (RCP8.5,
	Nigeria, Sudan, United				RCP2.6) w mean CC + extreme T
_	Republic				around crop anthesis (HSA) w/o
_	of Tanzania] (Islam et al.,				direct CO2 fert.] (Deryng et al.,
_	2016)				2014)
_	Oil seeds, Wheat, Rice; SSA				
	[Y;~-20%;MF;RCP8.5]				
	(Nelson et al., 2014)				
_	Cassava; SSA [Y ; ~[-15; -				
_	5]% ; MF ; A1B w 380 ppm				
_	CO2; Nigeria] (Mereu et al.,				
	2015)				
	Yams; SSA [Y ; (-33; -27)% ;				
	MF ; A1B, B1; Republic of				
	Bénin] (Srivastava et al.,				
_	2015)				
	Groundnuts; SSA [Y ; ~-9% ;				
_	MF ; RCP8.5] (Islam et al.,				
	2016)				
	Various crops; SSA [Y ; [-37;				
_	-11]% ; MF ; RCP8.5]				
	(Hertel & Baldos, 2016)				

Table A3 Projections of climate change and climate variability impacts on food access by considering all reviewed crops

change; time scale; scenario; location; other (when adeauate)] Literature reference". Food indicators (FI) are PC - Price; UND - Prevalence of undernourishment. A projected change on FI is represented by (+) when the change is positive, (-) when it decreases and "cte" when there is no change. Three time scales were considered: NF - near time future (with projections ranging IPCC Representative Concentration Pathways (RCP): RCP2.6, RCP4.5, RCP6, RCP8.5; w CO2 fert. or w/o CO2 fert. - means with the effects of CO2 fertilisation or without the effects of CO2 Notes: Variations always refer to reference scenarios, which may differ from a study to another. Date is organised as follows: "Region: [food product (when adequate); food indicator; projection from 2020s-2030s), MF - medium time future (from 2040s-2060s) and LF - longtime future (from 2070s-2100). IPCC Special Report Emissions Scenarios (SRES): A1, A1FI, A1T, A1B, A2, B1, B2; fertilisation; w tech adaptation – by including technological adaptation measures; HCI – high climate index (high temperature increase, high sensitivity of crops to global warming, and a CO₂ ertilisation effect at the lower end of published estimates). "Regions" include: Naf – North Africa, SSA – Sub-Saharan Africa, NWC – North & West & Central Asia, WSSE – South & Southeast asia. CEA – East Asia. NCA – North America. CSA – Caribbean & South America. EU – Europe.

	Global	oil crops, sugar crops, tubers,	lob [PC;[0, 5]%;MF;A2]	r et al., 2014)	crops; Glob (various crops) [PC ;	VF;RCP8.5] / [und;45%;MF;	(Hertel & Baldos, 2016)	^f ood products; Glob [und ; 50% ;	3; South America, Africa, Australia	ral Asia and partially Europe,	ist Asia, USA and Russia] (Dawson	16)			
		Cereals, o	fibers; Gl	(Mosnier	Various c	27.6% ; N	RCP8.5] (Various f	MF ; A1B	and cent	South-Ea	et al., 20			
	America	Oil seeds, Wheat,	Rice; America [PC;	~10%, 15%; MF;	RCP8.5 & avg. Econ.	models ; Brazil, USA]	(Nelson et al., 2014)								
	Asia	Oil seeds, Wheat, Rice; Asia [PC; ~15%, 25%; MF;	RCP8.5 & avg. Econ. models ; China, India] (Nelson et	al., 2014)	Maize, Wheat, Rice; WSSE [PC; [14;18]%, [18;27]%,	[26;32]% ; MF ; (CSIRO-B1,MIROC-A1); Pakistan] (Zhu	et al., 2013)	Cereals, oil crops, sugar crops, tubers, fibers; CEA [PC	; [-1; 4]%, [1; 6]%, 38%; MF ; A2; South Korea, China,	Mongolia] (Mosnier et al., 2014)	Cereals, oil crops, sugar crops, tubers, fibers; WSSE	[und ; 97% ; MF ; RCP8.5, South Asia] (Hertel &	Baldos, 2016)	Various crops; CEA [PC; [-5; -3]%; MF; A2; Japan]	(Mosnier et al., 2014)
	Europe	Oil seeds, Wheat,	Rice; EU [PC;	~10%; MF; RCP8.5	& avg. Econ.	models] (Nelson	et al., 2014)								
	Africa	Soybean; SSA [PC; 60%; MF;	avg[CSIRO (A1B, B1); MIROC(A1B,	B1)]; South Africa] (Dube et al.,	2013)	Oil seeds, Wheat, Rice; SSA [PC;	~15%; MF; RCP8.5 & avg. Econ.	models] (Nelson et al., 2014)	SSA (various crops) [PC ; 22.9% ;	MF ; RCP8.5] (Hertel & Baldos,	2016)				
, picc	F						17	E							Ē

References

- Ahmed, K. F., Wang, G., Yu, M., Koo, J., & You, L. (2015). Potential impact of climate change on cereal crop yield in West Africa. Climatic Change, 133(2), 321–334. https://doi.org/10.1007/s10584-015-1462-7 ÷
 - Asseng, S., Ewert, F., Martre, P., Rötter, R. P., Lobell, D. B., Cammarano, D., ... Zhu, Y. (2015). Rising temperatures reduce global wheat production. Nature Climate Change, 5(2), 143–147. https://doi.org/10.1038/nclimate2470 ų.
- Avelino, J., Cristancho, M., Georgiou, S., Imbach, P., Aguilar, L., Bornemann, G., ... Morales, C. (2015). The coffee rust crises in Colombia and Central America (2008– 2013): impacts, plausible causes and proposed solutions. *Food Security*, 7(2), 303–321. https://doi.org/10.1007/s12571-015-0446-9 ć.
 - Banerjee, S., Das, S., & Mukherjee, A. (2014). Adaptation strategies to combat climate change effect on rice and mustard in Eastern India. Mitig Adapt Strateg Glob *Change*. https://doi.org/10.1007/s11027-014-9595-y 4
- Berg, A., De Noblet-Ducoudré, N., Sultan, B., Lengaigne, M., & Guimberteau, M. (2013). Projections of climate change impacts on potential C4 crop productivity over tropical regions. Agricultural and Forest Meteorology, 170, 89–102. https://doi.org/10.1016/j.agrformet.2011.12.003 ы.
 - Broberg, M. C., Feng, Z., Xin, Y., & Pleijel, H. (2015). Ozone effects on wheat grain quality e A summary. Environmental Pollution, 197, 203–213. https://doi.org/10.1016/j.envpol.2014.12.009 .

 y. Y. (2015). The effictent, 148, 1–9. https: climate change on doi.org/10.1017/S0 A. (2015). Climate on doi.org/10.1017/S0 A. (2016). Modelling im 2016). The impact of weather of tudies, 16(3), 279–3 (2013). The impact of weather of tudies, 16(3), 279–3 (2013). The impact of weather of tudies, 16(3), 279–3 (2013). The impact of weather of tudies, 16(3), 279–3 (2013). The impact of weather of tudies, 16(3), 279–3 (2013). The impact of weather of tudies, 16(3), 279–3 (2013). The impact of weather of tudies, 16(3), 279–3 (2013). The impact of weather of tudies, 16(3), 279–3 (2013). The impact of weather of tudies, 16(3), 279–3 (2013). The impact of weather of tudies, 16(3), 279–3 (2013). The impact of weather of tudies, 16(3), 279–3 (2013). The impact of weather of tudies, 16(3), 279–3 (2013). The impact of weather of tudies, 16(10, 012/10.108). A genotype, envir os://doi.org/10.108). A genotype, envir os://doi.org/10.108). Of climate trends of the organ of climate trends of the organ of trends of the organ of trends of the organ of
--

2	 Mosnier, A., Obersteiner, M., Havlík, P., Schmid, E., Khabarov, N., Westphal, M., Albrecht, F. (2014). Global food markets, trade and the cost of climate change adaptation. <i>Food Security</i>, 1–16. https://doi.org/10.1007/s12571-013-0319-z
2	 Müller, C., Elliott, J., And, Chryssanthacopoulos, J., Deryng, D., Folberth, C., Pugh, T. A. M., & Schmid, E. (2015). Implications of climate mitigation for future agricultura production. <i>Environmental Research Letters</i>, 10(12), 125004. https://doi.org/10.1088/1748-9326/10/12/125004
2	 Nelson, G. C., Valin, H., Sands, R. D., Havlík, P., Ahammad, H., Deryng, D., & Elliott, J. (2014). Climate change effects on agriculture : Economic responses to biophysical shocks. PNAS, 111(9). https://doi.org/10.1073/pnas.1222465110
5	 Ray, D. K., Gerber, J. S., MacDonald, G. K., & West, P. C. (2015). Climate variation explains a third of global crop yield variability. Nature Communications, 6, 5989. https://doi.org/10.1038/ncomms6989
2	 Ray, D. K., West, P. C., Clark, M., Gerber, J. S., Prishchepov, A. V, & Chatterjee, S. (2019). Climate change has likely already affected global food production. PLOS ONE, 14(5), e0217148. Retrieved from https://doi.org/10.1371/journal.pone.0217148
2:	29. Rosenzweig, C., Jones, J. W., Hatfield, J. L., Ruane, A., Thornburn, K. J., Antle, J. M., Winter, J. M. (2014). The Agricultural Model Intercomparison and Improvement
3	Project (AgMIP): Protocols and pilot studies, 35. https://doi.org/10.1016/j.agrformet.2012.09.011 30 Srivastava & K. Gaiser T. & Ewert E. (2015). Climate change impact and notential adantation strategies under alternate climate scenarios for vam production in the
)	sub-humid savannah zone of West Africa. <i>Mitig Adapt Strateg Glob Change</i> . https://doi.org/10.1007/s11027-015-9639-y
Ċ	31. Stevens, T., & Madani, K. (2016). Future climate impacts on maize farming and food security in Malawi. Scientific Reports, 6(October), 36241.
č	nttps://doi.org/1u.1u38/srep36241
m	32. Sultan, B., Roudier, P., Quirion, P., Alhassane, A., Muller, B., Dingkuhn, M., Baron, C. (2013). Assessing climate change impacts on sorghum and millet yields in the Sudanian and Sabelian sevannas of West Africa. <i>Environmental Research Letters</i> , 8(1): 14040. https://doi.org/10.1088/1748-0326/8/1/014040
ñ	33. Tai, A. P. K., Martin, M. V., & Heald, C. L. (2014). Threat to future global food security from climate change and ozone air pollution. <i>Nature Climate Change</i> ,
	4(September), 817–821. https://doi.org/10.1038/nclimate2317
'n	34. Tao, F., & Zhang, Z. (2013). Climate change, wheat productivity and water use in the North China Plain: A new super-ensemble-based probabilistic projection.
31	Agriculture and rolest meteorology, 170, 140–103. Inteps.// doi.org/ 10.1016/j.agrionmet.2011.10.003 35 Mang 1 Jiu X Zhang X Smith P Ji J Fillev T R Pan G (2016) Size and variability of crop productivity both impacted by CO2 enrichment and warming-A case
)	study of 4 year field experiment in a Chinese paddy. <i>Agriculture, Ecosystems and Environment, 221</i> , 40–49. https://doi.org/10.1016/j.agee.2016.01.028
31	36. Wang, W., Yu, Z., Zhang, W., Shao, Q., Zhang, Y., Luo, Y., Xu, J. (2014). Responses of rice yield, irrigation water requirement and water use efficiency to climate
	change in China: Historical simulation and future projections. Agricultural Water Management, 146, 249–261. https://doi.org/10.1016/j.agwat.2014.08.019
'n	37. Waongo, M., Laux, P., & Kunstmann, H. (2015). Adaptation to climate change: The impacts of optimized planting dates on attainable maize yields under rainfed
č	conditions in Burkina Faso. <i>Agricultural and Forest Meteorology, 20</i> 5, 23–39. https://doi.org/10.1016/j.agrformet.2015.02.006
ň	38. Wel, L., Cherry, L. L., Glomrigu, S., & Zhang, T. (2014). Climate change impacts on crop yrem: Evidence nom China. Science of the Total Erivitorinicit, 433, 133–140. Retrieved from http://dx.doi.org/10.1016/i.scitotenv.2014.08.035
ŝ	39. Zhao, J., Guo, J., Mu, J., & Xu, Y. (2015). Exploring the dynamics of agricultural climatic resource utilization of spring maize over the past 50 years in Northeast China.
	Physics and Chemistry of the Earth, 87–88, 19–27. https://doi.org/10.1016/j.pce.2015.07.008
4	40. Zhou, Y., Li, N., Dong, G., & Wu, W. (2013). Impact assessment of recent climate change on rice yields in the heilongjiang reclamation area of north-east china. <i>Journal</i>
	of the Science of Food and Agriculture, 93(11), 2698–2706. https://doi.org/10.1002/jsta.6087
4	 Zhu, T., Ringler, C., Mohsin Iqbal, M., Sulser, T. B., & Arif Goheer, M. (2013). Climate change impacts and adaptation options for water and food in Pakistan: scenario analysis using an integrated global water and food projections model. Water International, 38(5), 651–669. https://doi.org/10.1080/02508060.2013.830682

6.2 Appendix B

Exposure of the EU food imports to extreme weather disasters in exporting countries

EWD	Region/ country	Period of occurrence	Negative impact	Reference
Drought and Heat wave	Italy, Germany, Austria, Spain, France and Portugal	2003	10 Mt of wheat and 9 Mt of maize	COPA-COGECA, 2003)
Drought	Slovenia	2003	100 M€ for agriculture sector	(EEA, 2010)
Drought	South and Central Europe	2003	20% on grain harvested	(Kovats et al., 2014)
Drought	Iberia Peninsula	2004-2005	40% on cereal production	(EEA, 2010)
Flood	England and Wales	2007	50M£ for agriculture sector	(Chatterton, Viavattene, Morris, Penning-Rowsell, & Tapsell, 2010)
Heat wave	France	2011	8% on wheat yield	(AGRESTE, 2011)

Table B1. Observed losses in European agriculture attributed to extreme weather disasters.

	Soy	beans			Bana	ana	
Floods	Droughts	Heat waves	Cold waves	Floods	Droughts	Heat waves	Cold waves
(n=43)	(n=27)	(n=14)	(n=18)	(n=71)	(n=29)	(n=3)	(n=6)
1964 BRA	1970 BRA	1968 BRA	1975 BRA	1979 BLZ	1970 BRA	1968 BRA	1990 BLZ
1969 BRA	1977 BRA	1975 BRA	1988 BRA	1990 BLZ	1977 BRA	1975 BRA	1975 BRA
1973 BRA	1983 BRA	2010 BRA	1994 BRA	1995 BLZ	1983 BRA	2010 BRA	1988 BRA
1977 BRA	1985 BRA	1966 USA	2000 BRA	2008 BLZ	1985 BRA		1994 BRA
1983 BRA	1987 BRA	1972 USA	2004 BRA	1964 BRA	1987 BRA		2000 BRA
1995 BRA	1994 BRA	1980 USA	1982 CAN	1969 BRA	1994 BRA		2004 BRA
1974 CAN	1998 BRA	1983 USA	1992 CAN	1973 BRA	1998 BRA		
1976 CAN	2001 BRA	1986 USA	2000 PRY	1977 BRA	2001 BRA		
1979 CAN	2004 BRA	1990 USA	2004 PRY	1983 BRA	2004 BRA		
1983 CAN	2007 BRA	1993 USA	2010 PRY	1995 BRA	2007 BRA		
1986 CAN	2010 BRA	1995 USA	2000 UKR	1989 CIV	2010 BRA		
1990 CAN	2012 BRA	1998 USA	2009 UKR	1996 CIV	2012 BRA		
1993 CAN	1977 CAN	2005 USA	2012 UKR	2007 CIV	1983 CIV		
1995 CAN	1984 CAN	2011 USA	1977 USA	2010 CIV	1971 CMR		
2002 CAN	1988 CAN		1989 USA	1988 CMR	1990 CMR		
2011 CAN	1983 PRY		1995 USA	1991 CMR			
1965 PRY	1999 PRY		2004 USA	1994 CIVIR	2005 CIVIR		
1971 PRT	2005 PRT		2009 USA	1999 CIVIR	2012 CIVIR		
1979 PKT	2006 PKT			2003 CIVIR	1996 COL		
1982 PKT	2012 FKT			2007 CIVIR	1973 CRI		
1988 PK1	1088 USA			2010 CMR	1994 CRI		
1992 PRV	1991 USA			1969 COI	1968 DOM		
1995 PRY	1999 USA			1973 COL	1964 FCU		
1997 PRY	2002 USA			1976 COL	1997 FCU		
2002 PRY	2002 USA			1979 COL	2009 FCU		
2009 PRY	2011 USA			1984 COL	2013 FCU		
2012 PRY	2011 00/1			1986 COL	1983 PAN		
1995 UKR				1993 COL	2013 PAN		
2001 UKR				1999 COL			
2003 UKR				1969 CRI			
2005 UKR				1980 CRI			
2008 UKR				1988 CRI			
2010 UKR				1991 CRI			
2013 UKR				1993 CRI			
1964 USA				1996 CRI			
1969 USA				1999 CRI			
1972 USA				2001 CRI			
1976 USA				2007 CRI			
1980 USA				2010 CRI			
1988 USA				1979 DOM			
1990 USA				1981 DOM			
1993 USA				1985 DOM			
				1988 DOM			
				1991 DOM			
				100C DOM			
				1996 DOW			
				1965 FCU			
				1967 FCU			
I	I	I	I	1 2007 200	I	I	I I

Table B2. List of the extreme weather disasters that happened per year in the exporting countries (ISO3 codes) supplying the EU with soybeans and banana. Data was aquired through (FAO, 2017).

	Soy	beans			Bana	ana	
Floods	Droughts	Heat waves	Cold waves	Floods	Droughts	Heat waves	Cold waves
(n=43)	(n=27)	(n=14)	(n=18)	(n=71)	(n=29)	(n=3)	(n=6)
				1970 ECU			
				1973 ECU			
				1982 ECU			
				1987 ECU			
				1989 ECU			
				1992 ECU			
				1997 ECU			
				2000 ECU			
				2006 ECU			
				2008 ECU			
				1966 PAN			
				1970 PAN			
				1972 PAN			
				1978 PAN			
				1984 PAN			
				1986 PAN			
				1991 PAN			
				1995 PAN			
				1999 PAN			

Table B3. List of the extreme weather disasters that happened per year in the exporting countries (ISO3 codes) supplying the EU with tropical fruits and coffee. Data was aquired through (FAO, 2017).

Note: Data on tropical fruits from Panama and on coffee from Ethiopia, were excluded from the analysis once yield values were not correctly reported or there is missing data at FAOSTAT. Those countries are, therefore, not indicated in this table.

			Tropic	al Fruits	5						Co	ffee			
Flood	ds	Drou	ghts	Heat	waves	Cold v	waves	Flo	ods	Drou	ughts	Heat	waves	Cold	waves
(n=14)	3)	(n=	70)	(n=	17)	(n=	28)	(n=:	141)	(n=:	109)	(n=	16)	(n=	28)
1964 B	BRA	1970	BRA	1968	BRA	1975	BRA	1964	BRA	1970	BRA	1968	BRA	1975	BRA
1969 B	BRA	1977	BRA	1975	BRA	1988	BRA	1969	BRA	1977	BRA	1975	BRA	1988	BRA
1973 B	BRA	1983	BRA	2010	BRA	1994	BRA	1973	BRA	1983	BRA	2010	BRA	1994	BRA
1977 B	BRA	1985	BRA	2003	DZA	2000	BRA	1977	BRA	1985	BRA	1965	IND	2000	BRA
1983 B	BRA	1987	BRA	2000	ISR	2004	BRA	1983	BRA	1987	BRA	1978	IND	2004	BRA
1995 B	BRA	1994	BRA	1968	MEX	1995	CHL	1995	BRA	1994	BRA	1985	IND	2001	GTM
1965 C	CHL	1998	BRA	1990	MEX	2000	CHL	1989	CIV	1998	BRA	1987	IND	2006	GTM
1974 C	CHL	2001	BRA	1975	PAK	2002	CHL	1996	CIV	2001	BRA	1994	IND	2011	GTM
1978 0	CHL	2004	BRA	1979	PAK	2004	CHL	2007	CIV	2004	BRA	1998	IND	1973	IND
1982 0	CHL	2007	BRA	1991	PAK	2010	CHL	2010	CIV	2007	BRA	2002	IND	1980	IND
1986 C	CHL	2010	BRA	1996	PAK	1992	ISR	1988	CMR	2010	BRA	2005	IND	1984	IND
1989 C	CHL	2012	BRA	1999	PAK	1988	MEX	1994	CMR	2012	BRA	2009	IND	1989	IND
1993 C	CHL	1968	CHL	2002	PAK	1992	MEX	1999	CMR	1983	CIV	2013	IND	1992	IND
1995 C	CHL	1991	CHL	2005	PAK	1995	MEX	2005	CMR	1971	CMR	1968	MEX	1998	IND
1997 0	CHL	1983	CIV	1983	PER	1997	MEX	2007	CMR	2001	CMR	1990	MEX	2000	IND
2000 0	CHL	1973	CRI	2000	TUR	2002	MEX	2010	CMR	2005	CMR	1983	PER	2007	IND
2004 0	CHL	1994	CRI	2007	TUR	2006	MEX	2012	CMR	2012	CMR			2010	IND
2008 0	CHL	1998	CRI			2011	MEX	1969	COL	1998	COL			1988	MEX
2012 0	CHL	1981	DZA			1990	PAK	1973	COL	1973	CRI			1992	MEX
1989 C	CIV	1964	ECU			2001	PAK	1976	COL	1994	CRI			1995	MEX
1996 C	CIV	1997	ECU			1991	PER	1979	COL	1998	CRI			1997	MEX

	Tropic	al Fruits			Co	ffee	
Floods	Droughts	Heat waves	Cold waves	Floods	Droughts	Heat waves	Cold waves
(n=143)	(n=70)	(n=17)	(n=28)	(n=141)	(n=109)	(n=16)	(n=28)
2007 CIV	2009 ECU		2003 PER	1984 COL	1987 GTM		2002 MEX
2010 CIV	2013 ECU		2009 PER	1986 COL	1994 GTM		2006 MEX
1969 CRI	1965 HND		1987 TUR	1993 COL	2001 GTM		2011 MEX
1980 CRI	1972 HND		2001 TUR	1999 COL	2009 GTM		1991 PER
1988 CRI	1994 HND		2004 TUR	1969 CRI	2012 GTM		2003 PER
1991 CRI	1997 HND		1996 ZAF	1980 CRI	1965 HND		2009 PER
1993 CRI	2000 HND		2007 ZAF	1988 CRI	1972 HND		2006 SLV
1996 CRI	2004 HND			1991 CRI	1994 HND		
1999 CRI	2009 HND			1993 CRI	1997 HND		
2001 CRI	2012 HND			1996 CRI	2000 HND		
2007 CRI	1964 IRN			1999 CRI	2004 HND		
2010 CRI	1999 IRN			2001 CRI	2009 HND		
1966 DZA	1999 ISR			2007 CRI	2012 HND		
1969 DZA	1965 KEN			2010 CRI	1966 IDN		
1973 DZA	1971 KEN			1973 GTM	1972 IDN		
1979 DZA	1979 KEN			1982 GTM	1978 IDN		
1981 DZA	1984 KEN			1987 GTM	1982 IDN		
1984 DZA	1991 KEN			1994 GTM	1984 IDN		
1992 DZA	1994 KEN			1999 GTM	1986 IDN		
1996 DZA	1997 KEN			2002 GTM	1997 IDN		
1999 DZA	1999 KEN			2005 GTM	2003 IDN		
2011 DZA	2004 KEN			2007 GTM	1964 IND		
1965 ECU	2008 KEN			1965 HND	1972 IND		
1967 ECU	2011 KEN			1976 HND	1979 IND		
1970 ECU	1978 MEX			1979 HND	1982 IND		
1973 ECU	1988 MEX			1981 HND	1987 IND		
1982 ECU	1995 MEX			1984 HND	1993 IND		
1987 ECU	1999 MEX			1986 HND	1996 IND		
1989 ECU	2002 MEX			1988 HND	2000 IND		
1992 ECU	2011 MEX			1990 HND	2002 IND		
1997 ECU	1999 PAK			1993 HND	2009 IND		
2000 ECU	1966 PER			1999 HND	1965 KEN		
2006 ECU	1969 PER			2002 HND	1971 KEN		
2008 ECU	1983 PER			2005 HND	1979 KEN		
1965 HND	1990 PER			2010 HND	1984 KEN		
1976 HND	1992 PER			1966 IDN	1991 KEN		
1979 HND	2002 PER			1970 IDN	1994 KEN		
1981 HND	2004 PER			1976 IDN	1997 KEN		
1984 HND	2006 PER			1998 IDN	1999 KEN		
1986 HND	1977 TUN			1964 IND	2004 KEN		
1988 HND	1988 TUN			1966 IND	2008 KEN		
1990 HND	1964 ZAF			1970 IND	2011 KEN		
1993 HND	1980 ZAF			1974 IND	1978 MEX		
1999 HND	1982 ZAF			1977 IND	1988 MEX		
2002 HND	1986 ZAF			1964 KEN	1995 MEX		
2005 HND	1988 ZAF			1968 KEN	1999 MEX		
2010 HND	1991 ZAF			1975 KEN	2002 MEX		
1966 IRN	1995 ZAF			1977 KEN	2011 MEX		

Prought Prought Prought Prought Preatwares Coldwares 1968 IRN 2004 ZAF [n=17] 1982 KEN 1994 NC [n=28] 1997 NC 1980 IRN 2004 ZAF [n=17] 1990 KEN 1997 NC [n=28] KEN 1997 NC 1986 IRN [n=17] 1960 KEN 1967 NE 1966 FR [n=17] [n=17] 1997 ISR [n=16] [n=17] MEX 1969 PR [n=17] 1997 ISR [n=16] [n=16] 1970 MEX 1990 PR [n=16] [n=17] 1997 ISR [n=16] [n=16] 1970 MEX 1990 PR [n=16] [n=16]		Tropic	al Fruits					Co	ffee	
1988 IRN 2004 ZAF 1980 IRN 1990 KEN 1997 KEN 1980 IRN 1990 KEN 1990 KEN 1990 2012 IRN 1965 MEX 1965 MEX 1967 1997 ISR 1965 MEX 1988 PER 1996 KEN 1997 MEX 1992 PER 1997 KEN 1997 MEX 1992 PER 1997 KEN 1998 MEX 2002 PER 1997 KEN 1998 MEX 2005 PER 1996 KEN 1998 MEX 1997 PKO 1990 KEN 1998 MEX 1997 PKO 1990 KEN 1998 MEX 1998 SLV 1990 MEX 1998 MEX 1998 SLV 1997 MEX 1998 MEX 1998 SLV 1997 MEX 1998 MEX 1998 SLV	Floods (n=143)	Droughts (n=70)	Heat waves (n=17)	Cold waves (n=28)	Floo (n=1	ods .41)	Drou (n=:	ights L09)	Heat waves (n=16)	Cold waves (n=28)
1980 IRN 1990 KEN 1997 NIC 1986 IRN 2001 KEN 2060 NIC 2012 IRN 2001 KEN 2060 NIC 2010 ISR 1967 MEX 1968 FER 1964 KEN 1977 MEX 1992 PER 1975 KEN 1978 MEX 2002 PER 1977 KEN 1978 MEX 2004 PER 1977 KEN 1982 KEN 2004 PER 1990 KEN 1982 MEX 2005 PER 1990 KEN 1984 MEX 2006 PER 1990 KEN 1986 MEX 1987 PNG 1991 KEN 1984 MEX 1997 NIC 1984 1992 MEX 1998 MEX 1997 NIC 1984 1993 MEX 1997 NIC 1984 TA 1993 MEX 1997 NIC	1968 IRN	2004 ZAF			1982	KEN	1994	NIC		
1986 IRN 1996 KEN 2000 NIC 2012 IRN 1965 KEN 1966 PER 1997 ISR 1965 MEX 1989 PER 1964 KEN 1970 MEX 1980 PER 1975 KEN 1977 MEX 1992 PER 1977 KEN 1978 MEX 2002 PER 1975 KEN 1978 MEX 1902 PER 1975 KEN 1984 MEX 1902 PER 1980 KEN 1984 MEX 1906 PER 1990 KEN 1984 MEX 1996 PER 1990 KEN 1984 MEX 1997 NIC 1990 KEN 1998 MEX 1997 NIC 1997 1971 MEX 1998 MEX 1997 NIC 1997 1972 MEX 1996 MEX 1997 NIC 1997 1974 MEX 1997	1980 IRN				1990	KEN	1997	NIC		
2012 IRN 2001 KEN 1966 PER 1997 ISR 1967 MEX 1989 PER 1964 KEN 1970 MEX 1989 PER 1968 KEN 1972 MEX 1992 PER 1975 KEN 1972 MEX 1992 PER 1982 KEN 1978 MEX 2002 PER 1997 KEN 1986 MEX 1980 PER 1998 KEN 1986 MEX 1980 PKS 1996 KEN 1998 MEX 1997 PNG 1996 KEN 1998 MEX 1998 SLV 1997 MEX 1998 MEX 1998 SLV 1997 MEX 1998 MEX 1998 SLV 1997 MEX 1998 MEX 1997 TCA 1998 MEX 1997 NC 1977 TZA 1998 MEX 1999 NC 1986 TZA </td <td>1986 IRN</td> <td></td> <td></td> <td></td> <td>1996</td> <td>KEN</td> <td>2000</td> <td>NIC</td> <td></td> <td></td>	1986 IRN				1996	KEN	2000	NIC		
1997 ISR 1965 MEX 1969 PER 2010 ISR 1967 MEX 1983 PER 1964 KEN 1970 MEX 1992 PER 1975 KEN 1972 MEX 1992 PER 1975 KEN 1984 MEX 2002 PER 1975 KEN 1984 MEX 2006 PER 1980 KEN 1984 MEX 2006 PER 1990 KEN 1986 MEX 1980 PIG 1990 KEN 1989 MEX 1987 PIG 1905 MEX 1998 MEX 1997 PIG 1906 MEX 1998 MEX 1998 SLV 1907 MEX 1908 MEX 1998 SLV 1970 MEX 1998 MEX 1998 SLV 1970 MEX 1909 NIC 1987 TA 1984 MEX 1999 NIC 1987 TA </td <td>2012 IRN</td> <td></td> <td></td> <td></td> <td>2001</td> <td>KEN</td> <td>1966</td> <td>PER</td> <td></td> <td></td>	2012 IRN				2001	KEN	1966	PER		
2010 ISR 1967 MEX 1983 PER 1964 KEN 1970 MEX 1990 PER 1975 KEN 1977 MEX 2002 PER 1975 KEN 1978 MEX 2004 PER 1975 KEN 1978 MEX 2004 PER 1980 KEN 1982 MEX 2004 PER 1990 KEN 1986 MEX 1980 PG 1990 KEN 1989 MEX 1982 SLV 1996 KEN 1993 MEX 1982 SLV 1996 MEX 1997 PIG 2001 SLV 1997 MEX 1998 MEX 1997 NC 1987 SLV 1970 MEX 1997 NIC 1987 SLV 1978 MEX 1970 NIC 1984 TA 1982 MEX 1990 NIC 1984 TA 1980 NIC 1984 TA 1984 <t< td=""><td>1997 ISR</td><td></td><td></td><td></td><td>1965</td><td>MFX</td><td>1969</td><td>PFR</td><td></td><td></td></t<>	1997 ISR				1965	MFX	1969	PFR		
1006 KKN 1907 MKX 1990 PER 1958 KEN 1972 MEX 1992 PER 1975 KEN 1972 MEX 2002 PER 1975 KEN 1984 MEX 2004 PER 1980 KEN 1984 MEX 2006 PER 1990 KEN 1984 MEX 2006 PER 1990 KEN 1984 MEX 2006 PER 1990 KEN 1989 MEX 1997 PNG 1995 MEX 1998 MEX 1997 PNG 1996 MEX 1998 MEX 1998 SLV 1997 MEX 1998 MEX 1994 SLV 1997 MEX 1996 MEX 1997 TZA 1986 MEX 1997 NIC 1977 TZA 1986 MEX 1997 NIC 1977 TZA 1986 MEX 1997 NIC 1977 TZA	2010 ISR				1967	MFX	1983	PFR		
1000 1000 1000 1000 PER 1975 KEN 1978 MEX 2002 PER 1977 KEN 1982 MEX 2006 PER 1980 KEN 1984 MEX 2006 PER 1990 KEN 1986 MEX 1980 PNG 1990 KEN 1986 MEX 1987 PNG 1990 KEN 1989 MEX 1987 PNG 1991 MEX 1989 MEX 1987 SLV 1967 MEX 1998 MEX 1997 PNG 1970 MEX 1998 MEX 1997 TZA 1987 MEX 1996 NIC 1987 TZA 1988 MEX 1990 NIC 1988 TZA 1984 MEX 1990 NIC 1988 TZA 1988 MEX 1990 NIC 1988 TZA 1988 MEX 1990 NIC 1988 TZA	1964 KEN				1970	MFX	1990	PFR		
1375 NEX 1378 NEX 1374 1375 NEX 1378 NEX 2002 PER 1377 NEX 1382 MEX 2004 PER 1395 KEN 1384 MEX 2006 PER 1396 KEN 1386 MEX 1390 PKI 1396 KEN 1398 MEX 1397 NKI 1396 KEN 1398 MEX 1397 NKI 1396 KEN 1398 MEX 1397 NKI 1396 KEN 1398 MEX 1397 NKI 1398 1397 MEX 1398 MEX 1398 NKI 1398 1397 MEX 1398 NKI 1397 NKI 1397 1398 MEX 1397 NIC 1397 TZA 1398 MEX 1397 NIC 1398 TZA 1398 MEX 1397 PER 2001 TZA 1399 MEX 1397 PER	1968 KEN				1972	MEX	1992	PFR		
1977 KEN 1982 MEX 2004 PER 1982 KEN 1984 MEX 2004 PER 1990 KEN 1986 MEX 1980 PNG 1996 KEN 1989 MEX 1997 PNG 1996 KEN 1989 MEX 1997 PNG 1996 KEN 1998 MEX 1997 PNG 1996 MEX 1997 PNG 1997 NEX 1996 MEX 1998 MEX 1989 SLV 1997 MEX 1998 MEX 1998 SLV 1970 MEX 1998 MEX 1997 TZA 1980 MEX 1999 NIC 1987 TZA 1981 MEX 1999 NIC 1988 TZA 1984 MEX 1999 NIC 1984 TZA 1993 MEX 1967 PER 2006 TZA 1994 MEX 1967 PER 1064 TZA	1975 KEN				1978	MEX	2002	PFR		
1392 KEN 1394 KEN 1206 FKR 1390 KEN 1384 MEX 1390 PNG 1390 KEN 1389 MEX 1392 PNG 1396 KEN 1393 MEX 1392 PNG 2001 KEN 1393 MEX 1394 SLV 1365 MEX 1396 MEX 1394 SLV 1366 MEX 1398 MEX 1394 SLV 1377 MEX 1398 MEX 2001 SLV 1378 MEX 1377 TZA 1384 MEX 1397 NIC 1384 TZA 1386 MEX 1397 NIC 1384 TZA 1386 MEX 1397 NIC 1384 TZA 1386 MEX 1397 NIC 1384 TZA 1398 MEX 1397 NIC 1384 TZA 1398 MEX 1397 VICA 1397 UGA 1398	1977 KEN				1982	MEX	2004	DER		
1000 NLL 1000 NLL 1980 NLL 1990 KEN 1980 MEX 1987 PNG 1996 KEN 1989 MEX 1982 SLV 1967 MEX 1993 MEX 1982 SLV 1970 MEX 1996 MEX 1998 MEX 1997 1970 MEX 1986 MEX 2001 SLV 1970 MEX 1986 MEX 2005 SLV 1970 MEX 1968 NIC 2007 SLV 1972 MEX 1968 NIC 1967 TZA 1984 MEX 1990 NIC 1984 TZA 1986 MEX 1999 NIC 1984 TZA 1986 MEX 1999 NIC 1984 TZA 1986 MEX 1999 NIC 1984 TZA 1986 MEX 1990 NIC 1984 TZA 1996 MEX 1997 NIC 1987	1082 KEN				108/	MEX	2004	DER		
1305 NLN 1305 NLN 1305 NLN 1996 KEN 1993 MEX 1997 PNG 1965 MEX 1996 MEX 1998 SLV 1965 MEX 1998 MEX 1998 SLV 1970 MEX 1998 MEX 1998 SLV 1970 MEX 1989 MEX 1998 SLV 1970 MEX 1989 MEX 1990 SLV 1978 MEX 1979 NIC 1977 TZA 1984 MEX 1990 NIC 1987 TZA 1984 MEX 1990 NIC 1987 TZA 1984 MEX 1990 NIC 1987 TZA 1989 MEX 2007 NIC 1987 TZA 1989 MEX 2007 NIC 1987 TZA 1993 MEX 1967 PER 2003 TZA 1994 PAK 1967 PER 1967 IGA	1000 KEN				1086	MEX	1080			
1395 NEN 1395 NEX 1993 NEX 1965 MEX 1994 MEX 1993 MEX 1994 SLV 1967 MEX 1998 MEX 1998 MEX 1998 SLV 1970 MEX 1998 MEX 1998 MEX 1998 SLV 1977 MEX 1968 NIC 2001 SLV 1978 MEX 1979 NIC 1977 TZA 1978 MEX 1979 NIC 1977 TZA 1988 TZA 1984 MEX 1990 NIC 1984 TZA 1985 MEX 1999 NIC 1984 TZA 1988 MEX 1990 NIC 1984 TZA 1998 MEX 1990 NIC 1984 TZA 1998 MEX 1967 PER 2007 TZA 1998 MEX 1967 PER 2064 TZA 1996 MEX 1967 PER 1962 TZA	1990 KEN				1000		1007			
2001 NEN 1995 NEX 1992 SLV 1965 MEX 1998 MEX 1998 SLV 1970 MEX 1998 MEX 1998 SLV 1970 MEX 1998 MEX 1995 SLV 1970 MEX 1978 MEX 1996 SLV 1972 MEX 1978 NIC 1967 TZA 1984 MEX 1979 NIC 1977 TZA 1984 MEX 1999 NIC 1984 TZA 1986 MEX 1999 NIC 1984 TZA 1986 MEX 2002 NIC 1984 TZA 1986 MEX 2007 NIC 1986 TZA 1998 MEX 2007 NIC 1986 TZA 1998 MEX 2007 NIC 1986 TZA 1998 MEX 1967 PER 2006 TZA 1998 MEX 1970 PER 1967 UGA	1990 KEN				1909		1002			
1965 MEX 1996 MEX 1994 SLV 1970 MEX 2013 MEX 2009 SLV 1972 MEX 1968 NIC 2009 SLV 1978 MEX 1976 NIC 1967 TZA 1982 MEX 1979 NIC 1984 TZA 1984 MEX 1999 NIC 1984 TZA 1986 MEX 1999 NIC 1984 TZA 1988 MEX 1999 NIC 1984 TZA 1996 MEX 2002 NIC 1984 TZA 1998 MEX 2007 NIC 1986 TZA 1998 MEX 1906 PER 2001 TZA 1998 MEX 1907 PER 2011 TZA 1998 MEX 1907 PER 1907 UGA 1916 PAK 1917 PER 1927 UGA 1917 PAK 1980 PER 1999 UGA	2001 KEN				1993		1982	SLV		
1967 MEX 1998 MEX 1998 SLV 1970 MEX 2013 MEX 2001 SLV 1972 MEX 1968 NIC 2003 SLV 1978 MEX 1976 NIC 1967 TZA 1982 MEX 1979 NIC 1977 TZA 1984 MEX 1990 NIC 1988 TZA 1985 MEX 1990 NIC 1988 TZA 1986 MEX 1990 NIC 1988 TZA 1998 MEX 2007 NIC 1986 TZA 1998 MEX 1906 PER 2003 TZA 1998 MEX 1970 PER 1063 1979 UGA 1976 PAK 1980 PER 1989	1965 MEX				1996	IVIEX	1994	SLV		
1970 MEX 2013 MEX 2001 SLV 1972 MEX 1968 NIC 2009 SLV 1978 MEX 1976 NIC 1977 TZA 1984 MEX 1979 NIC 1977 TZA 1984 MEX 1999 NIC 1984 TZA 1985 MEX 1999 NIC 1988 TZA 1986 MEX 2007 NIC 1991 TZA 1988 MEX 2007 NIC 1996 TZA 1998 MEX 2007 NIC 1996 TZA 1998 MEX 2007 NIC 1997 TZA 1998 MEX 2007 NIC 1997 TZA 1998 MEX 2007 NIC 1997 TZA 1998 MEX 1967 PER 2001 TZA 1994 PAK 1977 PER 1967 UGA 1977 PAK 1980 PER 1980 UGA	1967 MEX				1998	MEX	1998	SLV		
1972 MEX 1968 NIC 2009 SLV 1978 MEX 1976 NIC 1967 TZA 1984 MEX 1979 NIC 1984 TZA 1986 MEX 1999 NIC 1984 TZA 1986 MEX 1999 NIC 1984 TZA 1986 MEX 2002 NIC 1984 TZA 1987 MEX 2007 NIC 1984 TZA 1989 MEX 2007 NIC 1984 TZA 1996 MEX 2007 NIC 1986 TZA 1996 MEX 2007 NIC 1996 TZA 1998 MEX 1967 PER 2001 TZA 1998 MEX 1967 PER 2001 TZA 1998 MEX 1967 PER 1967 UGA 1967 PAK 1980 PER 1987 UGA 1976 PAK 1999 PER 2005 UGA	1970 MEX				2013	MEX	2001	SLV		
1978 MEX 1976 NIC 1967 TZA 1982 MEX 1979 NIC 1977 TZA 1984 MEX 1990 NIC 1984 TZA 1986 MEX 1999 NIC 1984 TZA 1987 MEX 2002 NIC 1991 TZA 1993 MEX 2002 NIC 1995 TZA 1994 MEX 2007 NIC 1996 TZA 1995 MEX 2007 NIC 1996 TZA 1996 MEX 2007 NIC 1997 TZA 1998 MEX 2007 NIC 1996 TZA 1998 MEX 1967 PER 2006 TZA 1998 MEX 1967 PER 1967 DGA 1977 PER 1967 PER 1979 UGA 1977 PAK 1992 PER 1980 LGA 1971 PAK 1999 PER 2002 UGA	1972 MEX				1968	NIC	2009	SLV		
1982 MEX 1979 NIC 1977 TZA 1984 MEX 1990 NIC 1984 TZA 1986 MEX 1999 NIC 1984 TZA 1989 MEX 2002 NIC 1984 TZA 1993 MEX 2007 NIC 1996 TZA 1994 MEX 2007 NIC 1996 TZA 1995 MEX 1965 PER 2003 TZA 1996 MEX 1967 PER 2006 TZA 1918 MEX 1967 PER 2006 TZA 1914 PAK 1977 PER 1967 UGA 1917 PER 1917 PER 1967 UGA 1918 PAK 1917 PER 1980 VGA 1919 PAK 1920 PER 1981 UGA 1919 PAK 1999 PER 2002 UGA 1919 PAK 1920 PER 1983 PIG	1978 MEX				1976	NIC	1967	TZA		
1984 MEX 1990 NIC 1984 TZA 1986 MEX 1999 NIC 1988 TZA 1989 MEX 2002 NIC 1996 TZA 1993 MEX 2007 NIC 1996 TZA 1996 MEX 1965 PER 2003 TZA 1998 MEX 1967 PER 2006 TZA 1994 PAK 1970 PER 2011 TZA 1964 PAK 1977 PER 1967 UGA 1975 PAK 1980 PER 1979 UGA 1976 PAK 1980 PER 1987 UGA 1988 PAK 1999 PER 2002 UGA 1991 PAK 1999 PER 2002 UGA 1991 PAK 1992 PER 1093 UGA 1991 PAK 2006 PER 1094 UGA 1991 PAK 2006 PER 1097 VIM	1982 MEX				1979	NIC	1977	TZA		
1986 MEX 1999 NIC 1988 TZA 1989 MEX 2002 NIC 1991 TZA 1993 MEX 2007 NIC 1996 TZA 1996 MEX 1965 PER 2003 TZA 1996 MEX 1967 PER 2006 TZA 1998 MEX 1970 PER 2011 TZA 1997 PER 1967 UGA 1979 UGA 1973 PAK 1980 PER 1987 UGA 1988 PAK 1999 PER 2002 UGA 1991 PAK 2006 PER 2005 UGA 1991 PAK 2006 PER 2005 UGA 1995 PER 1992 PIR 1997 VIM	1984 MEX				1990	NIC	1984	TZA		
1989 MEX 2002 NIC 1991 TZA 1993 MEX 2007 NIC 1996 TZA 1996 MEX 1965 PER 2003 TZA 1998 MEX 1967 PER 2006 TZA 1998 MEX 1967 PER 2006 TZA 1998 MEX 1967 PER 2001 TZA 1998 MEX 1967 PER 2001 TZA 1998 MEX 1967 PER 2001 TZA 1998 MEX 1970 PER 2011 TZA 1997 PAK 1998 PER 1997 UGA 1997 PAK 1999 PER 2002 UGA 1991 PAK 2006 PER 1992 VIGA 1991 PAK 2005 UGA 1993 VIM 1995 PER 1992 PIG 2011 UGA 1996 PER 1999 PIG 1911 UGA	1986 MEX				1999	NIC	1988	TZA		
1993 MEX 2007 NIC 1996 TZA 1996 MEX 1965 PER 2003 TZA 1998 MEX 1967 PER 2006 TZA 2013 MEX 1970 PER 2011 TZA 1964 PAK 1977 PER 1967 UGA 1975 PAK 1980 PER 1979 UGA 1976 PAK 1980 PER 1987 UGA 1976 PAK 1980 PER 1987 UGA 1976 PAK 1990 PER 1987 UGA 1976 PAK 1992 PER 1988 UGA 1988 PAK 1999 PER 2002 UGA 1991 PAK 2006 PER 2003 UGA 1991 PAK 1999 PER 2002 UGA 1991 PAK 1998 PIC 2004 UGA 1995 PER 1999 PIR 2004 1987	1989 MEX				2002	NIC	1991	TZA		
1996 MEX 1965 PER 2003 TZA 1998 MEX 1967 PER 2006 TZA 2013 MEX 1970 PER 2011 TZA 1964 PAK 1970 PER 2011 TZA 1967 PAK 1970 PER 1967 UGA 1973 PAK 1980 PER 1979 UGA 1976 PAK 1980 PER 1987 UGA 1977 PAK 1980 PER 1982 UGA 1978 PAK 1999 PER 2002 UGA 1988 PAK 1999 PER 2002 UGA 1991 PAK 2006 PER 2002 UGA 1991 PAK 2006 PER 2002 UGA 1991 PAK 1993 PK 2004 UGA 1965 PER 1999 PK 2004 PK 1997 1970 PER 2004 PK 1997 <	1993 MEX				2007	NIC	1996	TZA		
1998 MEX 1967 PER 2006 TZA 2013 MEX 1970 PER 2011 TZA 1964 PAK 1970 PER 2067 UGA 1967 PAK 1977 PER 1967 UGA 1973 PAK 1980 PER 1979 UGA 1976 PAK 1992 PER 1987 UGA 1978 PAK 1999 PER 1980 UGA 1978 PAK 1999 PER 1980 UGA 1988 PAK 1999 PER 2002 UGA 1991 PAK 2006 PER 2005 UGA 1991 PAK 2006 PER 2008 UGA 1995 PER 1999 PNG 2011 UGA 1967 PER 1999 PNG 1987 VNM 1970 PER 2004 PNG 1997 VNM 1980 PER 1999 NUM 1988 SLV	1996 MEX				1965	PER	2003	TZA		
2013 MEX 1970 PER 2011 TZA 1964 PAK 1977 PER 1967 UGA 1967 PAK 1980 PER 1979 UGA 1973 PAK 1980 PER 1987 UGA 1976 PAK 1992 PER 1987 UGA 1978 PAK 1992 PER 1988 UGA 1988 PAK 1999 PER 2002 UGA 1991 PAK 2006 PER 2005 UGA 1991 PAK 2006 PER 2005 UGA 1991 PAK 2006 PER 2008 UGA 1965 PER 1999 PNG 2011 UGA 1967 PER 1999 PNG 1987 VNM 1970 PER 2004 PNG 1997 VNM 1970 PER 2008 PNG 1997 VNM 1980 PER 1982 SLV 2005 VNM	1998 MEX				1967	PER	2006	TZA		
1964 PAK 1967 PAK 1967 PAK 1973 PAK 1974 PAK 1975 PAK 1976 PAK 1977 PER 1988 PAK 1991 PAK 2001 PAK 2001 PAK 1965 PER 1970 PER 1971 PER 1972 PK 1983 PAK 2001 PAK 2001 PAK 2001 PAK 2001 PAK 1983 PNG 2005 UGA 1995 PER 1992 PNG 2011 UGA 1967 PER 1999 PNG 1987 VNM 1970 PER 2004 PNG 1997 VNM 1980 PER 1992 VNM 2012 VNM 1980 PER 1982 SLV 2005 VNM 1999 P	2013 MEX				1970	PER	2011	TZA		
1967 PAK 1973 PAK 1974 PAK 1975 PAK 1976 PAK 1978 PAK 1988 PAK 1991 PAK 2001 PAK 2001 PAK 2001 PAK 1965 PER 1967 PER 1970 PER 1971 PER 1980 PER 1980 PER 1980 PER 1992 SLV 1994 PER 1995 SLV 1964 TUN	1964 PAK				1977	PER	1967	UGA		
1973 PAK 1976 PAK 1978 PAK 1988 PAK 1991 PAK 2001 PAK 2001 PAK 1965 PER 1976 PAK 2001 PAK 2001 PAK 1965 PER 1967 PER 1970 PER 1970 PER 1971 PER 1980 PER 1980 PER 1980 PER 1981 PER 1982 SLV 1983 SLV 1994 PER 1995 SLV 1996 TUN 1997 SLV	1967 PAK				1980	PER	1979	UGA		
1976 PAK 1988 PAK 1991 PAK 2001 PAK 2001 PAK 1965 PER 1967 PER 1977 PER 1970 PER 1971 PER 1972 PNG 1974 PER 1975 PER 1970 PER 1970 PER 1971 PER 1972 PER 1974 PER 1975 PER 1976 PER 1977 PER 1978 PER 1979 PER 1980 PER 1991 PER 1992 PER 1993 SLV 1994 PER 1995 SLV 1996 TUN 1997 TUN 1998 SLV 1999 SLV 1994 TUN 1995 SLV	1973 PAK				1986	PER	1987	UGA		
1988 PAK 1991 PAK 2001 PAK 2001 PAK 1965 PER 1967 PER 1970 PER 1977 PER 1978 PNG 1997 1970 PER 1977 PER 1978 PNG 1997 1970 PER 1971 PER 1972 PER 1974 PER 1975 PER 1976 PER 1977 PER 1980 PER 1980 PER 1980 PER 1991 PER 1992 PER 1993 PER 1994 PER 1995 SLV 1996 TUN 1997 TUN 1997 TUN 1997 TUN 1997 TUN	1976 PAK				1992	PER	1998	UGA		
1991 PAK 2001 PAK 1905 PER 1965 PER 1970 PER 1977 PER 1980 PNG 1997 1977 PER 1988 PNG 1997 1980 PER 1981 PNG 1997 1982 PNG 1997 1970 PER 1980 PER 1992 PER 1993 PER 1994 PER 1995 SLV 1996 FUN 1997 TUN 1998 SLV 1999 FUN 1990 SLV 1991 TUN 1992 TUN 1993 TUN 1994 TUN 1995 S	1988 PAK				1999	PER	2002	UGA		
2001 PAK 1965 PER 1967 PER 1970 PER 1977 PER 1978 PNG 1997 1977 PER 1988 PNG 1997 1987 PER 1988 PER 1988 PER 1999 SLV 1964 TUN 1965 TUN 1964 TUN 1973 TUN 1973 TUN 1974 TUN 1975 TUN 197	1991 PAK				2006	PER	2005	UGA		
1965 PER 1967 PER 1970 PER 1977 PER 1978 PNG 1997 VNM 1977 PER 1980 PER 2008 PNG 1999 VNM 1980 PER 2011 UGA 1997 VNM 1980 PER 2004 PNG 1999 VNM 1980 PER 2012 VNG 2002 VNM 1980 PER 1982 SLV 2005 VNM 1992 PER 1988 SLV 2005 VNM 1999 PER 1995 SLV 2005 VNM 1964 TUN 1999 SLV 2005 SLV 1963 TUN 2005 SLV 2011 SLV 2011 1973 TUN 2007 SLV 2011 2011 SLV 2011 1979 TUN 2011 SLV 2011 2011 2011 2011	2001 PAK				1983	PNG	2008	UGA		
1967PER1970PER1977PER1977PER1980PER1986PER1992PER1999PER1999PER1999PER1999PER1999PER1999PER1999PER1999PER1994TUN1995SLV1996TUN1997TUN1997TUN1999TUN1999TUN1999TUN1999TUN1999TUN1990TUN <td>1965 PER</td> <td></td> <td></td> <td></td> <td>1992</td> <td>PNG</td> <td>2011</td> <td>UGA</td> <td></td> <td></td>	1965 PER				1992	PNG	2011	UGA		
1970 PER 1977 PER 1977 PER 1980 PER 1986 PER 1997 PER 1999 PER 1990 PER 1991 PER 1992 PER 1993 PER 1994 TUN 1995 SLV 1996 TUN 1997 TUN 1998 SLV 1999 SLV 1999 TUN 1999 TUN 1999 TUN 1999 TUN 1999 TUN	1967 PER				1999	PNG	1987	VNM		
1977 PER 1980 PER 1986 PER 1992 PER 1999 PER 1999 PER 1906 PER 1997 PER 1998 SLV 1999 PER 1964 TUN 1965 SLV 1975 SLV 1975 SLV 1979 TUN 1979 TUN	1970 PER				2004	PNG	1997	VNM		
1980 PER 1986 PER 1992 PER 1999 PER 2006 PER 1997 PER 1998 SLV 1999 PER 1994 TUN 1995 SLV 1996 TUN 1997 TUN 1998 TUN 1999 TUN 1990 SLV 1991 SLV 1995 SLV 1996 TUN 1997 TUN 1998 SLV 1999 SLV 1999 SLV 1999 TUN 1999 TUN 1999 TUN 1999 TUN	1977 PER				2008	PNG	1999	VNM		
1986 PER 1992 PER 1999 PER 2006 PER 1964 TUN 1969 TUN 1973 TUN 2007 SLV 2011 SLV	1980 PER				2012	PNG	2002	VNM		
1992 PER 1999 PER 1999 PER 2006 PER 1964 TUN 1969 TUN 1967 TUN 1973 TUN 1979 TUN 2007 SLV 1979 TUN	1986 PER				1982	SLV	2005	VNM		
1999 PER 2006 PER 1964 TUN 1969 TUN 1973 TUN 1979 TUN 2007 SLV 1979 TUN	1992 PER				1988	SLV				
2006 PER 1995 SLV 1964 TUN 1999 SLV 1969 TUN 2005 SLV 1973 TUN 2007 SLV 1979 TUN 2011 SLV	1999 PER				1992	SLV				
1964 TUN 1969 TUN 1973 TUN 1979 TUN 2007 SLV 2011 SLV	2006 PER				1995	SLV				
1969 TUN 2005 SLV 1973 TUN 2007 SLV 1979 TUN 2011 SLV	1964 TUN				1999	SLV				
1973 TUN 1979 TUN 2011 SLV	1969 TUN				2005	SLV				
1979 TUN	1973 TUN				2007	SLV				
	1979 TUN				2011	SLV				
1982 TUN	1982 TUN				1964	TZA				

	Tropic	al Fruits			Co	offee	
Floods	Droughts	Heat waves	Cold waves	Floods	Droughts	Heat waves	Cold waves
(n=143)	(n=70)	(n=17)	(n=28)	(n=141)	(n=109)	(n=16)	(n=28)
1986 TUN				1968 TZA			
1990 TUN				1974 TZA			
2003 TUN				1978 TZA			
2007 TUN				1982 TZA			
2009 TUN				1986 TZA			
1964 TUR				1988 TZA			
1968 TUR				1993 TZA			
1974 TUR				1997 TZA			
1980 TUR				2000 TZA			
1984 TUR				2005 TZA			
1988 TUR				2008 TZA			
1990 TUR				2011 TZA			
1995 TUR				1997 UGA			
1998 TUR				2001 UGA			
2000 TUR				2006 UGA			
2009 TUR				2011 UGA			
2011 TUR				1964 VNM			
1968 ZAF				1966 VNM			
1974 ZAF				1970 VNM			
1977 ZAF				1978 VNM			
1981 ZAF				1980 VNM			
1987 ZAF				1984 VNM			
1993 ZAF				1990 VNM			
1999 ZAF				1998 VNM			
2006 ZAF							
2011 ZAF							

 Table B4. List of the extreme weather disasters that happened per year in the exporting countries (ISO3 codes) supplying the EU with cocoa. Data was aquired through (FAO, 2017).

	Coco	ba	
Floods (n=51)	Droughts (n=15)	Heat waves (n=1)	Cold waves (n=1)
1989 CIV	1983 CIV	2002 NGA	1992 NGA
1996 CIV	1971 CMR		
2007 CIV	1990 CMR		
2010 CIV	2001 CMR		
1988 CMR	2005 CMR		
1991 CMR	2012 CMR		
1994 CMR	1968 DOM		
1999 CMR	1964 ECU		
2005 CMR	1997 ECU		
2007 CMR	2009 ECU		
2010 CMR	2013 ECU		
2012 CMR	1971 GHA		
1979 DOM	1977 GHA		
1981 DOM	1983 GHA		
1985 DOM	1983 NGA		
1988 DOM			
1991 DOM			
1993 DOM			

	Coco	a	
Floods (n=51)	Droughts (n=15)	Heat waves (n=1)	Cold waves (n=1)
1996 DOM			
2002 DOM			
2007 DOM			
2009 DOM			
1965 ECU			
1967 ECU			
1970 ECU			
1973 ECU			
1982 ECU			
1987 ECU			
1989 ECU			
1992 ECU			
1997 ECU			
2000 ECU			
2006 ECU			
2008 ECU			
1968 GHA			
1989 GHA			
1991 GHA			
1995 GHA			
1999 GHA			
2001 GHA			
2007 GHA			
2013 GHA			
1985 NGA			
1988 NGA			
1994 NGA			
1998 NGA			
2009 NGA			
1996 SLE			
2004 SLE			
2007 SLE			
2009 SLE			

Irrigated area per agri-food product		
	n that the crop is irrigated.	indicati
be higher than harvested area), there is	g the effect of DR&HW in rainfed and non-tropical systems. Even if data on irrigated area is not consistent (as irrigation area is indicated to t	analysi
data from Cote d'Ivoire is removed while	analysis once yield, production and/or harvest area are not correctly reported or there is missing data at FAOSTAT. (6) ** means that banana c	from th
on banana from Suriname, are excluded	s on that specific category. (5) Data on tropical fruits from Panama and Ghana, on coffee from Ethiopia, on cocoa from Togo and Guinea and o	for crop
at AQUASTAT there is not available data	area higher than 40% are removed from the analysis. (4) Irrigation percentage of harvested area for 'Tropical Fruits' is not considered because a	harvest
countries with a percentage of irrigated	(2) A wet tropical climate is considered for the classifications 'Tropical rainforest climate' (Af) and 'Tropical monsoon climate' (Am). (3) Only the	climate
erference of irrigation and/or wet tropical	the ones that are not considered for the analysis of the impact of droughts & heat waves in rainfed and non-tropical systems, i.e. without the inte	grey ar
ountry. Notes: (1) Countries indicated in	5. Percentage of irrigated harvested area of the agrifood-product in each country and the dominant Koppen-Geiger climate classification per co	Table I

-				Irrigated area p	er agri-food product
Country	Agri-food product	numou	ητ κορρεη-Geiger classification (κοττεκ, Grieser, Beck, Kudoir, & Kubel, 2006)	(FAO, 2016)	and (FAO, 2017)
		Letter symbol	Classification	Percentage	Reference year
Algeria	Tropical Fruits	Csa	Hot-summer Mediterranean climate		
Belize	Banana	Am	Tropical monsoon climate	48%	2005
Brazil	Banana	Aw	Tropical savanna climate	5.4%	2006
Brazil	Coffee	Aw	Tropical savanna climate	11%	2006
Brazil	Soybeans	Aw	Tropical savanna climate	3%	2006
Brazil	Tropical Fruits	Aw	Tropical savanna climate		
Cameroon	Banana	Aw	Tropical savanna climate	6%	2000
Cameroon	Сосоа	Aw	Tropical savanna climate		
Cameroon	Coffee	Aw	Tropical savanna climate		
Canada	Soybeans	Dfb	Warm humid continental climate		
Chile	Tropical Fruits	Csb	Warm-summer Mediterranean climate		
Colombia	Banana	Af, Aw, Am	Tropical rainforest climate, Tropical savanna climate, Tropical monsoon climate	10%	1994
Colombia	Coffee	Af, Aw, Am	Tropical rainforest climate, Tropical savanna climate, Tropical monsoon climate	1%	2011
Costa Rica	Banana	Aw, Am, Af	Tropical savanna climate, Tropical monsoon climate, Tropical rainforest climate		
Costa Rica	Coffee	Aw, Am, Af	Tropical savanna climate, Tropical monsoon climate, Tropical rainforest climate		
Costa Rica	Tropical Fruits	Aw, Am, Af	Tropical savanna climate, Tropical monsoon climate, Tropical rainforest climate		
Cote d'Ivoire	Banana	Aw	Tropical savanna climate	242%	2008*
Cote d'Ivoire	Сосоа	Aw	Tropical savanna climate		
Cote d'Ivoire	Coffee	Aw	Tropical savanna climate		
Cote d'Ivoire	Tropical Fruits	Aw	Tropical savanna climate		
Dominica Republica	Banana	Aw, Af, Am	Tropical savanna climate, Tropical monsoon climate, Tropical rainforest climate		
Dominica Republica	Сосоа	Aw, Af, Am	Tropical savanna climate, Tropical monsoon climate, Tropical rainforest climate		
Ecuador	Banana	Aw	Tropical savanna climate	59%	2000
Ecuador	Сосоа	Aw	Tropical savanna climate	8%	2000

rrea per agri-food product 2016) and (FAO, 2017)	ge Reference year						2007								2010							2009				1994				2011	2013	
Irrigated a (FAO, 3	Percenta		%†				7%								13%							%86				5.1%				98%	1%	
ant Koppen-Geiger classification (Kottek, Grieser, Beck, Rudolf, & Rubel, 2006)	Classification	Tropical savanna climate	Tropical savanna climate	Oceanic climate	Tropical savanna climate	Hot semi-arid climates, Tropical savanna climate	Tropical rainforest climate	Hot-summer Mediterranean climate, Cold semi-arid climates, Hot humid continental climate	Hot-summer Mediterranean climate	Tropical savanna climate	Tropical savanna climate	Hot semi-arid climates, Hot desert climates	Hot semi-arid climates, Hot desert climates	Tropical savanna climate	Tropical savanna climate	Hot desert climates, Humid subtropical climate	Tropical monsoon climate, Tropical rainforest climate	Tropical monsoon climate, Tropical rainforest climate	Tropical rainforest climate	Humid subtropical climate	Oceanic climate	Oceanic climate	Tropical monsoon climate	Oceanic climate	Tropical rainforest climate	Tropical savanna climate						
Domin	Letter symbol	Aw	Aw	Cfb	Aw	Aw	Aw	Aw	Aw	Aw	BSh, Aw	Af	Csa, BSk, Dsa	Csa	Aw	AW	BSh, BWh	BSh, BWh	Aw	Aw	BWh, Cfa	Am, Af	Am, Af	Af	Cfa	Cfb	Cfb	Am	Cfb	Af	Aw	
Agri-food product		Tropical Fruits	Coffee	Coffee	Сосоа	Tropical Fruits	Coffee	Сосоа	Coffee	Tropical Fruits	Coffee	Coffee	Tropical Fruits	Tropical Fruits	Coffee	Tropical Fruits	Coffee	Tropical Fruits	Coffee	Сосоа	Tropical Fruits	Banana	Tropical Fruits	Coffee	Soybeans	Coffee	Tropical Fruits	Сосоа	Tropical Fruits	Banana	Coffee	
Country		Ecuador	El Salvador	Ethiopia	Ghana	Ghana	Guatemala	Guinea	Honduras	Honduras	India	Indonesia	Iran	Israel	Kenya	Kenya	Mexico	Mexico	Nicaragua	Nigeria	Pakistan	Panama	Panama	Papua New Guinea	Paraguay	Peru	Peru	Sierra Leone	South Africa	Suriname	Tanzania	

product	1-Geiger classification (Kottek, Grieser, Beck, Rudolf, & Rubel, 2006)	Irrigated area (FAO, 2010	per agri-food product 5) and (FAO, 2017)
etter symbol	Classification	Percentage	Reference year
sa, BWh, BSh	immer Mediterranean climate, Hot desert climates, Hot semi-arid climate.	5	
Csa, Dsb	¹ ot-summer Mediterranean climate, Warm humid continental climate		
Aw	Tropical savanna climate		
Dfb	Warm humid continental climate		
Dfa	Hot humid continental climate	%6	2008
Aw	Tropical savanna climate	54%	2005



Figure B1. Histograms of the 1000 resampled normalised means of aggregated crop production at the EWD years, i.e., floods (a), droughts & heat waves (b) and cold waves (c). Histograms show that data follows a normal distribution. In this example, statistically significant impacts are obtained for floods and droughts & heats waves with the majority of resampled normalised means bellow '1' (evidencing a negative impact). A non-significant impact is show in (c) where the majority of resampled normalised means is nearly distributed towards to '1', meaning no change on aggregated crop production during years with cold waves.



Figure B2. Observed influences of droughts and heat waves on crop production, yield and harvested area (columns) in rainfed and non-tropical systems. Results are shown for all crops aggregated (first row) and for soybeans, banana, tropical fruits, coffee and cocoa (second row). Mean event impact (%) is shown for the EWD years. "n" - number of droughts and heat waves considered in each normalised composite resampled 1000 times. The 1000 resampled normalised mean at the EWD is considered statistical significant for a level of significance up to 0.20 and not significant (n.s.) for higher than 0.20 (i.e., for confidence intervals below than 80%). For each EWD, the legend should read as: crop, mean event impact, number of selected events, level of significance.

References

- 1. AGRESTE, 2011. Les Prairies Vues par ISOP en Septembre 2011, Agreste Infos Rapides-Grandes Cultures et Fourrages, N°6/7. Montreuil, France.
- 2. Chatterton, J.C., Viavattene, J., Morris, E., Penning-Rowsell, Tapsell, S., 2010. The Costs of the Summer 2007 Floods in England Project: SC070039/R1. Bristol, UK.
- 3. EEA, 2010. Mapping the Impacts of Natural Hazards and Technological Accidents in Europe: An Overview of the Last Decade. Luxembourg.
- 4. FAO, 2017. FAOstat, Food and Agricultural Organization [WWW Document]. URL http://www.fao.org/faostat/en/ (accessed 1.5.19).
- 5. FAO, 2016. AQUASTAT Main Database. Food and Agriculture Organization of the United Nations (FAO) [WWW Document].
- IPCC, 2014. Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part B: Regional Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Barros, V.R., C.B. Field, D.J. Dokken, M.D. Mastrandre, Cambridge, United Kingdom and New York, NY, USA.
- C.B. Field, D.J. Dokken, M.D. Mastrandre. Cambridge, United Kingdom and New York, NY, USA.
 Kottek, M., Grieser, J., Beck, C., Rudolf, B., Rubel, F., 2006. World Map of the Köppen-Geiger climate classification updated. Meteorol. Z. 15, 259–263. https://doi.org/DOI: 10.1127/0941-2948/2006/0130

6.3 Appendix C

Drought and heatwave crop losses tripled over the last five decades in Europe



Figure C1. Relative distribution of cropping area by crops and climate region (a) Data is oriented from the highest to the lowest cropping area (%, averaged between 2008 and 2017) in the EU and in each dominant Koeppen-Geiger (KG) climate zones (Table C1, FAO, 2019 (FAO 2019)). Crop categories: cereals (CER), oil crops (OIL), olives (OLV), vegetables (VEG), grapes (GRP), roots and tubers (ROT), sugars (SUG), orchards (ORC), treenuts (TNT), citrus (CIT), soft fruits (STF) and other crops (OTR). Number of extreme weather disasters (EWD) reported between 1961 and 2018 per climate region (b) Data on EWD is taken from EM-DAT (EM-DAT 2018) and includes droughts (DR), heatwaves (HW), floods (FL), and cold waves (CW).



Figure C2. Flood crop responses in the first and second half of the observation record. The composite impact of cereal (CER), non-cereal (Non-CER) and both crop categories aggregated (Combined) is shown for production (1st row), yield (2nd row) and harvested area (3rd row), and is separated for the time slices 1964-1990 (1st column) and 1991-2015 (2nd column). Significance levels are as in Figure 4.1; n.n. if empirical bootstrapped distribution of the normalised mean is not normal (see Methods).


Figure C3. Cold wave crop responses in the first and second half of the observation record. The composite impact of cereal (CER), non-cereal (Non-CER) and both crop categories aggregated (Combined) is shown for production (1st row), yield (2nd row) and harvested area (3rd row), and is separated for the time slices 1964-1990 (1st column) and 1991-2015 (2nd column). Significance levels are the same as in Figure 4.1; n.n. if empirical bootstrapped distribution of the normalised mean is not normal (see Methods).



Normalised production anomaly

Table C1. Dominant Koeppen-Geiger (KG) climate zones (Kottek *et al* 2006) considered for each EU Member State (ISO 3166 alpha-2). Cfb – Temperate oceanic climate; Csa – Hot-summer Mediterranean climate; Dfb – Warm-summer humid continental climate; and Dfc – Subarctic climate.



Country	<u>ISO 3166</u> alpha 2	KG dominant classification	Country	<u>ISO 3166</u> <u>alpha 2</u>	KG dominant classification
Austria	AT	Dfb	Italy	IT	Csa
Belgium	BE	Cfb	Latvia	LV	Dfb
Bulgaria	BG	Cfb	Lithuania	LT	Dfb
Croatia	HR	Cfb	Luxembourg	LU	Cfb
Cyprus	CY	Csa	Malta	MT	Csa
Czechia	CZ	Cfb	Netherlands	NL	Cfb
Denmark	DK	Cfb	Poland	PL	Cfb
Estonia	EE	Dfb	Portugal	РТ	Csa
Finland	FI	Dfc	Romania	RO	Dfb
France	FR	Cfb	Slovakia	SK	Dfb
Germany	DE	Cfb	Slovenia	SI	Cfb
Greece	EL	Csa	Spain	ES	Csa
Hungary	HU	Cfb	Sweden	SE	Dfc
Ireland	IE	Cfb	United Kingdom	UK	Cfb

Crop category	Category code	FAOSTAT designation
Cereals	CER	Barley
Cereals	CER	Buckwheat
Cereals	CER	Cereals nes
Cereals	CER	Grain, mixed
Cereals	CER	Maize
Cereals	CER	Millet
Cereals	CER	Oats
Cereals	CER	Rye
Cereals	CER	Sorghum
Cereals	CER	Triticale
Cereals	CER	Wheat
Cereals	CER	Rice, paddy
Cereals	CER	Canary seed
Cereals	CER	Quinoa
Citrus	CIT	Lemons and limes
Citrus	CIT	Oranges
Citrus	CIT	Tangerines, mandarins, clementines, satsumas
Citrus	CIT	Fruit, citrus nes
Citrus	CIT	Grapefruit (inc. pomelos)
Grapes	GRP	Grapes
Oil crops	OIL	Hemp tow waste
Oil crops	OIL	Linseed
Oil crops	OIL	Oil, palm
Oil crops	OIL	Oilseeds nes
Oil crops	OIL	Poppy seed
Oil crops	OIL	Rapeseed
Oil crops	OIL	Soybeans
Oil crops	OIL	Sunflower seed
Oil crops	OIL	Castor oil seed
Oil crops	OIL	Groundnuts, with shell
Oil crops	OIL	Hempseed
Oil crops	OIL	Melonseed
Oil crops	OIL	Mustard seed
Oil crops	OIL	Seed cotton
Oil crops	OIL	Sesame seed
Oil crops	OIL	Safflower seed
Olives	OLV	Olives
Orchards	ORC	Apples
Orchards	ORC	Apricots
Orchards	ORC	Cherries
Orchards	ORC	Cherries, sour
Orchards	ORC	Peaches and nectarines
Orchards	ORC	Pears
Orchards	ORC	Plums and sloes

 Table C2. Crop categories considered in this study based on the crops (129 in total) listed by FAOSTAT (FAO 2019).

Crop category	Category code	FAOSTAT designation
Orchards	ORC	Quinces
Orchards	ORC	Kiwi fruit
Orchards	ORC	Figs
Orchards	ORC	Avocados
Orchards	ORC	Bananas
Orchards	ORC	Fruit, tropical fresh nes
Orchards	ORC	Mangoes, mangosteens, guavas
Orchards	ORC	Persimmons
Orchards	ORC	Pineapples
Orchards	ORC	Dates
Other	OTR	Flax fibre and tow
Other	OTR	Hops
Other	OTR	Mushrooms and truffles
Other	OTR	Spices nes
Other	OTR	Vanilla
Other	OTR	Anise, badian, fennel, coriander
Other	OTR	Cotton lint
Other	OTR	Cottonneseed
Other	OTR	Pepper (piper spp.)
Other	OTR	Peppermint
Other	OTR	Carobs
Other	OTR	Agave fibres nes
Other	OTR	Pyrethrum, dried
Other	OTR	Теа
Other	OTR	Bastfibres, other
Other	OTR	Coffee, green
Roots and tubers	ROT	Carrots and turnips
Roots and tubers	ROT	Potatoes
Roots and tubers	ROT	Sweet potatoes
Roots and tubers	ROT	Roots and tubers nes
Roots and tubers	ROT	Taro (cocoyam)
Roots and tubers	ROT	Yams
Soft fruits	STF	Berries nes
Soft fruits	STF	Currants
Soft fruits	STF	Gooseberries
Soft fruits	STF	Raspberries
Soft fruits	STF	Strawberries
Soft fruits	STF	Blueberries
Soft fruits	STF	Cranberries
Soft fruits	STF	Fruit, fresh nes
Soft fruits	STF	Fruit, stone nes
Sugar	SUG	Sugar beet
Sugar	SUG	Sugar cane
Treenuts	TNT	Walnuts, with shell
Treenuts	TNT	Almonds, with shell

Crop category	Category code	FAOSTAT designation
Treenuts	TNT	Chestnut
Treenuts	TNT	Hazelnuts, with shell
Treenuts	TNT	Nuts nes
Treenuts	TNT	Pistachios
Treenuts	TNT	Coconuts
Vegetables	VEG	Asparagus
Vegetables	VEG	Beans, dry
Vegetables	VEG	Beans, green
Vegetables	VEG	Broad beans, horse beans, dry
Vegetables	VEG	Cabbages and other brassicas
Vegetables	VEG	Cauliflowers and broccoli
Vegetables	VEG	Chillies and peppers, green
Vegetables	VEG	Cucumbers and gherkins
Vegetables	VEG	Eggplants (aubergines)
Vegetables	VEG	Garlic
Vegetables	VEG	Leeks, other alliaceous vegetables
Vegetables	VEG	Lettuce and chicory
Vegetables	VEG	Lupins
Vegetables	VEG	Maize, green
Vegetables	VEG	Melons, other (inc.cantaloupes)
Vegetables	VEG	Onions, dry
Vegetables	VEG	Peas, dry
Vegetables	VEG	Peas, green
Vegetables	VEG	Pulses nes
Vegetables	VEG	Pumpkins, squash and gourds
Vegetables	VEG	Spinach
Vegetables	VEG	Tomatoes
Vegetables	VEG	Vegetables, fresh nes
Vegetables	VEG	Vegetables, leguminous nes
Vegetables	VEG	Vetches
Vegetables	VEG	Watermelons
Vegetables	VEG	Chicory roots
Vegetables	VEG	Onions, shallots, green
Vegetables	VEG	Chick peas
Vegetables	VEG	Chillies and peppers, dry
Vegetables	VEG	Lentils
Vegetables	VEG	Cow peas, dry
Vegetables	VEG	Artichokes
Vegetables	VEG	Okra
Vegetables	VEG	String beans

the			ßÖ	0 2 0		RO	RO	RO	RO	RO	SK	S	ň	ň	ň	ň	Ν																				
/D of			1998	2001	2007	2009	2010	2012	2014	2018	2010	2012	1991	1997	2005	2009	2010																				
ne EV	aves	n 88**)	HR	R E		F	⊢	⊨	⊨	Ŀ	Ц	Ŀ	Ľ	N	N	N	N۲	N۲	٦L	ΡL	ΡL	ΡL	ΡL	Ы	ЪГ	ЪГ	ΡL	ΡL	ΡL	PL	ΡL	ΡL	ΡL	ΡL	ΡT	ΡT	ßÖ
han o	Cold wa	60 fron	2012	2001	2012 2012	2009	2010	2012	2018	2001	2010	2012	2018	2001	2003	2012	2005	2012	1997	1998	1999	2000	2001	2002	2008	2009	2010	2011	2012	2013	2014	2015	2016	2018	2009	2010	1993
nore t		=u)	AT I	AT L		ВЕ	BE	ВЕ	BG	ВG	BG	BG	BG	CZ	CZ	CZ	DE	DE	DE	DE	DE	Ш	Ш	Ш		ES	ES	ES	FR	FR	FR	FR	FR	FR	FR	ER ER	щщ
with r			6003	012	.985	2005	6003	012	998	000	2008	2010	2012	2010	2012	2018	1997	2005	6003	2010	2012	2012	2018	991	2012	971	.985	1995	.985	991	1997	2005	6003	2010	2012	2018	010
years able			N SK	X Z	× ×	N N	NK N	NK N	- Yu	NK N	NK	NK	NK	NK		NK N	UK]	NK N		UK	NK	NK	NN														
e are			005	000	013 013	977	993	994	997	998	666	000	001	002	004	007	008	600	012	013	014	015	017														
e ther sted in			20			30	30 1	30	30	30 1	30 1	30 2	30 2	30 2	30 2	30 2	30 2	30 2	30 2	30 2	30 2	30	30 2	02	õ	Ő	Э́Е	Ë	10	10	10	X	X	X	ž	X	×
 Sinc 			970	975 1	991	94 F	995 F	997	998	9 666	000	001 F	002 F	003 F	004 F	005 F	1 900	107	108	9 600	010 F	013 F	014 F	015 F	016 F	018 F	377 5	985 3	005	012	014	997 9	998	666	001	002	204
hd 4.4 by are			L 1			П 10	T 19	.0	.U	IL 19	JL 20	JL 20	L 2(۲ 2(L 2(L 20	۲ 20	ار 20	L 20	۲ 2(۲ 2(т 2(Т 19	Т 19	т 2(чТ 2(чТ 20	Т 19	T 19	T 19	чT 20	т 2(Т 2(
4.3 a molici			115 I	16	11 / 100 / 1	05 L	10 L)83 L	93 L	92 N	93 N	98 N	77 F	80 F	82 F	87 F	97 F	98 F	01 F	04 F	105 F	06 F	900 F	110 F)16 F	167 F	179 F	81 F	83 F	96 F	01 F	02 F	103 F	06 F	08 F	10 F	115 P
igures for sii			U 20	N 20		U 2C	U 20	U 15	U 15	19	19	19	19	19	19	15	15	16	20	20	20	20	20	20	20	15	15	15	15	15	20	20	20	20	20	20	20
ss in F which		274*)	02 H	10 1 1	н н 90	10 H	13 H	14 H	16 H	87 IE	93 IE	02 IE	31 EC	11 IE	17 IE	65 IT	66 IT	68 IT	70 IT	77 IT	85 IT	86 IT	87 IT	92 IT	94 IT	96 IT	D0	01 IT	02 IT	03 IT	D8 IT	TI 60	10 IT	11 IT	12 IT	13 IT	14 IT
e serie (**)	loods	from 2	20(200	200	20:	20:	20	20:	198	199	20(20(20	20:	19(19(19(19	19	198	198	198	199	196	196	200	20(200	200	200	J 20(20	J 20:	20:	J 20:	1 20
e tim /es 99	-	(n=141	3 FR		2 FR	3 FR	4 FR	6 FR	9 FR	0 FR	1 FR	2 FR	3 FR	5 FR	0 FR	1 FR	3 FR	4 FR	5 FR	6 FR	7 FR	8 FR	0 HF	1 HF	5 HF	6 HF	н	2 HF	4 HF	5 HF	8 HF	0 HL	е н	7 HL	0 HL	0 HC	1 HL
l in th Id way			198	198	199 199	199	199	199	199	200	200	200	200	200	201	201	201	201	201	201	201	201	200	200	200	200	201	201	201	201	201	197	199	199	199	200	200
cluded			EL C	ᆸ	나 나 나	EL	6 EL	7 EL	2 ES	2 ES	3 ES	8 ES	9 ES	2 ES	3 ES	7 ES	9 ES	4 ES	6 ES	7 ES	0 ES	2 ES	4 ES	6 ES	7 ES	0 ES	1 ES	2 ES	S ES	5 ES	6 ES	8 ES	Ē	7 FR	S FR	D FR	1 FR
are in 99(*) a			200	2010	201	201	201	201	196	197	197	197	197	198	198	198	198	199,	199(199	200	200	200	200	200	201(201	201	201	201	201	201	200	197	197	198(198
, but s is 30			BG	20	7 2	Ŋ	Ŋ	2	CZ	Ŋ	СΖ	С	DE	DE	DE	DE	DE	DE	EL	Ц	EL	EL	EL	EL		Ē	Ē	Ш									
siting flood			2018	1996	1998 1998	2000	2002	2005	2006	2009	2010	2013	1992	1993	1994	1997	1999	2002	2005	2006	2007	2009	2010	2011	2013	2016	2017	1977	1979	1994	1997	1998	2000	2001	2002	2003	2006
comp(AT	AT	ATA	AT	AT	AT	AT	AT	AT	AT	AT	AT	AT	AT	BE	BE	BE	BE	BE	BE	BE	BE	BE	BE	BE	ΒE	BG	BG	BG	ВG	ВG	BG	BG	BG	BG
r the (1980	1982	1985	1991	1995	1996	1997	1999	2002	2005	2006	2009	2013	2016	1971	1984	1987	1993	1994	1998	2002	2003	2005	2010	2011	2016	1997	2002	2005	2006	2007	2010	2012	2014	2015
ed to			E !		3 3	R	NL	NL	РТ	РТ	ΡТ	ΡТ	RO	SI	SK	SK	З	Ъ	NN																		
e omitt trv. th	aves	om 61)	2011	2018	2003	2003	2006	2018	2003	2005	2006	2018	1994	1996	1998	2000	2004	2005	2006	2007	2003	2003	2007	2003	2013	2018											
rd are	Heatw	=47 fr	AT :	AT 7	ы В С	BE	BE	BG	BG	∑	∑	∑	Ŋ	DE	DE	DE	Ц	Ц	Ш	ᆸ	Ц	ES	ES	ES	ES	ES	Æ	FR	FR	FR	FR	Ħ	HR	Ĥ	╘	⊨	F
a reco		Ľ	2003	2007	2006	2015	2018	2000	2007	1998	2000	2007	2003	2003	2006	2018	1985	1987	1988	2000	2007	1995	2003	2004	2006	2018	1990	2003	2006	2015	2018	2000	2003	2007	1998	2003	2007
in the		32)	BE	b B C B	2 2	ר ל	Ä	Ш	ES	ES	ES	ES	FR	FR	FR	FR	HR	Ĥ	ΠH	ΠH	F	F	⊨	⊨	Ľ	5	F	PL	ΡΤ	РТ	ΡΤ	ßÖ	ßÖ				
s of tr type	oughts	from 3	1976	983	0001 991	000	992	066.	.980	.981	066.	666.	.982	.989	.991	1997	003	.986	.992	003	1997	002	012	2017	1992	006	2018	018	.983	1997	004	.983	000				
year	Ď	(n=28		- c	0 4 1 1	5	6 1	7 1	8	9	10 1	11 1	12 1	13 1	14 1	15 1	16 2	17 1	18 1	19 2	20 1	21 2	22 2	23 2	24 1	25 2	26 2	27 2	28 1	29 1	30 2	31 1	32 2	33	34	35	36

1961 to 2018. The EWD considered in the composite analysis (Fig. 4.1 and 4.2) are indicated in black, while the ones in grey are excluded (see Methods). The first and last 3 Table C3. List of the year, country (ISO2 code of the EU members) and type of extreme weather disasters reported by the EM-DAT (EM-DAT 2018) data base between

129

fraction of the total EU cropping area. Average event impacts are shown (red if negative, green if positive) if statistically significant (*** if alpha < 0.05, ** if alpha < 0.10, * Table C4. Composite floods impact by crops and climate region. Observed production, yield and harvested area (H. area) impacts (%) associated with floods are separated for the EU level and for each Koeppen-Geiger (KG) region (Kottek et al 2006) between 1964 and 2015. Results are shown for major crop categories, ordered by the respective if *alpha* < 0.2, n.s. for not significant if *alpha* ≥ 0.2); n.n. if empirical bootstrapped distribution of the normalised mean is not normal. Statistical significance is based on 1000 bootstrap samples (see Methods). Column n indicates the number of events. Blank cells mean that a crop is not grown in the respective KG region.

													Dfh M/arm	rumor hum	id continor	+				
	ш	European Uni	ion-28		Cfb - Te	emperate oce	anic climate		Csa - Hot-sun	nmer Mediter	ranean clim	iate		climate		La la	Dfc	- Subarctic cli	nate	
-	Production	Yield	H.area	c	Production	Yield	H.area	c	Production	Yield	H.area	۲	Production	Yield	H.area	۲	Production	Yield	H.area	۲
-	-0.4 n.s.	0 n.s.	-0.2 n.s.	141	-0.9 n.s.	-0.3 n.s.	-0.6 n.s.	72	1.6 n.s.	1.2 *	-0.1 n.s.	42	-1.8 n.s.	-0.8 n.s.	0.6 n.n.	24	-7.1 *	-6.7 n.n.	-0.3 n.s.	æ
at	1 n.s.	0 n.s.	0.6 n.s.	141	-1 n.s.	-1.5 **	0.5 n.s.	72	4.1 n.s.	2.8 n.s.	0 n.s.	42	1.2 n.s.	-0.1 n.s.	1.1 n.s.	24	1.1 n.n.	-4.8 n.n.	6.6 n.n.	m
λä	-1.1 n.s.	-0.4 n.s.	-1.1 n.s.	141	-3.4 ***	-2.1 ***	-1.7 **	72	2.4 n.s.	2.8 n.s.	-0.7 n.s.	42	-0.9 n.s.	-0.7 n.s.	-0.4 n.s.	24	2.4 n.n.	-2.2 n.n.	4.8 n.n.	m
ze	1.7 n.s.	2.5 ***	-0.3 n.s.	128	5.3 ***	4.6 ***	1.3 n.s.	62	-1.8 n.s.	-1.6 **	-0.3 n.s.	42	-1.4 n.s.	4 n.s.	-4.3 ***	24				
als	-0.8 n.s.	-0.3 n.s.	-0.3 n.s.	141	-1.4 n.s.	-0.5 n.s.	-0.9 n.s.	72	1.6 n.s.	1.2 n.s.	0.1 n.s.	42	-2.7 n.s.	-1.8 n.s.	1.6 n.n.	24	-12.4 ***	-8.7 ***	-3.9 n.s.	m
15	-2.1 ***	-0.4 *	-1.8 ***	141	-3.3 ***	-0.8 ***	-2.4 ***	72	-1 ***	0.2 n.s.	-1.4 ***	42	-2 *	-0.8 n.s.	-1.3 n.s.	24	8.1 ***	3.4 *	4.7 **	m
bs	-2.1 *	0.7 n.s.	-2.3 *	141	3.3 *	-1.2 *	4.8 ***	72	-8.5 ***	1.4 n.s.	-9.4 ***	42	-4.6 *	4 **	-7.5 ***	24	0 n.n.	8.4 n.n.	-6.7 n.n.	m
es	2 n.s.	1.9 n.s.	-0.2 n.s.	52	0.5 n.s.	0.8 n.s.	-0.8 n.s.	18	2.8 n.s.	2.5 n.s.	0.2 n.s.	34								
es	-2.1 ***	-0.3 n.s.	-2.1 ***	141	-4 ***	0.1 n.s.	-4 ***	72	-0.1 n.s.	-0.5 *	-0.2 n.s.	42	-1.9 n.s.	-1.2 *	-1.4 n.s.	24	ۍ *	0.1 n.s.	5.9 *	e
es	-1.5 n.s.	-0.7 n.s.	-1.4 *	122	-4.9 *	-3.5 **	-2.3 n.s.	58	4.4 **	4.9 ***	-0.5 n.n.	42	-4.2 n.s.	-4.2 n.s.	-1 n.s.	22				
& S	-1.6 **	*	-1 n.s.	141	-2.5 **	0 n.s.	-2.8 ***	72	-0.3 n.s.	-1.4 **	0.7 n.s.	42	-4.3 *	-3.7 **	-1.6 n.s.	24	9.5 n.n.	5.1 ***	3.8 n.n.	ŝ
gar	-0.8 n.s.	0.1 n.s.	-1.7 n.s.	136	1.8 n.s.	-0.5 n.s.	1.3 n.s.	67	-7.5 ***	0 n.s.	-7.4 ***	42	8.1 ***	2.2 n.s.	3.9 *	24	12.9 n.n.	0.3 n.n.	11.5 n.n.	e
sp	-3 **	-1.9 ***	-1.3 ***	131	-6.4 ***	-3.4 ***	-2.7 ***	68	-0.4 n.s.	0.2 n.s.	-0.9 **	42	-1.7 n.s.	-4.8 **	2.4 n.s.	19	9.8 n.n.	7.5 n.n.	2.4 n.n.	7
ıts	-1.9 n.s.	-0.4 n.s.	-2.2 **	97	-3.1 n.s.	-2.8 n.s.	-0.9 n.s.	48	0.6 n.s.	2 *	-2.2 ***	40	-21.4 ***	-8.7 *	-11.9 *	6				
sn	-1.1 n.s.	0 n.s.	-1.8 *	57	-12.7 ***	-1.8 n.s.	-11.2 ***	15	2.1 *	0.5 n.s.	0.8 n.s.	42								
its	-1 n.s.	-0.8 n.s.	-0.1 n.s.	137	-2.9 **	-0.4 n.s.	-2.5 ***	71	-2.1 *	-2.2 ***	1.3 n.s.	42	4.9 n.s.	-1.3 n.s.	5 n.s.	21	23.4 n.n.	13.6 n.n.	4.7 *	m
bs	-3.6 ***	1.4 *	-3 *	138	-5.3 ***	-0.6 n.s.	-4.7 ***	72	-1.9 n.s.	3.1 *	-2.2 n.s.	42	-3.8 n.s.	2 n.s.	-0.7 n.s.	24				

separated for the EU level and for each Koeppen-Geiger (KG) region (Kottek et al 2006) between 1964 and 2015. Results are shown for major crop categories, ordered by the respective fraction of the total EU cropping area. Average event impacts are shown (red if negative, green if positive) if statistically significant (*** if a/pha < 0.05, ** if a/pha Table C5. Composite cold waves impact by crops and climate region. Observed production, yield and harvested area (H. area) impacts (%) associated with cold waves are < 0.10, * if alpha < 0.2, n.s. for not significant if alpha ≥ 0.2); n.n. if empirical bootstrapped distribution of the normalised mean is not normal. Statistical significance is based</p> on 1000 bootstrap samples (see Methods). Column n indicates the number of events. Blank cells mean that a crop is not grown in the respective KG region.

References

- EM-DAT 2018 The Emergency Events Database Universit'e catholique de Louvain (UCL) -CRED, D. Guha-Sapir. Brussels, Belgium Online: www.emdat.be/database ÷
- 2. FAO 2017 FAOstat, Food and Agricultural Organization Online: http://www.fao.org/faostat/en/
 - FAO 2019 FAOSTAT Online: http://www.fao.org/faostat/en/
- Kottek M, Grieser J, Beck C, Rudolf B and Rubel F 2006 World Map of the Köppen-Geiger climate classification updated Meteorol. Z. 15 259–63 Online: http://koeppengeiger.vu-wien.ac.at/present.htm 4